ENVIRONMENTAL PROGRESS

AUGUST 1992

Special Issue Featuring Remediation of Contaminate

.

0

Discussing remedial design services for groundwater and soil contamination (See page A6) Photo by Tom Graves, courtesy of CDM Inc., Cambridge, MA

PROCESS CONTROL CONTROL THEORY AND THE AND THE CPI...

Edited by Yaman Arkun W. Harmon Ray

CACHE

IS THERE A BRIGHTER FUTURE SHINING THROUGH ALL THAT JARGON?

espite its advances, process control theory has been slow to realize its full potential in the chemical processing industry.Control experts have "theorized" themselves out of the industrial loop. Control remains mostly an afterthought to process design. Yet safety and economics clearly demand more high-performance, non-linear control techniques in the CPI. Those are among the important issues addressed in the new volume of Chemical Process Control-CPCIV. This 700-page compendium contains 37 of the most significant papers from the Fourth International Conference on Chemical Process Control. It covers virtually every topic in the field that chemical engineers should know about. From inventory control in Japanese plants to modeling in neurobiology, these papers sum up five years of the very latest control applications from industry, academia, and government. This fully indexed volume covers:

- Industrial overviews from Japan, Europe, and North America
- · On-line sensors and data analysis
- · Dynamic process simulation, modeling, and identification
- · Model-based process modeling and control
- · Recent approaches to the control of nonlinear processes
- · Learning systems, adaptive, and AI control
- · Control technology in the year 2000

CHEMICAL PROCESS CONTROL—CPCIV, 1991, edited by Yaman Arkun, Georgia Institute of Technology, and W. Harmon Ray, University of Wisconsin. BE SURE TO ORDER YOUR COPY OF THIS IMPORTANT NEW REFERENCE! Mail this form to: AIChE Publications Dept. 345 E. 47th St., New York, NY 10017. or call (212) 705-7657, Fax (212) 752-3294.

Please send me a c CPC IV ISBN 0-81 Enclosed is my members take 20%	opy of CH 169-0549- check or off list pri	IEMICAL PR 5 money order fo ce. Internationa	OCES r \$95.0 d Price	S CONTROL— 0. AIChE \$133.00
Please bill my	Visa	Mastercard	#	
Expiration date	S	ignature		
SHIP TO:				
Name				
Organization				
Address				
City		State		_Zip
Country				

ENVIRONMENTAL PROGRESS

Environmenial Progress is a publication of the American Institute of Chemical Engineers. It will deal with multi-facened aspects of the pollution problem. It will provide through coverage of abatement, control, and containment of effluents and emissions within compliance standards. Papers will cover all aspects including water, at, liquid and solid wates. Progress and technological advances vital to the environmental engineer will be reported.

Editor-in Chief Mark D. Rosenzweig

> Editor Gary F. Bennett (419) 537-2520

Production Director **Daniel** Chillak

Managing Editor Maura N. Mullen (212)705-7327

Editorial Assistants Karen M. Simpson Thomas E. H. Campbell

> Copy Editor Arthur H. Baulch

> Washington Editor Martin W. Siegel

Book Review Editor **Robert W. Peters**

Software Review Ashok Kumar

Editorial Review Board

Robert C. Ahlert **R. Lee Byers** L.E. Erickson **Randy** Freeman Stephen C. James Atly Jefcoat Michael C. Kavanaugh William J. Lacy P. Lederman **R.** Mahalingham **Robert W. Peters** C. C. Reynolds C. J. Touhill J. A. Scher **Richard D. Siegel** Wel-Chi Ying

> Publisher Gary M. Rekstad

Published four times a year (February, May, August, and November) by the American Institute of Chemical Engineers, 35 B 47th 58, New York, NY, 10017 (ISSN 0278-4991) Manuscripte about be submitted to the Manuscript Center, AICM, 545 E 47th 58, New York, NY, 10017, Sataments and opinious in *Environmental Progress* are these of the con-tributors, and the American Institute of Chemical Bugineers year, S114-00, AICE Environmental Division Members, 252 included in dues, Coutside the US, pieses and S12 per subscription for postage and handling. Single copies S14, Justief the US, pieses del 33 for postage and handling. Payment must be made in U.S. dollars. Second-class postage point art york, NY, and edditional mailing offices. B 1992 by the American Institute of Chemical Engineers.

Special Remediation of Contaminated Soils and Groundwaters issue Robert W. Peters, Guest Editor

Contents

Implementation of Permanent and Alternative Remedial Technologies
Erik Dinemann, William Goldfarb and Robert C. Ahlert165
Critical Evaluation of PCB Remediation Technologies Laura J. Amend and Peter B. Lederman
Granular Activated Carbon Pilot Treatment Studies for Explosives Removal From
Contaminated Groundwater W. J. Wujick, W. L. Lowe, P. J. Marks and W.E. Sisk
Evaluation of a Hydraulically-Installed Suction Lysimeter to Obtain Representative Soil Water Samples
Mark Chaimberg, Ronald H. Carty and Joseph A. Scroppo
Ultrasonic Destruction of Chlorinated Compounds in Aqueous Solution Jiann M. Wu, H. S. Huang and C. D. Livengood
Remediation of Dichloromethane (DCM) Contaminated Ground Water Paul E. Flathman, Douglas E. Jerger and Patrick M. Woodhull
Effect of Chemical Pretreatment on Anaerobic Biodegradation of Refractory Organic Compounds <i>Vi-Tin Wang</i> , 210
Soil Decontamination via Microwave and Radio Frequency Co-Volatilization Clifford E. George, George R. Lightsey, Inchul Jun and Jiayi Fan
Supercritical Extraction of Polynuclear Aromatic Hydrocarbons from Soil S. Kothandaraman, R. C. Ahlert, E. S. Venkataramani and A. T. Andrews
The B.E.S.T. [®] Solvent Extraction Process Treatment of Soil, Sediment, and Sludges George R. Jones
Pneumatic Pumping Test for Soil Vacuum Extraction Jong Soo Cho and Dominic C. DiGiulio
Adsorption/Desorption Characteristics of Lead on Various Types of Soil Robert W. Peters and Linda Shem
Electrokinetic Removal of Selected Heavy Metals from Soil Sibel Pamukcu and J. Kenneth Wittle

		Door	ortmonte		
		naho	1101101110		
orgword					43
01011010	********	••••••	·····		
ditorial					A4
	·····	·····	••••••••	······	
nvironmental	Shorts				A6
SOOK HEVIEWS.		******	<i></i>	********	A/
Safeware Devie					
Sonware Hevie	W	·····	******	·····	AIU

Reproducing copies: The appearance of the code at the bottom of this page indicates the copyright owner's conint that Reproducing copies: In the appearance on the code as the contonion this page matcases the copyrigant owners or sustains usan for a stated fee copies of articles in this journal may be made for personal or internal use or for the personal or internal use of specific clients. This content is given on the condition that the copier pay the per-copy fee (appearing as part of the code) through the Copyright Clearance Center, Inc., 21 Congress St., Salem, MA 01970 for copying for general distri-permited by Section 107 or 108 of the U.S. Copyright Law. This content does not extend to copying for general distributton, for advertising or promotional purpose, for inclusion in a publication or for resale. Environmental Progress foe code: 0278-4491/92 \$3.00. Postmaster: Please send change o Progress, AIChE, 345 East 47th Street, New York, N.Y. 10017. : Please send change of addresses to Environ

A1



Now is the time to plan your entry, even though the judging is a few months away (February '93). Any person, firm, institution or government agency can enter, whether or not

affiliated with the Academy, any type of environmental engineering project - air, water supply, wastewater, hazardous waste, industrial hygiene, radiation protection or solid waste management.

The hallmark of the Academy's Award and the principal judging criteria include:

- · Demonstration of an integrated approach that considers all environmental media - air, water and land.
- Quality as evidenced by the degree of user satisfaction and proven . performance.
- Contribution to social and economic advancement.
- · Originality and innovation representing the application of new knowledge, a new application of existing knowledge or an innovative mix of existing knowledge.
- The complexity of the problem addressed.

Six GRAND PRIZES will be presented for excellence in - Research, Planning, Design, Operations/Management, Small Projects and University Research. Plus, other entries may qualify for HONOR AWARDS.

The Academy will actively promote the winning entries by featuring them in its exhibit at trade shows and conventions. Winning entries will also be featured in national publications. The public relations value is considerable.

Obtain your entry form and details by writing to Academy headquarters: 130 Holiday Court, Suite 100, Annapolis, MD or call 410-266-3311.



Evcellence in Environmental Engineering

RESEARCH

PLANNING

DESIGN

OPERATIONS/MANAGEMENT

This special issue of *Environmental Progress* is devoted solely to cleanup on contaminated soils and groundwaters. Conventional excavation and incineration with subsequent disposal in a licensed landfill is time-consuming, energy intensive and very expensive. New cost-effective technologies are needed for remediation of the many identified contaminated sites across our nation. Innovative treatment technologies are addressed in this special issue. The paradigm of employing synthetic feed-streams of high purity to simulate waste feed streams is highlighted in this issue's editorial. The papers contained in this special issue were presented at the AIChE Summer National Meeting (Pittsburgh) and the AIChE Annual Meeting (Los Angeles) in 1991. Of the thirty-four manuscripts submitted, thirteen were accepted for publication in this issue. All thirty-four papers were peer reviewed.

The first paper appearing in this issue addresses the Superfund remedy selection process. This paper is followed by an evaluation of PCB remediation technologies. In the remaining papers, many innovative remediation technologies and applications are covered, including granular activated carbon treatment for explosives, use of hydraulically installed lysimeters to obtain representative soil water samples, ultrasonic treatment for detoxifying groundwaters, effect of chemical pretreatment on anaerobic biodegradation of refractory organic compounds, supercritical extraction of polynuclear aromatic hydrocarbons, the B.E.S.T^m solvent extraction process, adsorption/desorption of metals from soil, electrokinetic removal of heavy metals, and soil vacuum extraction. These papers provide state-of-the-art technology for remediation of contaminated soils and groundwaters. They also indicate the broad scope of the remediation field, with the papers authored by professional staff in academia, research institutions/national laboratories, consulting firms, industry, and government agencies.

I wish to personally thank the authors for their cooperation in the preparation of their manuscripts for this special issue. I am especially grateful to the peer reviewers who assisted in the selection of the papers accepted for publication and helped to improve the technical quality of the papers. The entire issue is dedicated to all people interested in the field of soil and groundwater remediation.

Robert W. Peters, Guest Editor

Toward a Waste Treatment Research Paradigm in Academia

Louis J. Thibodeaux

At a recent conference, a speaker was proposing continuing research on an exotic catalyst for the thermal destruction of chlorinated hydrocarbons. In the proof-of-concept stage, several nines percent destruction efficiency was achieved. However, in response to a question, he admitted the process was constrained to a pure feed stream.

Investigators at numerous academic institutions are now performing waste treatment research using basically the same paradigm for chemical product/process development where extensive chemical separations to prepare the feed and purify the product are necessary. A dozen nines in efficiency will not render the catalytic process an economic reality since conventional incineration, in a rotary kiln, requires little feed preparation. The research should be discontinued immediately. Chemical product development projects typically progress through six somewhat distinct phases. These phases for waste treatment development are: (1) idea, (2) proof-of-concept, (3) bench-scale testing, (4) pilot-scale testing, (5) field demonstration, and (6) commercialization.

The research director and the investigators need to forge a relationship that fosters intellectual honesty since their goals are often times at odds. The former is interested in developing cost—effective, new technologies for waste treatment, while the latter is primarily interested in receiving funds to produce manuscripts and students.

The mutual understanding of a few simple facts and guidelines produces a clear paradigm for waste treatment research:

- the product is nearly always valueless;
- the feed is almost always "dirty";
- use actual waste as feed, never spiked sand as a surrogate soil for example;
- get expert advice on materials handling problems early; and
- multiple processing steps is usually a fatal flaw in the process design.

This is a partial list. The last item is one in several possible fatal flaws, so that an independent party should perform a rudimentary robustness check of a realistic operating scenario,

Environmental Progress (Vol. 11, No. 3)

including a cost comparison with the current available technologies. In a more extensive evaluation, systems analysis is performed to identify fatal flaws and limitations in the processing scheme, as well as performing cost analyses. In general, the project should be critically evaluated after each stage prior to proceeding further.

We must also question the wisdom of using the conventional six-step idea-to-commercialization approach. For example, consider the case of in situ bioremediation of contaminated bed sediment in a marine setting. Organisms are available that have shown promise of destroying the primary toxic contaminants and the PIs are seeking funds for the proofof-concept phase. Isn't it reasonable at this juncture to have the PIs assume that the microorganisms can do a six-nines conversion of the toxic contaminants and proceed immediately to Step 5, which is field demonstration, so as to concentrate on the engineering technology and associated costs of implementing a "facility" on the offshore site in order to: seed the bed with microorganisms and maintain the culture, provide and control the nutrient levels, monitor the conversion progress, etc., without substantial disturbance of the bed so as to release toxic materials to the water column. As this example illustrates, leapfrogging to step 5, as part of the research paradigm reality check, may shelve the project or cause its redirection. Proceeding algorithm-like through Steps 2, 3, and 4 developing an elegantly tailored microorganism system to later realize with back-of-the envelope engineering analysis at the eleventh hour that an obvious fatal flaw to implementation existed, suggested a gross lack of director involvement in the research program.

Limited resources force the research director to make early, hard decisions eliminating ideas with little pay-off potential. A clear research paradigm aids correct decisions and those investigators unwilling to operate under this altered and much constrained research paradigm should look elsewhere for funding.

Louis J. Thibodeaux, Ph.D., is Jesse Coates Professor of Chemical Engineering and Director, Hazardous Substance Research Center, South/Southwest, at Louisiana State University, Baton Rouge, LA 70803. Environmental Shorts

Remedial Design Services for Groundwater and Soil Contamination

Camp, Dresser & McKee Inc., (CDM) of Cambridge, Massachusetts, has develped advanced technical capabilities in all phases of remedial design for contaminated soils, groundwater, surface water, sediment, and airborne pollutants.

CDM engineers, scientists, and planners have worked with the in-house environmental staff of numerous industrial clients to expedite remedial designs.

Groundwater remediation

CDM has designed air and steam stripping and activated carbon adsorption treatment processes for alleviating groundwater contamination problems nationwide. For especially complex treatment problems they have explored even more sophisticated treatment technologies. They are currently using an ozone-hydrogen peroxide advanced oxidation process, combined with biological treatment, to clean up an aquifer contaminated by one of the largest gasoline spills in New York State.

To enhance their aquifer remediation programs, they have developed highly advanced computer goundwater models that illustrate how aquifer systems really work: in three dimensions. Their DYNFLOW model predicts groundwater flow vertically and horizontally; the DYNTRAK model maps the movement of contaminants through the aquifer. These models are able to delinate contaminant plumes from multiple sources.

Soil Contamination

CDM has also designed a wide range of soil remediation solutions using technologies as diverse as the contaminant problems themselves. For one industrial client, they designed an innovative - and cost-effective - vacuum extraction system to air-strip volatile organic contaminations from soils *in situ* without excavation, allowing plant operations to continue undisrupted. At an electric utilities site with high levels of pentachlorophenol in the soil, CDM specialists successfully pilot-tested a unique low-temperature thermal treatment system and developed a plan for solidification and capping of on-site waste material.

At a number of industrial sites, they are using innovative bioremediation processes for effective on-site soil treatment.

Air Emissions

CDM also provides full-service air quality management and emissions control, including ambient air monitoring and dispersion modeling in suport of permitting for hazardous waste incinerators.

New Degrees in Environmental Technology Offered at NYIT

Two new degrees in environmental technology are offered by New York Institute of Technology's School of Engineering and Technology, beginning this fall.

Heading the new environmental technology programs is Prof. Stanley M. Greenwald, P.E., who stated that "the rush by government agencies and industry to clean up the polluted and contaminated environment will not necessarily insure competitiveness in world markets nor a permanently clean environment".

The Bachelor of Technology degree is a traditional engineering technology program suitable for high school or community college graduates and military service personnel with interest in practical solutions to environmental problems. The Master of Science degree will provide advanced training in pollution control, alternative fuels, waste management or regulatory compliance for practicing professionals or individuals looking to change career directions.

Further information can be obtained by calling the Office of Admissions at 1-800-345-NYIT or the Dean's office (516) 686-7931.

Wastewater Treatment System Remains in Compliance During Cold Weather Operation

Cold temperatures can turn an otherwise well operating wastewater facility into a lower performance facility that often is not in compliance during the winter months. Sybron Chemicals Inc., Birmingham, N.J., has introduced a BI-CHEM[®] 1738 CW a custom blend of selectively adapted bacterial cultures that has been developed to enhance removal efficiencies duyring lower temperatures.

The effective, psychophillic (coldloving) blend of bacterial cultures works year-round to eliminate cold temperature non-compliance problems occurring duringthe winter months. Sybron BI-CHEM bacterial cultures have been selectively adapted to enhance the cold weather removal of emulsifiers, phenols, and detergents typically found in the wastewaters of chemical, petrochemical, refining, pulp and paper, food processing and textile industries.

With the onset of cold temperatures, most chemical and biochemical reactions exhibit a predictable slowdown. Sybron has found that BI- CHEM 1738, introduced at regular intervals prior the onset of cold weather, permits the culture to acclimate to improve the rate of waste reduction. During cold periods, BI-CHEM 1738 adapts quickly to thermal shocks that may reduce the ambient temperature extremes from 80° F to 39° F (27° C to 3.9° C).

Severe temperature drops slow the transport of nutrients and oxygen across the cell membrane of the bacteria and reduce the effectiveness of the culture. Membranes which typically contain high concentrations of saturated fatty acids interfere with nutrient transfer by becoming more rigid at lower temperatures.BI-CHEM 1738, which contains a much higher concentration of unsaturated fatty acids in the cell membrane, reduces the impact of low temperatures on nutrient transport by permitting the cell membrane to remain more fluid.

For more information contact Mark Krupka, Sybron Chemicals Inc., Birmingham Road, Birmingham, N.J. 08011. Phone (toll-free) 1-800-678-0020 or call (609) 893-1100. Hazardous Waste Management, Charles A. Wentz, McGraw-Hill, New York, N.Y., 461 pages. ISBN 0-07-069291-2, U.S. List Price:\$ 46.95 (1989).

Up to this point, I have been unable to find a book that I felt comfortable using as a text for a hazardous waste course for no book (in my opinion) had been written for that purpose. Consequently, I was delighted to read in the preface of the text:

.. The primary purpose of Hazardous Waste Management is to integrate a broad field into a single book that deals with all phases of this important subject ...

This textbook (emphasis added by the reviewer) has been written at a level suitable for the senior undergraduate or beginning graduate student who has knowledge of the basic fundamentals of science and engineering.

The material (in the text) brings together legislation, regulations, technology and business matters relating to hazardous wastes management. In order to understand this dynamic field better, numerous case studies have been provided, which should assist the reader in relating to actual environmental situations.

The author has achieved his stated purpose, writing 14 different chapters covering all the significant aspects of hazardous waste disposal. He provides enough information to educate the reader, but he does not suffocate him with the voluminous amount of detail that is available. This reviewer is well aware of the amount that could have been written. as more that one book devoted solely on the topic of each chapter rests on his bookshelf.

One desirable aspect of the book is the inclusion of problems/exercises at the end of each chapter and provision of a Solutions Manual for the instructor. Clearly, these exercises enhance the book's value as a text.

The first chapter starts rather slowly with the RCRA definition of hazardous waste. Although this may be an appropriate starting point (i.e., defining the legal aspects of the problem), I would personally have "overviewed" the topic before diving into details (especially details as boring as the law), even though they are very important. Additionally, I question Wentz's inclusion of almost five pages of text on the reactor accident at Seveso, Italy, where dioxin contamination resulted. I will admit that one result of the accident was Resource Recovery" in which

the creation of hazardous waste to be disposed of, but that was really a tangential problem to the health risk caused by the dioxins. The second chapter, devoted to Risk Assessment, is a concise discussion of the topic, but unfortunately it is too concise, thus manifesting the problem noted in the opening paragraph. I said that Wentz avoided overwhelming the reader with information, but there is a fine line between too little and too much information. However, I wished for a little more information here perhaps in the form of a numerical example of risk calculation, especially of risk that might occur from hazardous waste management or disposal. The two examples given in the chapter deal with virgin chemicals (ethylene dibromide and lead) resulting from use, not from waste disposal.

Chapter 3 is a short one, overviewing all of the significant Federal environmental laws from the Rivers and Harbors Act of 1899 to the 1986 Amendments to CERCLA. Approximately one page is devoted to giving the reader an overview of each law. Wentz followed with a whole chapter devoted solely to TSCA, RCRA, and CERCLA/SARA. This is very appropriate because these three laws are the keys to the text (or at least they are the legal keys). Surprisingly, Wentz also included some material which is questionable regarding its relation to hazardous waste. The material consists of a discussion of the contamination of the Life Sciences Plant at Hopewell, Virginia, in 1974. That was an interesting problem (and a most significant pollution problem for the local POTW and the chemical plant employees) but until the plant was razed, it was not a hazardous waste problem.

Chapter 5 is a strange combination of hazardous waste characterization and site assessment. The first 10 pages of the chapter are devoted to a discussion of what characteristics make a waste hazardous. I would have placed this section in the previous chapter in which the laws governing hazardous waste were discussed, and then continue with site assessment as the author did in a separate chapter. In the site assessment section, he thoroughly reviewed two interesting contaminated sites: (1) PCB contamination at a GE site in Oakland, California, and (2) solvent spills at the Biocraft Laboratories Plant in Waldeck, New Jersey.

Chapter 6 is the real beginning of the text (at least from the technical viewpoint of hazardous waste management). It is titled "Waste Management and manifestation of the new concept of thepositive aspects of waste disposal are preventing its generation are addressed. Interestingly, early in the chapter, Wentz discusses government policy which is a dangerous but necessary process because it changes so quickly. Government policy is even more volatile in the field of waste management. Indeed, the U.S. EPA has established a new office of "Pollution Prevention", a little more far reaching concept than simple waste minimization. Several bills on this topic are before Congress at the moment.

Successive chapters deal with Chemical, Physical, and Biological Treatment (dealing mainly with aqueous hazardous waste streams) and Thermal Processes (including incineration of PCBs under TSCA). Not included in the incineration discussion, however, was a intensive discourse on air pollutants and their control, otherwise this is an excellent chapter. Also included was a very good discussion of Chemical Waste Management's incinerator ships, Vulcanis I and Vulcanis II, transportation of hazardous waste including placarding and manifesting and emergency aspects. Another chapter dealt with groundwater contamination, detection, and sampling.

Chapters 10 and 11 are devoted to what one chemical engineer has called perpetual" storage: Landfill Disposal and Deepwell Injection. In chapter 11, Wentz begins his discussion with a case study — The Love Canal — to show what can happen when landfills are improperly managed.

The book ends with two not very hopeful chapters. In chapter 14, Wentz discusses the very different topics of "Process Selection" and "Facility Siting" The problem here is the NIMBY Syndrome and the great public fear of hazardous chemicals/wastes. Indeed, Wentz himself painfully experienced public wrath when he tried to permit an incinerator in North Dakota.

Frankly, I note that there are four short appendices parenthetically. I routinely chastise authors for including long lists or excerpts from the Register of Material that is not essential to the reader and could easily be obtained from the source document. Such is not the case here. Wentz lists: (A) priority pollutants, (B) hazardous waste from non-specific sources, (C) hazardous waste from specific sources, and (D) listing of EPA hazardous waste from commercial chemical products, intermediates, and residues. All of the appendices contain useful data and utilize only 25 pages.

In summary, although I feel there are

topics or organizations I would change (or have written differently) I could never have produced such an excellent book. As I said at the beginning, the author has produced the first book on hazardous waste that I could cheerfully adopt as a course text. Hazardous Waste Management contains a good, basic discussion of the laws, technology, and background of the hazardous waste problem and its solution (or at least management). The use of case histories provide an excellent point of departure from class discussion to be followed by assignment of the problem/exercise at the end of each chapter. I personally would supplement the book with outside readings in areas I wanted to cover more thoroughly, but would be confident that the student had the basic knowledge from reading the text to allow him to handle that subject.

Gary F. Bennett, Ph.D., P.E.

Professor of Biochemical Engineering Department of Chemical Engineering, University of Toledo, Toledo, OH 43606

Contamination of Ground Water: Prevention, Assessment, Restoration Pollution Technology Review No. 184 Michael Barcelona, Allen Wehrmann, Joseph F. Keely, and Wayne A. Pettyjohn, Noyes Data Corp. Park Ridge, NJ, 213 pages, ISBN 0-8155-1243-0, U.S. List Price: \$45.00 (1990)

Groundwater contamination has been detected in every state and the number of cases is on the increase. The subsurface environment of groundwater is characterized by a complex interplay of physical, geochemical, and biological effects that govern the release, transport, and fate of a variety of chemicals. This book addresses three main areas: (1) prevention of contamination, (2) assessment of the extent of contamination, and (3) restoration of groundwater quality. The stated purpose of the book is "to discuss measures that can be taken to ensure that uncertainties do not undermine our ability to make reliable predictions about the response of contaminants to various corrective or preventive measures". The authors are from the Illinois State Water Survey, the U.S. Environmental Protection Agency, and Oklahoma State University. They are experienced and well qualified in the subject area of this book.

The book is broken into two parts and contains nine chapters and an appendix.

Part I addresses the framework for protecting groundwater resources while Part II addresses the scientific and technical background for assessing and protecting the quality of groundwater resources.

In Part I, Chapter 1 begins with groundwater contamination addressing definitions (such as contaminant and groundwater contamination), the extent of groundwater contamination, the general mechanism of groundwater contamination, sources of groundwater contamination, and movement of contaminants in groundwater. Chapter 2 deals with groundwater quality investigations (regional, local, or site), protocol for conducting the investigation, and descriptions and results from several investigations. Chapter 3 addresses groundwater restoration covering topics such as subsurface effects on contaminant mobility, physical containment techniques, hydrodynamic controls, withdrawal and treatment, in situ treatment techniques, and treatment trains.

Part II begins with Chapter 4 on basic hydrogeology in which precipitation, infiltration, surface water, groundwater, and the relationship between surface water and groundwater are discussed. Chapter 5 concerns monitoring well design and construction, covering groundwater monitoring program goals, monitoring well design components, and monitoring well drilling techniques. Groundwater sampling is discussed in Chapter 6 beginning with an introduction and subsequent discussions on the establishment of a sampling point and elements of the sampling protocol. The use of groundwater tracers is discussed in Chapter 7. Topics covered in this chapter include: general characteristics of tracers, public health considerations, direction of water movement, travel time, sorption phenomena of tracers, hydrodynamic and molecular dispersion, practical aspects, and types of tracers. Chapter 8 discusses the use of models in managing groundwater protection programs and covers the utility of models; assumption, limitation and quality control; applications in practical settings; and liabilities, costs, and recommendations for managers. Chapter 9 deals with basic geology. Topics discussed in this chapter include: geologic maps and cross-sections; groundwater in igneous rocks, metamorphic rocks, sedimentary rocks, sedimentary rocks and unconsolidated sediments; the relationship between geology, climate, and groundwater quality; minerals; rocks; weathering; erosion and deposition; geologic structure; and geologic time. The book concludes with an appendix listing sources of information about groundwater contamination. Further, each chapter provides a reference section listing all the references cited in that chapter.

The book contains a vast array and wealth of information and should be a useful reference book to both inexperienced and experienced people in the field of groundwater contamination and remediation.

> Robert W. Peters, Ph.D., P. E., Environmental Systems Engineer Energy Systems Division Argonne National Laboratory 9700 South Cass Avenue Argonne, IL 60439

Experimental Design in Biotechnology, Statistics: Textbooks and Monographs, Volume 105 Perry D. Haaland Marcel Dekker, Inc., New York, NY, 259 pages [ISBN No.: 0-8247-7881-2] U.S. List Price: \$ 89.75 (1989).

While engineers have long relied on empirical experimentation to develop processes, scientists have often argued that a complete understanding of a process is needed before it can be properly optimized. Consequently, engineers often engage in complex and expensive empirical factorial design experiments while scientists often engage in complex and expensive mechanistic studies aimed at modeling a process by attempting to reconstruct it from its component parts: a process which may ignore higher-order interactions. For complex processes where many variables can affect the output, both approaches may be difficult to apply successfully. Fortunately, in recent years, due to the development of statistics and computer programs which make their application usable by non-specialists, there are now powerful techniques available which can greatly enhance the information and efficiency of experimentation. Unfortunately, many of the statistics texts which describe these methods are difficult for the non-statistician to understand and apply to their particular problem.

This book is a welcome exception. It gives a lucid account of the experimental issues, the appropriate experimental designs, the application of statistics, and the interpretation of the results. While focussing on specific problems encountered in biotechnology, this book is very readable for the "non-expert" and may be generally applied to any complex biological problem where limited time and resources are available. The book contains 9 chapters and an appendix containing specific experimental designs and statistical tests for easy reference. The introductory chapter provides an overview of difficulties inherent in collecting information-rich data, separating signals from noise, applications to actual examples in biotechnology processes, and alternative approaches for problem solving.

Chapter 2 addresses the step-by-step approach taken for optimizing *in-vivo* monoclonal antibody production. This includes a description of the problem, initial screening experiments, an optimization experiment, and a verification experiment. This example is a full factorial design experiment in which factors are assumed to have two levels (e.g., high and low).

Chapter 3 discusses clear signal designs for reducing the size of the experiment while maintaining information about two factor interactions. The optimization of a sandwich-immunoassay is used to examine the effects of a variety of fractional factorial designs and the reasons why clear signal designs are preferable.

Statistical models and methods used for separating signals from the noise are examined in Chapter 4. In this case, the stability of the enzyme-immunoassay is examined. A variety of statistical methods are used including graphical plots and regression analyses. These methods are standard for several popular statistical packages.

Chapter 5 covers the types of experimental designs, their power and sample size, and types of potential errors. In this chapter, optimization of a cell culture system for producing monoclonal antibodies is investigated.

Chapter 6 discusses the statistical transformations which may be needed when developing an appropriate multiple regression model. This chapter is useful since this aspect of multiple regression is often mystifying to nonstatisticians.

Chapter 7 addresses mixed-level designs in which one or more factors have three rather than two levels. In this case, non-orthogonal designs are considered and applied to examining factors which can affect a bioactive coating of a polyurethane surface.

Chapter 8 examines the use of iterative strategies for solving empirical problems. This chapter presents a strategy approach to determine which variables are significant by identifying important experimental factors, eliminating unimportant experimental factors, and identifying near optimal settings through a series of small experiments.

The final chapter ties together the iterative strategies discussed in Chapter 8 with the experimental designs described in previous chapters. In this case, methods are combined in order to improve the stability of a microbead-based immunoassay. Finally, the Appendix lists the experimental designs used including quick reference tables for screening experiments and tests for curvature.

This book is both comprehensive, well written, and appropriate for non-experts in experimental design, statistical methods, and biotechnology. It is recommended reading for any biological scientist or engineer who wishes to begin utilizing powerful state-of-the-art statistical design methods in their experimental program in order to reduce the number of experiments needed, maximize the amount of information obtained, and properly interpret the results.

James R. Frank, Ph.D., Manager, Waste Management & Bioengineering Section Argonne National Laboratory 9700 South Cass Avenue Argonne, IL 60439

Put Successful Process Safety Systems to Work NEW BOOKS & VIDEOTAPES

GUIDELINES FOR HAZARD EVALUATION PROCEDURES, Second Edition with WORKED EXAMPLES 1992 461pp \$120* 0-8169-0491X

Over 6,000 copies of the first edition in use. Chemical Engineering, among others, praised it as "...laudable." This book is a correrstone of safety planning. State and federal regulatory agencies cite it. Now, this classic work is presented in a totally new edition with chapters and sections added to keep you upto-date. The worked examples make it very useful to training courses.

PLANT GUIDELINES FOR TECHNICAL MANAGEMENT OF CHEMICAL PROCESS SAFETY 1992 382pp \$120* 0-8169-04995

An easy to use work for the plant level. Provides a practical blueprint for implementing the 12 elements of chemical process safety management: Accountability • Process Knowledge and Documentation • Capital Project Review & Design Procedures • Process Risk Management • Management of Change • Process and Equipment Integrity • Incident Investigation • Training & Performance • Human Factors • Standards, Codes, & Laws • Audits and Corrective Action • Enhancement of Process Safety Knowledge

For a full catalog/to order: Publications Sales, American Institute of Chemical Engineers, 345 E. 47 Street, New York, NY 10017 *Call 212-705-7657 for speedy credit card orders and sponsor/member & quantity prices.

Center for Chemical Process Safety / AIChE

Two New Training Courses on Videotape

The colorful graphics and latest in videotape technology keep interest focused. Learning has never been easier.

MANAGING PROCESS SAFETY – A VIEW FROM THE TOP

\$495 tape + Leader's Guide 800-80

A lively presentation, developed from CCPS' current best selling book Guidelines for Technical Management of Chemical Process Safety, designed for manufacturing and engineering managers. Stress is given to accountability; process risk management; standards, codes and laws; and capital project review. The comprehensive Leader's Guide relates the 12 key elements of process safety to OSHA 29 CFR Part 1910.119, Process Safety Management of Highly Hazardous Materials.

PROCESS SAFETY FOR PLANT PERSONNEL \$495 tape + Leader's Guide 800-81

This tape provides information and motivation for plant personnel. Built from Plant Guidelines for Technical Management of Chemical Process Safety, just published and described above, this training video uses real world, in-plant scenarios to show the important role of each of the individuals responsible for operating and maintaining process systems. It spotlights process documentation, training and performance, process equipment integrity, management of change, and human factors, and includes explanations of all the 12 key elements.

Sources of Environmental Software

Ashok Kumar and Kode Srinivas

Department of Civil Engineering, University of Toledo, Toledo, OH 43606

INTRODUCTION

Software dealing with environmental pollution problems are used by environmental professionals every day. The applications include regulatory compliance, design of pollution control equipment, processing and analysis of monitoring and emission data, preparation of environmental impact assessment and emergency planning. One of the questions asked by environmental managers is: Where do I look for current pollution software dealing with pollution problems? The purpose of this paper is to discuss various sources of environmental software. Discussion also includes some examples of available software.

SOURCES OF SOFTWARE

Information of environmental software can be found in journals, trade publications, books and environmental bulletin boards. Various sources may be grouped as follows:

(a) Journal Reviews:

These are important sources for obtaining views on current software. Over the last ten years environmental journals have been adding a software review column or publishing a directory of available software for their readers. Software are reviewed on a regular basis in the Journal of the Air and Waste Management Association, Environmental Progress, Pollution Engineering, American Industrial Hygiene Association Journal and others. By looking at these journals one can obtain different view points on the same software. This will help in the selection of software for your particular application.

Table 1 and Table 2 are examples of software articles published in the Journal of the Air and Waste Management Association during 1990-91 and Environmental Progress during 1989-92, respectively.

(b) Directory: An annual directory of environmental software is published by the pollution Engineering Journal [1], the Chemical Engineering Progress (AIChE) [2], and Donley Technology [3]. These directories are detailed and include information on the application of software.

> The level of details of environmental software has increased dramatically since Pollution Engineering began to

review programs in 1985. Early environmental programs examined in Pollution Engineering were merely general software packages repackaged under an environmental label. The 1992 annual review of Pollution Engineering gives the details of the company and its software packages related to areas of air, physical/ chemical data, cost, energy analysis, fluids (air and water), comprehensive environmental data management and water. This review also gives the various aspects in which a particular software can be used.

Chemical Engineering progress software directory [2] is an important source of information for environmental software. More than 1,200 programs from vendors and chemical engineering faculties in the United States of America (U.S.), Canada, the U.K., Europe and elsewhere are covered in 1991-92 directory. The programs are divided into 27 categories. Each of these categories provide a brief description for every program. The tables in the directory also indicate how many years the program has been on the market. The information on hardware requirements, operating systems, peripheral devices, and other features such as source code availability, hot line assistance and tutorials is also included in the directory.

Donley Technology's Environmental Software Directory gives information on commercial and government software, on-line systems, and databases in the environmental field. The description, hardware and software requirements, a point of contact, and cost information for about 500 systems are listed in the annual directory (1990-91 edition). The new edition contains over 100 new systems and the updates for previously listed systems.

This quarterly journal publishes re-Software Journal: viewed papers on environmental software. The papers include technical details, application and availability of software. The journal is published by Elsevier Science Publishers Ltd.

Environmental

(c)

NAME	VENDOR	REQUIREMENTS	PURPOSE/TYPES
Graph Plus Ver. 1.3 etc.	Micrographix, Inc. Richardson, TX	IBM-PC, 640 Kb hard disk	presentation, graphics
Registrar, Ver. 2.0	Silton-Bookman Cupertino, CA	IBM PC with hard disk, also available in LAN format	student training, tracking system
Spectra Calc.	Galactic Indust. Corp. Salem, NH	IBM-PC with at least 512 k memory	Scientific data processing/ package
Reg. Scan RCRA	Regulation Scanning Corp.	IBM-PC, MS-DOS 3.0, 400 kb of RAM, 3 MB on hard disk	It provides, in computer ac- cessible form, the complete text of the solid waste regu- lations in the Code of Fed- eral Regulations for title 40 CFR parts 240-299.
Waste Reduction Advisory System (WRAS), Ver. 2.3	Illinois Hazardous Waste Research and Information Center Savoy, Illinois	IBM-PC, 640 kb RAM, DOS 2-1 or higher, printer	It is intended for use as an interactive waste manage- ment tool designed to in- crease a generator's knowl- edge range of options for re- ducing and recycling indus- trial waste and to access citations and abstracts of ar- ticles on waste reduction.
Fluid Mechanics Programs for IBM PC, 1987 ver.	Mc-Graw Hill Book Company New York, NY	IBM-PC, MS-DOS 2.0, 64 kb of RAM	Educational programs writ- ten in BASICA.
Equil	Micromath Scientific Software Salt Lake City, Utah	IBM-PC with at least 640 k of memory	To assist the solution of complex equilibrium prob- lems.
Airstrip, Ver. 1.2	AirStrip, Ames, IA	IBM-PC with hard disk and printer	It is represented as a design tool or air-stripping towers to be used for removal of volatile organic contaminants (VOC's) from water supplies.
MINSQ	Micromath Scientific Software Salt Lake City, Utah	IBM-PC with at least two drives	Nonlinear curve fitting and parameter estimation pro- gram. It is particularly useful for solving problems in chemical kinetics.
CHEM Master Ver. 2.10, 2.10	Softtouch Envirogenics Langhorne, PA	IBM-PC/XT, PC/AT, 640 kb RAM with 540 kb free RAM	Right to know data base pro- gram. It might be helpful in maintaining information on hazardous chemicals, keeping track of the use of those chemicals and prepare com- munity and worker right-to- know reports.
EPI Code, Ver. 4.1	Homann Associates, Inc. Fremont, CA	IBM-PC, PC/XT, PC/AT, or compatibles	A Gaussian model for evalu- ating the atmospheric disper- sion of wide range of chem- ical releases. It was designed to provide quick estimates of downwind concentrations and can run on a minimal IBM compatible personal computer system.
Audit Master, Ver. 3.02	Utilicom, Inc. Pittsford, NY (Continued)	IBM-PC, PC/XT, PC/AT, 512 kb RAM	Audit Master is a modular computer program that as- sists you in conducting health, safety, and environ- mental audits, to determine if operations and practices comply with applicable regu- lations.

Table 1 Software Reviews published in Journal of the Air and Waste Management Association (AWMA) during the period 1990 and 1991

Environmental Progress (Vol. 11, No. 3)

August, 1992 A11

(Continued from musican more	x		
Chemtox Database Release 1	Resource Consultants Inc. Brentwood, Tennessee	IBM-PC, PC/XT, PC/AT, 512 kb RAM. At least one 5 1/4" or 3 1/2" double sided diskette drive. 10 MB free space on hard disk.	This program is very useful when one wants to access a comprehensive information reference for toxic sub- stances. It functions much like a technical reference li- brary, with information on physical properties, toxico- logical information, regula- tory requirements, emergency response, personnel protec- tion, and much more.
Map Viewer, Ver. 1.0	Golden Software, Inc. Golden, Colorado	80286 & higher processor, windows 3.0 and higher, hard disk with 3 MB free space	Map viewer is a graphics program to create maps to communicate info. for spe- cific geographic areas.
Reg-A-Dex, Ver. 2.4	J. J. Keller & Associates Inc., Neenah, Wisconsin	IBM-PC, PC/XT, PC/AT, 1 MB free space on hard disk	Reg-A-Dex is a chemical reg- ulation data base which al- lows you to quickly deter- mine whether a substance ap- pears on any one of several regulatory lists, such as SARA, RCRA lists.
Environmental Series Infobase	Legislative Information Systems Corporation Annandale, VA	IBM-PC, PC/XT, PC/AT & compat. 512 kb RAM	Environmental regulation in- formation data base main- tains 18 volumes of the federal "Code of Federal Regulations" in computer- readable "information" and distributes quarterly and monthly updates based on actions published in the "federal register." The envi- ronmental series infobase must be used in conjunction with views by Folio (or pre- views, its runtime version)
Graftool, Ver. 3.3	3-D Visions, Inc. Redondo Beach, CA	IBM-PC, PC/AT, PC/XT, DOS Ver. 3.0 and higher, 2 MB free on hard disk	Technical graphics
OAQPS Technology Transfer Network Bulletin Board	U.S. EPA	Access to a PC modem, and communication software	Computer bulletin board, which provides access to info. regarding federal air quality regulatory programs, documents, data bases, and software tools.
Map Info. for windows Ver. 1.1	Map Info. Corporation Troy, NY	IBM PC/AT, PS/2 or com- patible, 2 MB RAM	This is a geographic informa- tion system. It might be helpful when working with large amounts of data and need to visualize and analyze those data graphically.
note: IBM-PC in this table implies all I	BM-PC's and compatibles		

- (d) Research Software:
 This category includes software originated from universities and research institutions. A review of journals will give you an idea on the type of software developed. Generally, you will have to contact researchers to find out about the availability of software. Some of the software developed at these institutions will appear as commercial software at a later date.
- (e) Commercial Software:
 There is a tremendous growth in the development of environmental software by consulting companies in the U.S., Canada and Europe. Advertisements appearing in trade journals and magazines and promotions during pollution conferences will assist you in locating the right software for your plant.
 (f) Electronic
- Bulletin Boards: Inese are inexpensive sources of obtaining pollution software. A paper

Table 2 Examples of software artic	les published in Env	ironmental Progress during 1989-92
TITLE	VOLUME/NO.	DESCRIPTION
PC-SARA: A Software for Emergency Planning and Analysis	Vol. 8, No. 2	PC-SARA is a software developed by Technol- ogy and Management Systems, Inc., Burlington, BA, for emergency planning and analysis. It is designed to help industries and local communi- ties in the U.S. comply with many of the re- quirements of SARA Title III.
WHAZAN: A Software for Chemical Hazard Analysis	Vol. 8, No. 3	The WHAZAN (World Bank Hazard Analysis) was developed by Technica International, Fuller- ton, CA. It can be used during an emergency in- volving chemicals to calculate and predict the concentration as a result of a chemical release.
HWI: A Software for Hazardous Waste Inciner- ators	Vol. 8, No. 4	HWI, developed by Professors Theodore and Reynolds from Chemical Engineering Depart- ment, Manhattan College, Riverdale, NY, assists in the preliminary design and analysis of inciner- ators.
TEST AIR: A Program for Air Pollution Test- ing Methods	Vol. 9, No. 1	The procedure used in the program is in accord- ance with the calculations given in the Title 40 of the Code Reference of Federal Regulations (United States of America), part 60, Appendix A (40 CFR 60, Appendix A). Help is available from Dawn Graphics, Winchester, MA.
The use of TK Solver for Environmental Engi- neering Problems	Vol. 9, No. 2	This software, developed by Universal Technical Systems Inc., Rockford, IL, helps environmental scientists/engineers in industry, regulatory bodies or universities in solving problems which involve equations (algebraic as well as differential) con- suming lot of time.
An Indoor Air Quality Model	Vol. 9, No. 3	IAQ (Indoor Air Quality) model by Dr. Leslie E. Sparks of Air and Energy Engineering Research Laboratory of the U.S.E.P.A., NC, estimates the impact of various pollutant sources on in- door air quality.
Software for Regulatory Compliance of Chemi- cal Hazards	Vol. 9, No. 4	This article briefly describes the available soft- ware for managing chemical hazards. They are grouped into six categories.
Use of Bulletin Board Systems for Air Quality Modeling	Vol. 10, No. 1	This article reviews about the available software programs related to air quality modeling from bulletin board systems like SCRAM, NTIS, etc.
Software to Study Environmental Effects of Hazardous Waste Sites	Vol. 10, No. 2	The software for hazardous waste sites are grouped into five categories and are discussed in this article.
Use of Electronic Bulletin Boards for Environ- mental Management	Vol. 10, No. 3	This article gives a brief description of the avail- able 9 Bulletin Board Systems.
A Review of Three Toxic Release Screening Models	Vol. 11, No. 1	This article gives a brief description about the models TSCREEN, ARCHIE, and CAMEO which are widely used in the U.S. The sources and the cost of obtaining each model is also listed in the article.
Environmental Data Bases	Vol. 11, No. 2	Available popular data bases, their use, and the contact address to acquire them are briefly discussed in this article. Data bases concerned to air are of prime focus in this article.

by Kumar and Rao [4] provides information on some of the electronic bulletin boards available to you. They allow you to download pollution software, emission data, monitoring data, and regulatory information at your convenience. (g) Newsletters:

Newsletters published by the California Air Resources Board (P.O. Box 2815, Sacramento, CA 95812), The European Association for the Science of Air Pollution (available from Prof. P. J. H. Builtjes, MT-TNO, P.O. Box 342, 7300 AH Apel-

RISK * ASSISTANT

Developed by Hampshire Research Institute

Distributed by Thistle Publishing, Alexandria, VA

Purpose:

To assess chemical exposures and risks at specific sites. It allows users to rapidly evaluate exposures to, and human health risks from, chemicals in the environment, and to present the results in the form of detailed reports.

Hardware/Software Requirements

Runs on IBM/compatible microcomputers **RAM** Required 512 K Hard-disk Memory 5 Mb **Operating System** DOS 2.0 or greater Other Requirements an 80 column printer Cost \$199.95 (according to November, 1991 issue) Technical support, database updates, and user bulletin board available at the developer. VOC-CALC Developed by Air Quality Consultants, Inc. Diamond Bar, CA Purpose: To keep daily records of VOC's emitted into the atmosphere from their facility. It tracks VOC emissions by permit, equipment, and product according to the South Coast Air Quality Management District (SCAQMD) requirements. Hardware/Software Requirements Runs on IBM/Compatible microcomputers Diskettes: 5 1/4" or 3 1/2" Hard-disk Memory 4 Mb RAM required 640 Kb **Operating System** DOS 2.0 or greater Language written in: Clipper Cost

\$1,495 (according to November, 1991 issue)

Free demo available: technical support, users manual, and training available at the developer.

		doorn, The Netherlands), Donley Technology's Environmental Soft- ware Report, and major consulting companies are also useful for learn- ing about software and the modifi- cation in software. For example, the programs RISK * ASSISTANT and VOC-CALC are listed in Table 3, which are taken from Donley Tech- nology's Environmental Software report, November 1991.
(h)	Book Publishers:	Book Publishers are also entering into the lucrative market of pollution software. They are packaging the software and user's guide like a book. For example, Lewis Publish- ers announced the availability of At- mospheric Oxidation Rate Program, Biodegradation Probability Pro- gram, and Hydrolysis Rate Program in 1992.
(i)	Information Database:	A search through information da- tabase is also helpful in locating soft- ware. For example, AIRBASE database is maintained by Air Infil-

tration and Ventilation Centre, University of Warwick Science Park, Coventry, U.K.

CONCLUDING REMARKS

Information on environmental software is scattered throughout the literature. Over 3,000 environmental software are available from the sources listed in this article. However, there is not a single source to obtain details on all the available software. It is hoped that the information presented will be helpful in your search for a software.

LITERATURE CITED

- Rich, G., "Annual Review 92," Pollution Engineering, pp. 45-66 (Jan. 1992).
- 2. "CEP Software Directory 1991-1992," (A Supplement to Chemical Engineering Progress), AIChE (1991).
- "Environmental Software Directory," Donley Technology, Garrisonville, VA (1991).
- Kumar, A., and H. G. Rao, "Use of Electronic Bulletin Boards for Environmental Management," Vol. 10, No. 3, pp. A12-14, Environmental Progress (1991).

Evolution of the Superfund Remedy Selection Process, Including an Assessment of Implementation of Permanent and Alternative Remedial Technologies

Erik Dienemann

Merck & Co., Inc., 126 E. Lincoln Ave, Rahway, NJ 07065

William Goldfarb

Rutgers University, Dep't. Environmental Resources, Cook College, New Brunswick, NJ 08903

Robert C. Ahlert

Rutgers University, Dep't. Chemical and Biochemical Eng'g., PO 909, Piscataway, NJ 08855

The history of the Superfund program has been controversial, especially with respect to the development of the remedy selection process. During the early 1980's, USEPA rarely selected permanent remedies or alternative treatment technologies, relying on disposal and/or containment remedies. Dissatisifaction with the pace and extent of cleanup of the Nation's hazarous waste sites resulted in public and political pressure to address this situation. In response, USEPA's selection of alternative remedial technologies accelerated somewhat during the mid 1980's. At the same time, the original Superfund law and key regulations were being overhauled, culuminating in enactment of the Superfund Amendments and Reauthorization Act (SARA) in late 1986. SARA emphasized selection of permanent remedies and alternative remedial technologies that treat a site's contaminated source materials. SARA also established the Superfund Innovative Technology Evaluation (SITE) program to evaluate alternative remedial technologies. Since SARA, USEPA claims continued acceleration in the selection of permanent remedies and alternative technologies, while critics question the permanence of many of EPA's selected remedies.

INTRODUCTION

In general, there are two kinds of technological innovations: i) innovations driven by the pursuit of scientific knowledge, without regard for possible applications, and ii) innovations driven by public and/or political demands to solve newly perceived problems. In the former, applications are viewed as secondary to the advancement of knowledge, while in the latter, the application of existing technology to new problems constitutes innovation, in and of itself. In the field of hazardous waste site cleanup, most technological innovations have been of the latter kind, that is, application of existing technologies to newly perceived problems, in response to public and political demands. The use of any permanent or alternative treatment technology has been considered to be an example of "technological innovation" (which is different, however, from "innovative alternative technology", as defined under Superfund). Technological innovations are necessary but not sufficient to effect the remediation of hazardous wastes and hazardous waste sites. Political will and an effective regulatory framework are also required, to spur technology development and implementation.

REMEDY SELECTION UNDER SUPERFUND

Evolution of CERCLA/SARA

The Comprehensive Environmental Response, Compensation and Liability Act (CERCLA) became law in December 1980 [*I*]. In enacting CERCLA, Congress had two basic goals: (i) to effect cleanup of abandoned/uncontrolled hazardous waste sites and (ii) to ensure that the parties responsible for creating these problems would bear the response costs [2]. To implement CERCLA, Congress provided a "Superfund" of \$1.6 billion over five years. The Superfund, generated largely from taxes on the petroleum and chemical industries, was to be used when Responsible Parties (RPs) refused or were unable to pay for or perform site cleanups, or when no responsible parties could be identified. At the time, it was hoped that this program would be sufficient to repair the damage from careless hazardous waste disposal [3].

The cornerstone of CERCLA was a provision to revise and amend the National Contingency Plan (NCP), issued under Section 311 of the Clean Water Act for emergency cleanup of oil spills. The revised NCP was supposed to specify criteria for hazardous waste site identification, investigation and prioritization, and provides technical criteria for cleanup. As part of the NCP, EPA was instructed to develop a National Priority List (NPL) of at least 400 of the Nation's worst sites to be eligible for Superfund monies. The focus of CERCLA is remediation of these NPL sites, although CERCLA also contains provisions for emergency removal actions at non-NPL sites.

Promulgation of a revised NCP was necessary before EPA could begin a full-scale cleanup program. Delays in establishing a new NCP, as well as the unexpectedly slow obligation of Superfund revenues for emergency and remedial response actions, led to the resignations of EPA Administrator Burford and several of her appointed staff in February 1983 [4]. EPA published the revised NCP in July 1982 [5].

By 1985, it had become apparent that the extent of the hazardous waste site problem was much greater than orginally anticipated. EPA estimated 2,000 eventual NPL sites [6], while the Office of Technology Assessment (OTA) estimated 10,000 eventual NPL sites [7]. The time and expense of remediating all of these sites were vast; estimates varied between 5 and 100 years at a cost of between \$ billion and \$100 billion [7]. Furthermore, the pace of site cleanups was far below original expectations; at the end of 1984, EPA considered cleanup actions to be completed at only 10 of the 538 sites on the NPL [8]. The effectiveness of site removal and remedial actions was dubious; in general, wastes were excavated from one site and land-disposed at another or were contained on-site [7], even though most land disposal/containment facilities were expected to fail eventually [9].

In March 1985, Congress began discussion of CERCLA reauthorization, eventually leading to passage of the Superfund Amendments and Reauthorization Act (SARA) [10], in October 1986. Congress greatly reduced EPA discretion under SARA, especially in the area of site cleanup criteria [11, 12]. SARA cleanup standards, the 1990 revisions to the NCP, and the EPA Superfund Innovative Technology Evaluation (SITE) program have had a significant impact on hazardous waste site remediation, especially with respect to the selection of permanent and/or alternative cleanup technologies. In addition, SARA authorized an expanded \$8.5 billion Superfund, over its 5-year authorization period, for hazardous waste site problem was much larger than originally expected.

SARA Cleanup Standards

The 1980 Statute and the 1982 NCP did not specify any particular cleanup standards; these criteria were left largely to

the discretion of EPA. The NCP simply stated that a selected remedy should represent the lowest cost alternative that effectively protects public health, welfare and the environment. The lack of clearly defined cleanup standards led EPA to frequently select "inexpensive", impermanent remedies, such as off-site disposal and on-site containment [7].

SARA Section 121, entitled "Cleanup Standards", establishes requirements governing selection of remedies at NPL sites, specifying that the President select remedial actions that are protective of human health and the environment, costeffective, in accordance with the rest of Section 121, and in accordance with the NCP. In evaluating cost-effectiveness, both short and long-term costs must be taken into account. These requirements do not seen different from those in CER-CLA and the 1982 NCP; however, the rest of Section 121 marks a major policy departure regarding site cleanups.

Two basic principles governing remedial action selection are contained in Section 121(b). The first is that permanent treatment is strongly preferred over land disposal and containment. SARA states that, "Remedial actions in which treatment which permanently and significantly reduces the volume, toxicity or mobility of the hazardous substances, pollutants, and contaminants is a principal element, are to be preferred over remedial actions not involving such treatment." The phrase "actions not involving such treatment." The phrase "actions not involving such treatment." The phrase "actions not involving such treatment solutions and alternative technologies to the maximum extent practicable". How this statement is interpreted and implemented (via the NCP) by EPA directly influences the use of permanent and alternative treatment technologies at Superfund sites.

The second principle is that "off-site transport and disposal" of hazardous materials should be the "least favored alternative remedial action where practicable treatment technologies are available". This bias against off-site disposal is strengthened by Section 121(b) (2) which allows selection of an "alternative" remedial action even if never utilized at any other similar site. This seems to be an attempt to circumvent the situation associated with most "alternative" technologies, that is, a technology cannot be used unless it has been "proven"; "alternative" technologies are "unproven"; therefore, "alternative" technologies cannot be used.

Congress did not specifically define language such as "permanent", "alternative", "maximum" extent practicable", "significantly reduces...", etc., to the degree necessary to determine the "best" remediation schemes for particular sets of circumstances. However, Congress' general intent was clear: stop moving the wastes from site to site and actually start cleaning these sites up. Programmatic details were left for EPA to develop in a revised NCP.

Section 121(d) contains provisions for determining the degree of cleanup for a selected remedial action to protect human health and the environment. It provides, for hazardous substances remaining on-site, the remedial action selected must attain "legally applicable or relevant and appropriate Federal and State standards, requirements, criteria or limitations" (ARARs). In particular, cleanups "shall at least attain Maximum Contaminant Level Goals (MCLGs) established under the Safe Drinking Water Act and water quality criteria established under Section 303 or 304 of the Clean Water Act", if relevant and appropriate. Establishment of uniform cleanup standards, such as ARARs, was Congress' attempt to provide a nationally consistent "how clean is clean" policy.

Finally, section 121(e) provides that "No Federal, State or local permit shall be required for the portion of any removal or remedial action conducted entirely on-site". This resolved a long-standing controversy [12] and encourages utilization of a greater diversity of remedial technologies, a least on-site. No permit exemptions for off-site treatment are made, strengthening the position against off-site remedies.

The Remedy Selection Process and Criteria

The remedial process, detailed briefly next, has not changed much over the years, but the evaluation criteria for some of the steps have changed substantially, reflecting SARA and the evolving NCP. After a site is placed on the National Priorities List, a Remedial Investigation and Feasibility Study (RI/FS) is carried out to delineate site contamination and risks to human health and the environment, and to evaluate candidate remedial alternatives. Then, EPA issues a Record of Decision (ROD) detailing the selected remedy and the Agency's selection rationale. After issuing the ROD, a Remedial Design (RD) study is performed to provide engineering and construction details for the remedy. Finally, Remedial Action (RA) commences. Some remedies require long-term monitoring to determine remedy effectiveness, and occasionally a ROD may be reopened and amended. When a cleanup is considered complete, the site can be delisted from the NPL.

SARA (Section 105) directed EPA to revise and republish the NCP by April 1988. The proposed NCP revisions were published in December 1988 and EPA promulgated the revised NCP in 1990 [14]. The revised NCP describes how EPA intends to implement the policy mandated by SARA. Prior to the revised NCP, the only guidance published by EPA regarding site cleanups was a July 1987 interim guidance, outlining nine key cleanup criteria [15]. The document did not rank these criteria; all appeared to have equal weight.

Section 300.430, Remedial Investigation/Feasibility Study (RI/FS) and Selection of Remedy, is the heart of the NCP, with respect to selection of remedial technologies. In general, the RI characterizes the site and its risks, while the FS establishes remedial action objectives, and develops and evaluates a range of remedial alternatives. This range should include alternatives that completely destroy all hazardous substances to the maximum extent feasible, as well as alternatives employing permanent treatment that reduces the toxicity, mobility or volume of hazardous substances to varying degrees.

The last step of an FS is a detailed comparison of alternatives with respect to a number of evaluation criteria. The 1990 NCP specifies nine criteria (similar to the nine criteria outlined in the 1987 interim guidance), that encompass statutory requirements and include other gauges of overall feasibility and acceptability. The nine evaluation criteria are organized into two "threshold criteria", five "primary balancing criteria" and two "modifying criteria". The two threshold criteria must be satisfied, and include i) overall protection of human health and the environment, and ii) compliance with ARARs. The five "primary balancing criteria" are used to compare and rank remedial alternatives and include: i) long-term effectiveness and permanence; ii) reduction of toxicity, mobility or volume; iii) short-term effectiveness; iv) implementability; v) cost. The two "modifying criteria" are i) State acceptance, and ii) community acceptance, and can prove to be as important to remedy selection as the technical criteria.

The 1990 NCP implies a more structured comparison of remedial alternatives than the 1987 Guidance or previous versions of the NCP. However, the "threshold criteria" are simply restatements of SARA's statutory requirements, and since no "weighting factors" are provided for ranking remedial alternatives with respect to the "primary balancing criteria", substantial Agency discretion appears to have been preserved. This is especially true with respect to the statutory provision to utilize permanent treatment to the maximum extent practicable—if this provision is not satisfied, EPA need only explain why in the ROD. In essence, EPA is only required to consider a range of permanent treatment technologies, and at least one innovative treatment technology, in addition to the usual disposal, containment and no-action alternatives.

It should be noted that in the 1990 NCP, EPA calls for the use of MCLs, instead of the more stringent MCLGs (maximum contaminant level goals) specified in SARA. Some think EPA by-passed SARA by using the less protective MCLs [15]. However, former Assistant Administrator Porter has said, "If EPA puts out a standard (MCLs are EPA's applicable drinking water standards) for 200 million people to drink, can I really tell 10 people living next to a Superfund site that it's a different number for them? [16]." This statement has merit, especially when achieving some of the MCLGs (many of which are zero) would be exorbitantly expensive.

The SITE Program

In response to reports that Superfund site remediations rarely used permanent or alternative technologies [7, 8, 17], Congress provided for a research, development and demonstration program when reauthorizing CERCLA. SARA specifies that ten field demonstration projects of alternative or innovative treatment technologies are to be conducted each year, providing \$20,000,000 annually to fund EPA supervision and quality assurance efforts, and to fund up to 50% of the demonstration project, if warranted.

As a result of SARA's mandate, EPA developed the Superfund Innovative Technology Evaluation, or SITE, Program. The general goal of the SITE program is to "enhance the development and demonstration, and thereby establish the commercial availability, of innovative technologies at Superfund sites as alternatives to the containment systems presently in use" [18]. The SITE Program aims to achieve this goal by identifying impediments to the development and commercialization of alternative technologies and, if possible, removing these impediments by conducting alternative technology demonstrations, and developing procedures and policies to encourage selection of these alternative remedies at Superfund sites.

The SITE Program Manual [18], drawing from work done in previous impediment studies [7, 19, 20], identifies three primary categories of impediments to the acceptance and use of innovative alternative technologies. These are informational, regulatory and institutional impediments. Informational impediments include lack of adequate performance and cost data regarding specific technologies and uncertainty regarding the existence and size of the market for innovative technologies. A major focus of the SITE Program is to generate the data needed to address these informational impediments. Regulatory barriers include EPA's complex procurement procedures and the co-existence of different Federal, State and local environmental programs; the latter creates uncertainties regarding issues such as permitting and delisting. The SITE Program does not directly address regulatory impediments, because these impediments do not affect SITE demonstrations.

Institutional impediments include i) private party liability concerns, especially considering the uncertain nature of innovative technology performance; ii) governmental necessity for fund-balancing, as SARA's \$8.5 billion covers only a fraction of eventual cleanup costs; iii) community preferences for complete removal of hazardous wastes from nearby sites-the "OOMBY" (out of my backyard) syndrome; and iv) minimal involvement of academic and non-profit institutions, as these institutions are limited by SARA's 50% funding limit for SITE demonstrations. SARA addresses SITE Program private party liability concerns by allowing for some contractor indemnification during SITE demonstrations. The SITE Program also addresses community concerns through efforts to keep the public informed and allay public fears concerning innovative technologies. Non-technical issues, such as local politics and media coverage, can often influence remedial selections more than any technical issues. The other institutional impediments discussed are not addressed by SARA or the SITE Program.

Permanent, Alternative and Innovative Technologies Defined

Before assessing whether and how the use of permanent and alternative or innovative technologies has changed since SARA and the SITE Program, it is important to discuss how SARA and SITE define the terms "permanent", "alternative" and "innovative". SARA Section 121(b) (cleanup standards) does not define the difference between "permanent solutions" and "alternative treatment technologies"; however, it would not make sense to use both terms if there were no difference. The answer may lie in SARA Section 311(b) [10]; the terms "alternative or innovative technologies" are defined as "those technologies, including proprietary or patented methods, which permanently alter the composition of hazardous waste through chemical, biological, or physical means so as to significantly reduce the toxicity, mobility or volume of the hazardous waste

The underlined phrase distinguishes alternative or innovative (which are equated, here) technologies from permanent remedies. Permanent remedies are those that "significantly reduce toxicity, mobility or volume ... ", while alternative or innovative technologies are those that "significantly reduce toxicity, mobility or volume...by altering the composition of the hazardous wastes". This implies that, under SARA, alternative or innovative technologies are a subset of possible permanent remedies. For example, a containment technology can be seen as "permanent" if it reduces the mobility of contaminants, although many argue that containment can never be truly "permanent"; however, containment cannot be viewed as alternative or innovative, since the contaminated source materials, or wastes, are not altered. On the other hand, technologies such as solidification, which is considered to be a permanent and alternative (since it does "treat" the wastes) technology under the SARA and SITE definitions, is not considered by many to be a truly permanent remedy. Finally, SARA Sections 311(b) and 121(b) also imply that alternative or innovative technologies are not readily available, since they have not been demonstrated at a site or scale necessary for use at a Superfund site.

In the SITE Program, EPA seems to agree with these interpretations. The SITE Program categorizes three different levels of alternative technologies that "permanently alter the composition of hazardous wastes" [18]. The first category is available alternative technology. This is a technology, such as several forms of incineration and solidification/stabilization, that is in routine commercial or private use, especially if it qualifies under RCRA as a "best demonstrated available technology" for a particular waste stream. The second category is innovative alternative technology. This refers to a technology that is developed through the laboratory or pilot-scale, but requires fullscale testing to develop more complete cost and performance data before it is considered proven and available for routine use. The third category is emerging alternative technology. This refers to a technology that is in an early stage of development; the technology is conceptually promising, but has not passed laboratory or pilot-scale testing.

Remedial Technologies

Prior to evaluating remedial technologies, it is important to adequately define the nature and extent of site contamination, for example, what levels of which contaminants are present in particular media, and what risks do these contaminants pose to human health and the environment. Variations in contaminants, contaminant matrixes, and transport concentrations, and local human and ecological receptor patterns, make remediation of hazardous waste sites challenging and expensive. Integrated treatment process trains, consisting of several process steps, often are required to attain site cleanup goals. Since site characteristics are variable, design and operation of treatment systems must be tied to specific site conditions; thus, it is difficult to transfer treatment processes from one site to another. Site variability is one of the primary reasons EPA has retained significant discretization in the remedy selection process.

A wide variety of physical, chemical, biological, thermal and fixation/stabilization technologies are potentially applicable to Superfund wastes [23, 24]. It should be kept in mind that metals can never be "destroyed", only separated, concentrated and possibly recovered or immobilized and disposed, whereas all organic species can be "destroyed", under the right conditions, being converted into harmless residuals. Most of these technologies have been used by the chemical and petrochemical industries for decades; use for hazardous waste remediation is "technologically innovative" because application for this purpose is novel. Since each of these technologies treats wastes, they are considered alternative technologies under SARA. Whether or not they are available or innovative alternative technologies (considering SITE definitions) depends on whether they have been demonstrated at full-scale at a Superfund site, or are in routine use elsewhere, for example, under RCRA (Resource Conservation and Recovery Act).

Physical treatment utilizes physical characteristics to effect separation or concentration of waste stream constituents, generally requiring further treatment to destroy/immobilize/recover the contaminants of interest. Chemical treatment employs chemical reactions to alter the chemical composition of contaminants, that is, to convert organic species into mineralized end-products, such as carbon dioxide, and to convert metals to less toxic forms, by changing redox state. Biological treatment employs microorganisms to degrade organic species to less toxic end-products, and can be aerobic, anaerobic or a serial combination of both. Thermal treatment employs high-temperature thermal destruction of organic matter; the most important thermal technologies are the various forms of incineration. Solidification technologies are either physical or chemical processes and are important enough to warrant a separate category. These technologies immobilize toxic constituents, usually contaminated soils or sludges. Long-term effectiveness of these technologies has not been proven [25], and this is a source of contention with respect to whether or not solidification remedies should be considered permanent.

REVIEW OF REMEDIES SELECTED UNDER SUPERFUND

Permanent and Alternative Technology Use: Introduction

It is difficult to distinguish between permanent remedies, and available alternative and innovative alternative technologies, and to quantify their use at Superfund sites. For the most part, these terms have not been distinguished by EPA and most reviewers; most reporters of the use of permanent remedies do not discuss whether these permanent remedies use alternative technologies (available or innovative) to treat contaminated source materials. The EPA Fiscal Year ROD Annual Reports [22] summarize use of permanent remedies at Superfund sites for particular years, but do not state whether these remedies use alternative technologies to treat the contaminated source materials and do not provide details on how data were generated.

Furthermore, although it is relatively easy to tally the use of alternative technologies over the years, it is far more difficult to determine whether the use of an alternative technology constitutes a permanent remedy for the following reasons: i) there are some alternative technologies, such as solidification, that many do not consider to be permanent remedies, ii) in many cases an alternative technology may only be applied to a small amount of a site's contaminated source materials, leaving the permanence of the overall site remedy in question, and iii) many sites engender multiple RODs, pertaining to multiple operable units (discrete areas), complicating determination of remedial permanence. Therefore, it is easier to report which RODs use alternative remedial technologies, that is, those that treat contaminated source materials, and leave open the interpretation of whether the alternative technology used constitutes a permanent remedy.

Considering the preceding discussion, it is difficult to establish more than general trends regarding patterns in use of permanent and alternative technologies under Superfund. However, it is possible to delineate three significant periods in Superfund history with respect to remedy selection, corresponding to important milestones in the evolution of the Statute and the NCP. The first is 1980-1984, during which containment and offsite disposal were almost always the selected remedies. The second period is 1985-1986, during which the 1985 NCP was published and CERCLA was being reauthorized. During this period, the selection of alternative technologies increased, in response to public and Congressional pressure, and in anticipation of SARA directives. The third period is 1987 to the present, subsequent to SARA, during which the use of alternative technologies accelerated even further; however, the performance of these remedies is disputed, as will be discussed.

Table 1 summarizes RODs selecting alternative remedial technologies at Superfund sites through 1990 (some RODs selected multiple alternative technologies, but the ROD is only counted once for this table). This tables summarizes data from the authors' review of every ROD abstract through 1990, updating the authors' previous review of ROD abstract sthrough 1988 [26]. Interestingly, EPA's Technology Innovation Office (TIO) recently performed a similar survey, and discrepancies between the authors' data and the TIO data are minor [27].

Technologies considered alternative for this table are either available alternative technologies or innovative alternative technologies. Available alternative technologies include incineration and solidification, while, in general, all other alternative technologies are considered innovative alternative technologies—these include soil washing, soil flushing, thermal desorption, in-situ and on-site bioremediation, vapor extraction, dechlorination, vitrification, wetlands application and waste recycling. The Table also includes estimates of the percentage of alternative technologies that could be considered available or innovative.

Permanent and Alternative Technology Use: 1980-1984

A review of the ROD abstracts for this period (see Table 1) indicates that 6 of 55 RODs (11%) selected alternative remedial technologies; the performance of these remedies is question-

Table 1 Sele	ection of Alternati at Superfur	ve Remedial nd Sites	Technologies
Year	*RODs Selecting Alternative Tech	Total RODs	% RODs Selecting Alternative Tech
1980-1984	6	55	11%
1985	17	70	24%
1986	27	84	32%
1987	28	77	36%
1988	68	153	44%
1989	75	144	52%
1990	89	168	53%
Total	310	751	41%

*Of the 310 RODs selecting alternative technologies, ~42% selected innovative alternative technologies, and ~58% selected available alternative technologies (incin. or solid.)

According to the GAO report [17], of the ten permanent remedies selected during this period, incineration was chosen three times, solidification was chosen three times, and groundwater pumping and treatment accounted for the other four permanent remedies. Under SARA and the SITE program, incinceration and solidification would be considered examples of permanent, available alternative treatment technologies. They would be considered alternative in that they alter the composition of the hazardous substances. Further, they would be considered available (especially incineration), since they were in routine use by industry for RCRA hazardous wastes [28]. However, as has been discussed, many critics feel that solidification is not a permanent remedy as it only immobilizes the hazardous substances contained in source materials and should be termed enhanced containment. These data also show that no RODs selected innovative alternative technologies during this period.

There is also disagreement over whether groundwater pumping and treatment is either permanent or alternative. Unlike the GAO report [17], the 1987 EPA report [22] did not consider groundwater pumping followed by treatment to be a permanent remedy; thus, EPA took credit for permanent treatment remedies for only 6 of 55 RODs (11%) during this period. Interestingly, EPA seems to contradict itself in the 1991 TIO report [27], in which it considers groundwater pumping and treatment to be a permanent remedy, employing "source control", distinguishing these remedies from alternative technologies that treat contaminated source materials. However, none of the reports considers groundwater pumping and treatment to be an alternative technology since it only addresses control/collection of waste streams, not contaminated source materials. Note that, groundwater pumping and treatment should be distinguished from in-situ soil flushing followed by treatment, which directly addresses contaminated source materials, accelerating site cleanup relative to pumping and treatment.

Generally, the reports acknowledged that for the RODs counted as selecting permanent remedies only a small portion of the total cleanup was actually considered permanent. It is doubtful whether these remedies would satisfy the later SARA requirement for selecting permanent and alternative technologies to the maximum extent practicable. As an example, at the Burnt Fly Bog site in NJ, wastes in drums were to be incinerated off-site for \$50,000, while the rest of the site materials were to be excavated and disposed off-site at a cost of \$8,000,000. Even so, this site was given credit for selection of a permanent technology.

Permanent and Alternative Technology Use 1985-1986

The years 1985 and 1986 were transition years for the Superfund program, with respect to implementation of permanent and alternative technologies. Our review of the ROD abstracts indicates that in 1985, 17 of 70 RODs (24%) selected alternative treatment technologies for contaminated source materials and in 1986, 27 of 83 (33%) RODs did so. According to the 1991 EPA-TIO report [27], in 1985, 19 of 70 RODs (27%) selected alternative treatment technologies and in 1986, 27 of 83 (33%) RODs did so. Finally, the 1987 EPA report [22] indicated that in 1985, 15 of 70 RODs (21%) selected permanent remedies and, in 1986, 27 of 83 (33%) RODs did so. [Remember that this report did not differentiate between permenant remedies and alternative technologies, although it is likely that the report refers to alternative technologies, since it did not include groundwater pumping and treatment as a kind of permanent remedy.] All of the these reports show marked improvement compared to 1980–1984, probably due to pressure resulting from general dissatisfaction with the cleanup record, coupled with the 1985 NCP revisions and Superfund reauthorization momentum.

It is important to note that great difficulty is involved in reporting the use of permanent and alternative treatment technologies, as well as comparing different reports in this area. For example, reference [13] claims that, in 1985, "some form of treatment" was used in seventy-one cases and, in 1986 "some form of treatment" was used in eighty-four cases. In this report, "some form of treatment" includes groundwater pumping and treatment, and multiple treatment units per site are included. These data might mistakenly lead to the conclusion that permanent remedies were being selected more frequently than was true.

For the 44 RODs selecting alternative treatment in 1985-1986 (17 in 1985 and 27 in 1986), incineration was chosen twenty times, solidification/stabilization was chosen thirteen times, soil aeration (to volatilize organics from the soil) was chosen seven times, soil flushing was chosen five times, and soil-based biodegradation was chosen two times. These total forty-seven because EPA selected multiple possible remedies for three RODs in 1986. Incineration and solidification would be considered available alternative technologies. However, soil aeration, soil flushing and soil-based biodegradation had never been used at a Superfund site and were not in routine use by industry. Using SITE Program definitions, the latter are examples of the use of innovative alternative technologies; thus innovative alternative technologies accounted for 29% of the alternative technologies selected.

One hundred-five RODs signed during 1985-1986 did not select alternative treatment technologies. No-action, off-site disposal, and on-site containment were the principal remedies selected; about half of these RODs employed some form of groundwater pumping and treatment, to prevent off-site migration of contaminants at the site [17]. While in 1980-1984 most of the RODs selecting alternative treatment selected it to clean up only a small portion of site, for 1985-1986, the alternative technologies selected addressed significantly more contaminated materials than in 1980-1984 [12].

Permanent and Alternative Technology Use: 1987-1990

Starting with FY 1987, that is, October 1986, all RODs were to be performed in accordance with SARA, which was enacted in October 1986. As discussed, SARA directs EPA to favor the selection of permanent treatment technologies over land disposal and containment, and to select permanent and alternative treatment technologies to the maximum extent practicable.

A review of the ROD abstracts during this period (see Table 1) shows a continued increase in selection of alternative technologies: in 1987, 28 of 87 RODs (36%) selected alternative treatment technologies for contaminated source materials; in 1988, 68 of 153 RODs (44%) selected alternative treatment technologies; in 1989, 75 of 144 RODs (52%) selected alternative treatment technologies, and in 1990, 89 of 168 RODs (53%) selected alternative treatment technologies. Overall, 260 of 542 RODs (48%) selected alternative treatment technologies during this period. These figures are within a few percent of those in the 1991 EPA-TIO report [27]. Once again, of the RODs not selecting alternative technologies, most selected no action, offsite disposal or on-site containment; many of these also selected groundwater pumping and treatment.

Of the 260 RODs during this period that selected alternative treatment technologies, close to 50% selected innovative alternative technologies. This represents a significant increase in the use of innovative alternative technologies relative to the period prior to SARA and the SITE Program. Innovative alternative technologies selected included soil washing, soil flushing, thermal desorption, *in-situ* and on-site bioremediation, vapor extraction, dechlorination, vitrification, wetlands application and waste recycling. The remaining alternative treatment RODs selected available treatment technologies—incineration and solidification.

When does an innovative technology becomes an available alterntive technology-is it after the remedial action is completed, or only after the technology has been in operation long enough to generate reliable data? At this point, soil aeration, soil flushing and soil-based degradation, in particular, would probably still be considered innovative alternative technologies, even though they were selected at other sites, during 1985-1986. This is because the remedies at these sites have not been completed, and performance and cost data are still incomplete. Also, application of these technologies at Superfund sites has been much less extensive than application of incineration and solidification/stabilization technologies, which are generally considered as "available". However, EPA still considers certain types of incineration and solidification/stabilization technologies to be at the innovative alternative technology development stage, since many variations of these technologies were accepted into the SITE demonstration program [27].

Since 72% of all RODs signed through 1990 were signed since SARA, this period is an especially important one in Superfund history. Although this review and EPA's-TIO report [27] show that nearly half of all RODs since SARA selected alternative treatment technologies, there is much disagreement over whether EPA actually fulfilled the mandate for permanent remedies. The report "Right Train, Wrong Track" (RTWT), published by a combination of environmental and industrial interests, is highly critical of EPA's performance since SARA [15]. The report concludes that EPA has not fulfilled SARA mandates regarding permanent remedies and is continuing to rely primarily on disposal and containment remedies.

The RTWT report found that i) 68% of the remedies selected in FY 1987 failed to treat contaminated source materials; ii) 24% selected partial or ineffective treatment remedies; and, iii) only 8% used treatment technologies to the maximum extent practicable, as required by SARA. RTWT does not consider solidification/stabilization to be a permanent remedy, and it points to several remedies that only treated a small fraction of source materials that were credited as permanent. The report blames weak Federal management, especially continued use of cost and effectiveness as equal partners in the remedy selection process. The report concludes by saying that it is diffcult to imagine a more specific or better funded statute than SARA, but that SARA cannot be made to work when EPA lacks the motivation to make it work.

Another report, OTA's "Are We Cleaning Up?", [29] echoes RTWT findings regarding selection of permanent remedies, since SARA. This report does not provide as broad a summary as RTWT, but focuses on ten detailed case studies to illustrate programmatic shortcomings. The case study of the Crystal City, Texas site illustrates some of these shortcomings.

At Crystal City, the selected remedy for soil contaminated with high concentrations of pesticides (including DDT) and metals was the excavation of the soils and disposal on-site in an unlined, capped landfill. The on-site landfill did not comply with applicable RCRA requirements for landfill design. No treatability study supported the conclusion that this remedy, based on adsorption and immobilization of the contaminants by site soil, was permanent. Permanent alternatives, such as incineration, soil washing/treatment, and thermal separation were rejected; according to OTA the reason was cost. This type of problem is common according to OTA. The primary reasons for this are i) too much flexibility and lack of central management control; ii) emphasis on detailed results instead of evaluation of what the results accomplish; iii) lack of information regarding permanent technologies, focus on shortterm costs; and, iv) Agency inability to hire and retain competent staff.

Given the discrepancies between the two reports just discussed [15, 29] and the EPA reports on the use of permanent and alternative technologies [22, 27], it is difficult to determine to what extent SARA directives have been satisfied. These discrepancies led to severe disagreement between EPA personnel and the authors of the two critical reports [30]. Reality probably lies somewhere between the divergent views presented. While it is unarguable that the use of alternative treatment technologies (both innovative and available) has greatly accelerated since SARA, it is probably true that the use of permanent remedies has improved since SARA, but not as much as EPA has claimed.

Finally, it is too early to tell what, if any, impact the 1990 revisions to the NCP have had on remedy selection. However, the 1987 interim guidance and the 1988 draft NCP anticipated the 1990 final NCP, and were probably partly responsible, in conjunction with the SARA statute itself and the SITE Program, for the acceleration in the consideration and selection of permanent and alternative remedies during this period.

Permanent and Alternative Technology Use: Other Considerations

Even if consistent and reliable cost and performance data were available for hazardous waste treatment technologies and EPA made a stronger effort to select permanent and alternative remedial technologies, increased use of these technologies would not be guaranteed. The primary reason is that permanent and alternative treatment remedies are more expensive than containment and monitoring remedies, and may not be feasible for thousands of Superfund sites. For the 1987 RODs selecting alternative treatment, the average estimated remedial costs were in excess of \$15,000,000 per site [22]. More recent estimates indicate an average remedial cost of \$30,000,000 per site [32].

As has been discussed, many of these remedies were not truly permanent remedies, which would have been even more expensive. Furthermore, the final number of NPL sites could be 10,000 or more [7], meaning it might eventually require 300-5700 billion to permanently clean up all of these sites. If funding were maintained at current levels, it would require decades to permanently remediate all Superfund sites. Worse yet, it has been estimated that it may eventually cost several hundred billion dollars for remediation of Federal waste sites from the Departments of Defense and Energy. Additionally, the future funding level for Superfund and Federal sites remediations cannot be predicted.

A recent GAO report [33] stated that "it is doubtful that the Superfund Program, as currently operated, will ever clean up all hazardous waste sites". It is not encouraging that after ten years and \$11 billion spent, EPA considers cleanup work to be completed at only 60 of the 1,2000 NPL sites [32]. Finally, important regional and global environmental problems, such as ozone depletion, climate change, habitat destruction and resource depletion, have emerged over the last few years, lessening the relative importance of hazardous waste site cleanup in the eyes of many. The discussion above is not an argument against the use of permanent and alternative technologies, but rather an acknowledgement that EPA must better prioritize among remedial activities, as well as better prioritize remedial activities among the suite of other environmental problems demanding attention.

SITE Program Performance

Through 1991, the SITE Program has gone through six technology solicitations, and there are currently 58 technologies in the innovative alternative technology demonstration

Environmental Progress (Vol. 11, No. 3)

program with 30 demonstrations completed [31]. For the SITE demonstration program, accepted technologies include innovative incinerator/thermal treatment technology designs, innovative solidification/stabilization technologies, innovative biological processes, innovative soil extraction/washing technologies, innovative physical separation technologies, and more. For the 30 field demonstrations completed at Superfund sites, about a dozen final reports have been released [31].

Results from the SITE Program indicate that most of these technology demonstrations have been successful in generating important baseline performance and cost data, and many have been successful in achieving desired remedial goals. Of the 58 technologies involved in the demonstration program, several have been selected as remedial technologies in about 14 subsequent RODs (some technologies have been selected multiple times) [31]. In addition, many more of these 58 technologies are under consideration for future RODs. Thus, it seems that the SITE Program has been reasonably successful in achieving its goals, especially given its modest funding; currently the SITE Program accounts for less than 1% of the Superfund budget; in contrast, most technological companies spend 5-10% of their budgets on research and development.

CONCLUSIONS

The Superfund remedy selection process has changed considerably over the history of the Superfund program, as reflected by trends in the selection of permanent remedies and alternative remedial technologies. There have been three significant periods in Superfund history with respect to remedy selection, corresponding to important milestones in the evolution of the Superfund statute and regulations. The first is 1980-1984, during which containment and offsite disposal were almost always the selected remedies; during this period, only 11% of 55 total RODs selected alternative treatment technologies. The second period is 1985-1986, during which the 1985 NCP was published and CERCLA was being reauthorized. During this period, the selection of alternative technologies increased to 29% of 154 total RODs, in response to public and Congressional pressure, and in anticipation of SARA directives. The third period is 1987 through 1990, subsequent to SARA, and during which the SITE program commenced, and the NCP was revised. During this period, the use of alternative technologies accelerated even further, to 48% of 542 total RODs. SARA (and the NCP) directives promoting consideration and selection of permanent remedies and alternative technologies, as well as the SITE Program's success in demonstrating alternative technologies, were instrumental in this acceleration. However, although acceleration in the selection of alternative technologies is indisputable, many critics have questioned the degree of performance of many of the remedies that employ alternative technologies.

ACKNOWLEDGMENTS

We would like to thank Drs. Merry Morris, Michael Greenberg and David Kosson for their help in guiding the dissertation research effort that led to this paper. We would also like to thank John Martin, from EPA's SITE Program, for his insight into alternative technologies and the SITE Program.

LITERATURE CITED

- 1. 42 U.S.C. Sections 9601-9675.
- Glass, E. A., "Superfund and SARA: Are There Any Defenses Left?", *Harvard Environmental Law Review*, 12(385), 385-463, 1988.
- 3. Harris, R. H. and G. C. Wrenn, "Making Superfund

Work," Issues in Science and Technology, 54-58, Spring, 1988.

- Anderson, F. R., D. R. Mandelker and A. D. Tarlock, Environmental Protection: Law and Policy, Little, Brown and Co., Boston, 1984.
- 47 RF 31180, et seq., July 16, 1982, codified at 40 CFR 300.
- U.S. Environmental Protection Agency, Superfund Comprehensive Accomplishments Plan, Washington, D.C., December, 1984.
- U.S. Congress, Office of Technology Assessment, Superfund Strategy, Washington, D.C., 1985.
- U.S. General Accounting Office, Status of EPA's Remedial Cleanup Efforts, Washington, D.C., 1985.
- U.S. General Accounting Office, Cleaning Up Hazardous Wastes: An Overview of Superfund Reauthorization Issues, Washington, D.C., 1985.
- Superfund Amendments and Reauthorization Act of 1986, Pub. L. No. 99-499, 100 Stat. 1613, 1986.
- Edelson, J. L., "Superfund: Still in the Dumps," Technology Review, 27-32, Nov/Dec 1988.
- Arbuckle, J. G., et al., *Environmental Law Handbook*, 9th Edition, Government Institutes Inc., Rockville, MD, 1987.
- White, D. C., et al., "Summary of Hazardous Waste Treatment at Superfund Sites," BNA Environmental Reporter, 1121-1129, August 21, 1987.
- U.S. Environmental Protection Agency, Fact Sheet for Revisions to the National Oil and Hazardous Substances Pollution Contingency Plan, Washington, D.C., Nov. 1988.
- Environmental Defense Fund, Hazardous Waste Treatment Council, National Audubon Society, National Wildlife Federation, Natural Resources Defense Council, Sierra Club, U.S. PIRG, Right Train, Wrong Track: Failed Leadership in the Superfund Cleanup Program, June 20, 1988.
- Stanfield, R. L., "Stewing Over Superfund," National Journal, 19, 2030–2032, August 8, 1987.
- U.S. General Accounting Office, Hazardous Waste: EPA's Consideration of Permanent Cleanup Remedies, Washington, D.C., July, 1986.
- U.S. Environmental Protection Agency, Superfund Innovative Technology Evaluation (SITE) Strategy and Program Plan, Washington, D.C., December, 1986.

- U.S. Environmental Protection Agency's Science Advisory Board, Environmental Engineering Committee Report to the Administrator on Superfund Demonstration Program, Washington, D.C., 1986.
- U.S. General Accounting Office, EPA's Innovative Technology Program for Wastewater Treatment Needs Better Controls, Washington, D.C., August, 1984.
- Dienemann, E. A., Greenberg, M., and Ahlert, R. C., "Remediation of the Lipari Landfill, The Nation's #1 Ranked Superfund Site," *Impact Assessment Bulletin*, Terre Haute, IN, In Press.
- U.S. Environmental Protection Agency, ROD Annual Reports for FY 1987, 1988, 1989 and 1990, Washington, D.C.
- Freeman, H. M. (ed), Standard Handbook for Hazardous Waste Treatment and Disposal, Academic Press, New York, 1989.
- 24. U.S. Environmental Protection Agency, A Compendium of Technologies Used in the Treatment of Hazardous Wastes, Washington, D.C., September, 1987.
- Weitzmann, L., L. E. Hamel, and E. Barth, "Evaluation of Solidification/Stabilization As A Best Demonstrated Available Technology," USEPA Fourteenth Annual Research Symposium, May, 1988.
- Dienemann, E. A., Policy and Technical Issues in Remediating Superfund Sites, Ph.D. Dissertation, Rutgers University, New Brunswick, NJ, May, 1989.
- U.S. Environmental Protection Agency, Technology Innovation Office, *Innovative Treatment Technologies:* Semi-Annual Status Report, Washington, D.C., Sept., 1991.
- U.S. Congress, Office of Technology Assessment, Technologies and Management Strategies for Hazardous Waste Control, Washington, D.C., March, 1983.
- U.S. Congress, Office of Technology Assessment, Are We Cleaning Up?: 10 Superfund Case Studies, Washington, D.C., June, 1988.
- "Superfund Official Confronts Critics, Calls Negative Reports Unfair, Misleading?", BNA Environment Reporter, 419-420, July 29, 1988.
- 31. Various Personal Communications with John Martin, USEPA SITE Program Demonstration Coordinator.
- 32. "Superfund Mess", New York Times, Business Section, New York, NY, June 16, 1991.
- 33. U.S. General Accounting Office, *Environmental Protection* Agency Issues, Washington, D.C., November, 1988.

Critical Evaluation of PCB Remediation Technologies

Laura J. Amend and Peter B. Lederman

Roy F. Weston, Inc., Raritan Plaza 1, Raritan Center, Edison, NJ 08837

Polychlorinated biphenyls (PCBs) are a family of aromatic compounds consisting of a biphenyl molecule which has been chlorinated to form a large number of possible congeners. The number of chlorine atoms they can contain (from 1 to 10) varies, as does the placement of the chlorine atoms on the biphenyl molecule. Thus, a polychlorinated biphenyl is one of 209 compounds having the formula: $C_{12}H_{10-n}Cl_n$. Due to their excellent heat stability, fire retardance and electrical insulating properties, they gained widespread use as dielectric fluid incorporated in electrical transformers and capacitors. They were also incorporated in products as various as plastics, hydraulic fluids, surface coatings, inks, paints, pesticides, lubricants, and carbonless copy paper.

INTRODUCTION

From 1929 through 1980, the world production of PCBs was about 2.4 billion pounds. It is estimated that in the United States, about 404 million pounds of PCBs are still in use and accessible, about 262 million pounds are still in use and generally not accessible, and about 585 million pounds have been destroyed or are in storage, landfills, or the environment. The PCBs still in use and accessible are in transformers, capacitors, and other electrical equipment. Within the last category are PCBs in sediments, soil, vegetation and animals, the atmosphere, and fresh water totalling about 24 million pounds, another 13 million pounds are in the oceans, about 385 million pounds are in landfills and other storage, and an estimated 162 million pounds have been degraded.

In the mid 1960's, concerns regarding the toxicity and persistence of PCBs in the environment led to their eventual control in the United States under the Toxic Substances Control Act of 1976 (TSCA). TSCA regulates the manufacturing, processing, distribution in commerce and use of PCBs (40 CFR 761 et seq.). Subpart G of 40 CFR 761, the PCB Spill Cleanup Policy, specifies "how clean is clean" for spills of PCBs which occur after the effective date of the policy (May 4, 1987). The PCB Spill Cleanup Policy does not apply to several categories of spill situations, including historic spills (those which occurred prior to the policy's effective date), spills which directly contaminate or migrate into surface waters, sewers or sewer treatment systems, private or public drinking water sources or distribution systems, and spills which contaminate animal grazing lands or vegetable gardens. For spills which fall in any of these categories, a cleanup standard must be negotiated with the cognizant regulatory agency(s). Determination of an acceptable residual concentration is imperative prior to selecting a remedial technology for treatment of the specific waste stream(s).

The primary basis of establishing a cleanup standard is the protection of public health and the environment. However, the ability to reach a cleanup level in a given situation is also considered. Thus, current knowledge of the capabilities of remedial technologies is vital. Development of PCB treatment technologies continues due to the ubiquitous nature of the chemical. This paper discusses the current state of development of the available technologies.

INCINERATION

Incineration is a recognized technique of PCB destruction. TSCA requires a "six-nines" (99.9999%) destruction and removal efficiency (DRE) of PCBs for thermal treatment systems. Therefore, for an incineration system to be permitted to destroy PCB containing wastes, it must be able to demonstrate conformance with this strict standard. The high temperatures and long residence times usually required to conform to this standard generally destroy all other organic components of the waste as well.

There are less than ten TSCA permitted commercial PCB incinerators currently operating in the United States. These vendors indicate that PCB destruction costs range from approximately \$500 to \$800 per ton for solid waste and \$3.25 to \$5.00 per gallon for liquid waste. Costs are dependent upon waste characteristics such as heating value and burn characteristic of the waste, PCB concentration, waste matrix, and total amount of waste treated, among other factors.

In some instances, it may be more economically feasible to bring a TSCA permitted incinerator to the waste, rather than transport the waste to a commercial incinerator. However, public acceptance of on-site incineration may be limited. Early regulatory agency and public involvement prior to selection of this remedial alternative may help alleviate the historical "NIMBY" (not in my backyard) attitude against thermal destruction units.

Transportable incinerators are comprised of the same unit operations as permanent, commercial facilities, namely: primary and secondary combustion chambers, feed systems, bottom and fly ash handling units, air emissions control (filters, electrostatic precipitators, wet scrubbers), heat exchangers, and exhaust stacks. Waste water treatment for cooling/scrubbing water recycle/disposal and a mobile on-site laboratory can be integral supporting operations for transportable incineration systems.

Roy F. Weston, Inc., among others, operates a TSCA permitted Transportable Incineration System (TIS) which utilizes a 23 million BTU/hr rotary kiln as the primary combustion chamber. The kiln is of standard countercurrent cylindrical design which allows retention times of fifteen to ninety minutes. Operating temperatures can be varied below 1200 to 2200°F and the system can process from two to ten tons per hour of contaminated soils. It is also designed to be capable of incinerating fume, hazardous liquids and sludges at rates up to 233 million BTU/hr of heat release. A demonstration test performed in May 1988 indicated that the system could process an average of 13,650 lb/hr of soil spiked with PCB oil to an average concentration of 14,740 parts per million (ppm). The DREs achieved during this demonstration ranged from 99.99983% to 99.999990%, with residual PCB concentrations in the bottom ash of .01 ppm to .15 ppm.

The feed conveyor system requires a maximum particle size of two inches, therefore, materials handling must assure this limit can be met. For loose, sandy soil, simple screening can accomplish this end. Soils with high cohesion will require an operation, such as shredding, which will break down the feed to the appropriate size. As with all incineration systems, inorganic (heavy metal) constituents are not destroyed, but passed through to the ash streams. Stabilization of these streams may be required to immobilize the inorganic components to acceptable levels.

The TSCA permit granted to the TIS eliminates the need to get additional Federal TSCA permits as the unit is moved from site to site. However, National Pollutant Discharge Elimination System (NPDES), Publicly Owned Treatment Works (POTW) discharge, and/or air emissions permits may still be required. Additionally, if the waste feed can also be characterized as Resource Conservation and Recovery Act (RCRA) hazardous waste, a RCRA Treatment Storage and Disposal Facility (TSDF) permit must be gained.

The cost of remediating PCB contaminated soils via the TIS can range from \$140 to \$600 per ton; and generally fall in the \$180 to \$200 per ton range. These costs are mainly dependent on the heating value of the waste, the concentration of PCBs to be destroyed, and the total quantity of waste to be remediated.

Ogden Environmental Services owns and operates a transportable 10 million BTU/hr Circulating Bed Combustor (CBC). The CBC can provide DREs of $99.9999\%_0$, and is available for on-site operation. Soil feed rates vary based on soil moisture content and heating value of the waste. Solids, liquids and sludges may be treated utilizing this system, however, a maximum particle size of $1^{"}$ for solid feed will determine the precombustion material preparation and handling requirements.

The thermal treatment unit of the CBC consists of ceramic lined combustion loop which is comprised of a combustion chamber and cyclone. Waste is fed into the loop along with limestone, which neutralizes acid gases formed during the thermal destruction reactions. The waste contacts hot recirculating soil from the cyclone. As it heats, it is entrained with air and travels upward through the combustor into the cyclone, where the solid fraction is separated from the combustion gases. The hot solids drop down to reenter the combustion chamber, where they will aid in heating new feed being introduced. Remediated solids remaining in the lower portion of the combustion chamber are removed at a controlled rate by an ash conveyor system.

Like the TIS, the CBC is TSCA permitted. However, air emissions, water discharge and RCRA permits must be obtained as required. Treatment costs for remediating soils by CBC typically range from \$100 to \$300 per ton, depending primarily on soil moisture and quantity of material to be processed.

Westinghouse Environmental Services has developed and markets a Pyroplasma plasma arc process which utilizes a plasma torch to pyrolyze liquid organic materials at temperatures up to 15,000 °C. Throughput capacities vary from one to three gallons per minute in single stage units. Waste liquids are aspirated into a superheated gas stream, where the molecular bonds of the waste are broken into their component atoms. These atoms recombine in the reaction chamber to form nonhazardous gases such as carbon monoxide, nitrogen, hydrogen, and some low molecular weight hydrocarbons. Acid gases and particulate carbon are removed in a wet scrubber. The resultant gaseous stream, primarily hydrogen and carbon monoxide, is drawn off and flared.

The Pyroplasma unit is suitable for liquid wastes only. Under a TSCA Research and Development Permit, it has demonstrated DREs greater than 99.9999% for PCB liquids with concentrations of 70-80% PCBs by weight. This technology is available in mobile units. Permitting and public acceptance issues for this treatment system are the same as those for the TIS and CBC.

A transportable Infrared Incineration System has been designed by Shirco Infrared Systems, Inc., which utilizes electrically powered silicon carbide rods to provide infrared energy that heats organic waste to combustion temperatures. Remaining combustibles are incinerated in an afterburner. A demonstration of this system in 1987 under the United States Environmental Protection Agency's (EPA's) Superfund Innovative Technology Evaluation (SITE) program demonstrated its ability to destroy PCBs at concentrations ranging from 10 ppm to 669 ppm in the feed with DREs exceeding 99.99%. While this demonstration did not meet TSCA's requirement for PCB destruction by thermal treatment, other tests with the same unit have achieved DREs of 99.9999%. Therefore, this incineration system, based on waste specific trial burn results, may be granted TSCA approval for remediation of PCB contaminated soils.

IN-SITU VITRIFICATION

A variation of traditional thermal treatment systems in the In-Situ Vitrification (ISV) Technology developed by the Pacific Northwest Laboratories Division of Battelle Memorial Institute and marketed exclusively by Geosafe Corporation. ISV is a process which melts inorganic soil for the purpose of thermally destroying its organic component. This is accomplished as follows.

Electrodes are placed in the soil to the desired treatment depth. A conductive starter path is placed on the surface between electrodes, and the treatment area is covered by an offgas collection hood. As a result of the electric potential between the electrodes, current which flows through the starter path generates heat which in turn melts the underlying soils. The molten soils are generally quite electrically conductive, allowing the process to continue to depth. The high temperatures within the molten soil cause organic material (either naturally occurring or contaminants) to volatilize and pyrolyze. The resultant gases either become dissolved in the melt or move to the surface, where they combust on contact with air. Products of pyrolization and combustion are collected in the off-gas collection hood, and are treated prior to emission.

As the soils melt, the void space is removed. A volume reduction of 20-40% is typical. When the melt mass reaches an optimum, power to the electrodes is removed and the soil is allowed to cool naturally. As it cools, it vitrifies into a monolith which resembles naturally occurring obsidian (vol-

canic glass). The surface depression which results from the reduction in volume may be backfilled with clean soils, allowing for unrestricted future use of the area upon approval by the cognizant regulatory agency.

Potentially toxic inorganic contaminants are chemically incorporated in this matrix. Leach, hydration, and weathering tests of vitrified waste indicate that it has excellent long-term stability and can expect a mean life of thousands of years.

The vendor indicates this process exceeds PCB DREs of 99.9999% as required by TSCA, but requires treatability testing on actual waste samples to verify the process is capable of satisfying cleanup requirements. They also indicate this method is applicable to a variety of organic, inorganic and radioactive materials.

The ISV process is an in-situ treatment, and the treated vitrified waste may be left in place. As such, materials handling requirements are minimized. While the process can operate in fully saturated soils, the energy required to vaporize water could more cost effectively be used in vitrifying soils. Organic and inorganic debris may be left in place and treated within the melt volume, dependent on their expected effect on offgas generation and residual product.

This technology is commercially available through Geosafe, which holds worldwide exclusive rights to it. Costs for utilizing this technology are on a similar order of magnitude as incineration, but can vary greatly based on variables such as wastesoil properties, volume of material, depth of required vitrification, unit price of electricity, moisture content, and distance of the site from the vendor's base location in Washington.

DESTRUCTION USING QUICKLIME

At several EPA remedial action sites in the recent past, observations by regional staff had suggested that treatment with quicklime (CaO) may achieve destruction of PCB contaminated soils and sludges.

EPA's Risk Reduction Engineering Laboratory (RREL) initiated studies, both in-house and through a contracted laboratory (RMC Environmental and Analytical Laboratories) to verify the results indicated in the field. The initial study conducted by RMC were extremely limited in scope, and indicated that reductions in PCB concentrations, on the order of 90%, were achieved in the waste matrices. The in-house project designed by RREL and conducted by their on-site contractor, Technology Applications Inc., indicated that the reductions detected in the waste matrices were due to evaporation and steam stripping in PCBs during the lime shaking process. Furthermore, only minimal amounts of products of phenyl-phenyl bond cleavage were detected, further supporting the theory that this process is not applicable to the destruction of PCBs.

BIOREMEDIATION

Biological treatment technologies use microorganisms to detoxify or decompose degradable organics. Naturally occurring (native) species may be utilized or specially adapted microorganisms may be introduced into the contaminated media.

Biodegradation of PCBs as a remedial technology is in the developmental stages. Many studies have identified several microbes which exhibit an ability to degrade specific PCB congeners to varying degrees. These microorganisms include several species of Achromobacter, Alcaligenes, and Pseudomonas, as well as a fungus, Phanerochaete chrysosporium. Best results can be obtained in biological reactors which can insure the necessary aeration, mixing, temperature control and nutrient delivery to optimize the biokinetics.

Biological treatment systems are living systems. Survival of the necessary microbes is vital for this technology to be feasible. It is therefore necessary to conduct treatability testing on actual waste samples to determine if the microorganisms can degrade the specific PCB congeners present or even survive. While the inherent toxicity of PCBs indicates that their biodegradation would occur only at dilute concentrations, laboratory studies performed by General Electric Company, Corporate Research and Development have indicated degradation of soils spiked with PCBs in concentrations as high as 500 ppm Aroclor 1242.

The extremely limited amount of data on field applications of biological processes to remediate PCB contaminated wastes makes it difficult to specify levels of detoxification possible in nonlaboratory conditions. Likewise, it is not possible to quantify chemical or matrix interferences which may be encountify chemical or matrix interferences which may be encountiered, or the costs to be expected on a per unit basis. However, bioremediation is gaining favor in hazardous waste remediation in both Europe and the United States. As it is utilized, more data will be made available for assessing the efficiency of the technology for a myriad of contaminants, including PCBs. Specifically, the results of a field study of biodegradation in PCB contaminated Hudson River sediments, being conducted by General Electric, will make field application data available for future bioremediation schemes.

CHEMICAL DECHLORINATION

In 1978 at the Chemistry and Biosciences Laboratory, Franklin Research Center, a chemical reagent was synthesized and found to destroy PCBs by the process of dechlorination. Since that time, a family of alkali metal polyethylene glycolate (APEG) reagents have been developed which effectively dechlorinate PCBs in soils and liquids quickly. These reagents are known as KPEG or NaPEG, dependent on whether the alkali metal utilized is potassium or sodium. Research has shown that KPEG is more effective in destroying PCBs than NaPEG.

In a patented process, the KPEG reagent is prepared by reacting potassium hydroxide with one of several polyethylene glycols. This reaction produces an alkoxide which reacts with one of the chlorines on the aryl ring of the PCB to produce an ether (hydroxyl substituted aryl ring) and potassium chloride. The substitution of the chlorine atom with the etherlinked polyethylene glycol molecule detoxifies the PCB. Additional research indicated this process can also dechlorinate (and thereby detoxify) polychlorinated dibenzodioxins (PCDDs) and polychlorinated dibenzofurans (PCDFs) in either a solid or liquid matrix.

This process and variations of this process have undergone numerous laboratory studies and field demonstrations. A field demonstration conducted by EPA RREL in Guam on PCB contaminated soils utilized a batch reactor equipped with a mixer and an external steam jacket to allow for process heating. The steam jacket was designed to allow it to serve as a process cooling water system as well. A vent from the reactor to a condensate collection system was also provided. Following dechlorination the treated soil was allowed to cool. Sulfuric acid was then added to adjust the pH to the range of 6 to 9. Treated wastes with unacceptable concentrations of PCBs were subsequently re-treated to provide further dechlorination. Soil was discharged from the reactor, sampled and stored pending analytical results. Over several runs, approximately 15 cubic yards of soil with an average concentration of 3,500 ppm PCBs were treated to greater than 99% reduction of PCBs.

Waste soils pretreatment requirements (that is, largest particle size), are highly dependent on the physical configuration of the mixer employed. For the Guam demonstration, a maximum particle size of 1/2 inch was required to prevent fouling of the mixer blades against the reactor wall.

Implementation of this process utilizes commercially available process equipment, therefore is readily available. However, implementation of this process may be slowed by the need to obtain TSCA, RCRA, and potentially air and water discharge permits. Additionally, care must be taken so as not to infringe on the patent holder's legal rights.

The cost to apply this technology will vary greatly depending upon the volume of treated waste and the current reagent prices. Cost estimates for treating soil in plants with a treatment capacity of 25 tons per batch operation have been estimated to range from \$242 to \$347 per ton of soil, with capital-related and chemical costs accounting for approximately 84 percent of total cost.

Several vendors offer dechlorination of PCB articles (transformers with PCB oil at concentrations greater than 500 ppm) and PCB contaminated equipment (transformers filled with dielectric fluid at concentrations between 50 and 500 ppm). The systems utilized are generally flatbed trailer mounted and operate using a closed loop circulation system. This allows for reclamation of transformer oil on-site, without draining it from the transformers. For a cost premium, the operation can be performed on energized transformers, reducing downtime. Proprietary reagents are utilized to dechlorinate, dewater, degassify and desludge transformer oils.

Properly utilized under the vendors' TSCA permit, these types of procedures allow for declassification of PCB and PCB contaminated electrical equipment. This in turn should reduce future liability for the electrical equipment owners, since equipment with less than 50 ppm PCBs is less likely to release harmful quantities of PCBs to the environment.

The specific nature of the proprietary reagents will determine any potential chemical and/or matrix interferences. Vendors need to be contacted for the costs of these operations. Factors which will affect the cost of treatment include type of unit to be serviced, the volume and concentration of dielectric fluid, access to equipment, and whether it will be energized or deenergized.

SOLVENT EXTRACTION

Solvent extraction of PCBs from soils makes use of PCBs' preferential solubility in oils and various extractants over water. HARBAUER, Sanivan Group, and Resources Conservation Company (RCC) have all developed processes when PCB contaminated oily soils or sludges mixed in an extraction vessel (generally a countercurrent soil washer or mixed reactor) with the appropriate solvent. During this step, the PCBs preferentially enter the oily solvent phase, leaving the soil phase with a greatly reduced PCB concentration. The solvent phase is decanted, and the soil phase is dried if necessary. The oily solvent fraction is then separated using standard separations techniques into oil, water, and solvent fractions. Ideally, solvent can be regenerated or recycled for future use, and all PCBs are concentrated within the oil fraction. These processes do not destroy PCBs, but rather, concentrate them into one phase of the original waste stream with a greatly reduced volume as compared to the original waste. The PCBs in the concentrated stream, generally the oil fraction, must then be either thermally or chemically destroyed as required by regulation.

HARBAUER and RCC have reported removal efficiencies on the order of 98 percent with oily soils contaminated with PCBs in the less than 20 ppm range during actual field scale on-site remediations. Sanivan Group will be conducting a field demonstration of their process under the auspices of EPA's SITE program on soils contaminated with PCBs in the range of 5 to 50 ppm.

Feed preparation requirements for solvent extraction processes are determined, in part, by the physical configuration of the extraction vessel. Maximum particle size and percent fines in the soil/waste matrix will likely be a factor for soil washing techniques, since a higher ratio of fines in the waste results in a greater volume of sludge requiring final disposition.

Potential chemical and matrix interferences will depend on the specific extractant and process equipment utilized. These processes have been designed for treatment of contaminated soils, but modifications may be possible to allow for treatment of liquid phase wastes as well.

Soil treatment by solvent extraction is currently commercially available. Availability of the concentrated effluent must also be determined to judge overall implementability of these processes. As with all on-site treatment options it is recommended that the appropriate regulatory agency(s) be notified to expedite the approval/permitting process prior to implementation.

Costs are very site/waste specific. A potentially significant portion of the overall cost of treatment is the cost for destruction of the concentrated PCB waste stream. Vendors state that application of this technology is less expensive and more cost effective than incineration, but did not quote figures.

STABILIZATION

Chemical stabilization is a treatment process used to immobilize hazardous waste constituents in a solid matrix through mixing the original waste stream with additives and binders (that is, cement). For treating contaminated soil and sediments, the process generally involves excavation and subsequent conversion of the matrix to a solid mass which immobilizes leachable contaminants. An identical process can be applied to stabilize residual sludge streams with leachable contaminants resulting from various other treatment methods. Stabilized materials are typically sent to permitted landfills for disposal. Stabilization of soils may be applied in situ by adding stabilizing agents via a deep soil mixing auger, although effectiveness may vary. Stabilization reduces the mobility of contaminants, but does not destroy them. Therefore, the risk of exposure to humans or the environment resulting from leaching contaminants at low rates over the long term still exists.

As with all on-site treatment technologies, a determination of the permits required for remediation should be sought from the applicable regulatory agency(s).

Another limitation to stabilization of wastes is the inhibiting effect that organic constituents generally have on the curing process and final quality of the stabilized waste. Several vendors have developed proprietary additives which, when blended with cement, water and the contaminated soils, can overcome the deleterious effect of the organic constituents in the waste.

HAZCON, Inc. has performed a successful field demonstration of immobilizing low level (50 to 80 ppm) PCBs in soils in the presence of up to 25 percent by weight of oil and grease using a cement based process with their proprietary additive, Chloranan. Using a 10:10:1 ratio of waste:cement:Chloranan produced a solidified mass in which analysis of leachate could not detect PCBs at a detection limit of 1 ug/L. A volume increase of 120 percent was observed, and cost of \$250 per ton for the demonstration was calculated.

International Waste Technologies (IWT) has developed their own proprietary soil stabilization additive, HWT-20. IWT has performed a demonstration of in-situ stabilization of soils contaminated with PCBs in concentrations up to 950 ppm (with most of the samples under 300 ppm) utilizing HWT-20 and a deep soil mixing technique developed by Geo-Con, Inc. Results of this demonstration, conducted as part of EPA RREL's SITE program, indicate that the process may immobilize PCBs. However, due to the low concentrations of PCBs in the treated soils (after dilution resulting from addition of cement and HWT-20), absolute confirmation of this tentative finding could not be proven. Additionally, due to the very limited amount of volatile organics detected in the waste matrix, it is unable to predict if HWT-20 can counteract their effect on the final product. Furthermore, while unconfirmed compressive strength, permeability and wet/dry weathering tests indicated the solidified mass could maintain its integrity over a long period of time, unsatisfactory water loss was observed during the freeze/thaw weathering tests. As compared to HAZCON's process, volume increase (8.5 percent) and cost (\$194 per ton) observed in the IWT/Geo-Con process were moderate. However, due to uncertainties in the quality of the final product, further studies should be conducted prior to commercial application of this process.

CONCLUSIONS

When selecting a technology to treat PCB contaminated wastes, it is necessary to take into account the applicability of the technology to the specific waste stream. Availability and reliability of the technology is also of vital importance. As delineated in this paper, there are currently several technologies (for example, incineration, chemical dechlorination and waste water treatment technologies) which have proven their ability to destroy PCBs and for which a significant amount of applications data exists. There are several others which can be considered emerging technologies-they have been proven technically viable, yet limited field applications data reduces the ability to effectively judge the feasibility of utilizing these technologies to treat specific waste streams. These emerging technologies include in-situ vitrification, solvent extraction, surface decontamination, and stabilization. Finally, biodegradation and destruction using quicklime are developmental technologies. A significant amount of study will be required to determine if these technologies will prove effective in destruction of PCBs.

LITERATURE CITED

- 1. Ahmed, M., and D. D. Focht, "Degradation of Polychlorinated Biphenyls by Two Species of Achromobacter," Can. J. Microbiol., 19: 47-52 (1972).
- 2. Brunner, W., F. H. Sutherland, and D. D. Focht, "Enhanced Biodegradation of Polychlorinated Biphenyls in Soil by Analog Enrichment and Bacterial Inoculation," J. Environ. Qual., 14: 324-328 (1985).
- 3. Chynoweth, E., "New Options Take on Incineration," Chem. Week, 147: 49-50 (August 22, 1990).
- 4. Clark, R. R., E. S. K. Chian, and R. A. Griffin, "Degradation of Polychlorinated Biphenyls by Mixed Microbial Cultures," Appl. Environ. Microbiol., 37: 680-685 (1979).
- 5. Einhaus, R. L., Honarkhah, I., and Erickson, P., "Fate of Polychlorinated Biphenyls (PCBs) in Soil Following Stabilization With Quicklime," EPA/600/2-91/052 (1991).
- 6. Erickson, M. D., Analytical Chemistry of PCBs, Butterworth Publishers, Boston, MA, p. 508 (1986).
- 7. Evans, B., Ed., "Bioprocessing: White Rot Fungus Fights Toxic Waste," *Mech. Eng.*, 3: 80-83 (1989). 8. Fouhy, F., and A. Shanley, "Mighty Microbes," *Chem.*
- Eng., 98: 30-35 (1991).
- 9. Fox, C. A., "Status and Trends in Bioremediation Treatment Technology," Remediation, 1: 293-303 (1991).
- 10. Geosafe Corporation, "Application and Evaluation Considerations for In-Situ Vitrification Technology: A Treatment Process for Destruction and/or Permanent Immobilization of Hazardous Materials," Product Literature, GSC, 1901 (1989).
- 11. Joseph, Dr. M. F., and W. H. Reed, "Pyroplasma Process for Organic Waste Destruction," Westinghouse Environmental Services Product Literature.
- 12. Kane, J. E., and J. V. Mehta, "Cleanup and Closure of a PCB Contaminated Pond," Poll. Eng., 18: 28-30 (1986).
- 13. Lauber, J. D., "Disposal and Destruction of Waste PCBs," PCBs and the Environment, J. S. Weid, Ed., CRC Press, Inc., Boca Raton, FL, Vol. III, pp. 83-151 (1987).

- 14. Leuser, R. M., L. A. Velazquez, A. K. Cohen, and J. Janssen, "Remediation of PCB Soil Contamination by On-Site Incineration," presented at the American Chemical Society National Meeting (April 1989).
- 15. Liu, D., "Enhancement of PCBs Biodegradation by Sodium Ligninsulfonate," Water Res., 14: 1467-1475 (1980).
- 16. McDermott, J. B., R. Unterman, M. J. Brennan, R. E. Brooks, D. P. Mobley, C. C. Schwartz, and D. K. Dietrich, "Two Strategies for PCB Soil Remediation: Biodegradation and Surfactant Extraction," Env. Prog., 8: 46-51 (1989).
- 17. Nielson, R., "Applications of Thermal Technologies for Decontaminating Soil," The Weston Way, Roy F. Weston, Inc., Product Literature (1989).
- 18. Resources Conservation Company, "Hazardous Waste Cleanup, The B.E.S.T. Solution from Resources Conservation Company," Product Literature, 1988.
- 19. RMC Environmental & Analytical Laboratories, Interim Report on the Disappearing PCBs Project (1991).
- 20. Savage, P., "Cleaning Up PCBs On-Site," Chem. Week, 140: 13-14 (1987).
- 21. Sawyer, S., and M. K. Stinson, "In-Situ Stabilization/ Solidification of PCB-Contaminated Soil," EPA/600/D-89/119 (1989).
- 22. Snider, E. H., "PCBs and DEHPs in Wastewater," Poll. Eng., 19: 48-49 (1987).
- 23. Sonnen, Dr. H. D., W. Groschel, and M. Nels, "Experience with the HARBAUER PB3 Soil Cleaning System," Presented at USEPA Forum on Innovative Hazardous Waste Treatment Technologies: Domestic and International (1989).
- 24. Soundararajan, Dr. R., Final Report on the "Disappearing PCBs" Project, RMC Environmental Analytic Laboratories (1991).
- 25. Stinson, M. K., "EPA SITE Demonstration of the International Waste Technologies/Geo-Con In-Situ Stabilization/Solidification Process," J. Air Waste Manage. Assoc., 40: 1569-1576 (1990).
- 26. Sudell, G. W., Project Summary Evaluation of the B.E.S.T. Solvent Extraction Sludge Treatment Technology Twenty-Four Hour Test, EPA/600/S2-88/051 (1988).
- 27. Sunohio Company, PCBX Product Literature (1985).
- 28. Taylor, M. L., J. A. Wentz, M. A. Dosani, W. Gallagher, and J. S. Greber, Project Summary Treating Chlorinated Wastes with the KPEG Process, EPA/600/S2-90/005 (1990).
- 29. U.S. EPA SITE Technology Demonstration Summary, Technology Evaluation Report, SITE Program Demonstration Test, HAZCON Solidification, Douglassville, Pennsylvania, EPA/540/S5-89/001 (1989).
- 30. U.S. EPA SITE Technology Demonstration Summary, Technology Evaluation Report, Shirco Pilot-Scale Infrared Incineration System at the Rose Township Demode Road Superfund Site, EPA/540/S5-89/007 (1989).
- 31. U.S. EPA SITE Fact Sheet, Proposed Demonstration of the Sanivan Group Extraksol Solvent Extraction Technology, Pinette's Salvage Yard Superfund Site, Washburn, Maine (1991).
- 32. Wilbourn, R. G., and B. M. Anderson, "Contaminated Soil Remediation by Circulating Bed Combustion," Presented at U.S. EPA Forum on Innovative Hazardous Waste Treatment Technologies: Domestic and International (1989).
- 33. Woodyard, J. P., and J. J. King, "Recent Technology Developments for PCB Destruction and Oil Recycling, Presented at DOE Oak Ridge Model Conference (1987).
- 34. Woodyard, J. P., and E. M. Zoratto, "State-of-the-Art Technology for PCB Decontamination of Concrete," Presented at IEEE Conference on PCBs and Replacement Fluid (Motech '86) (1986).

Granular Activated Carbon Pilot Treatment Studies for Explosives Removal From Contaminated Groundwater

W. J. Wujcik, W. L. Lowe, P. J. Marks

Roy F. Weston, Inc., 1 Weston Way, West Chester, PA 19380-1499

and

W. E. Sisk

USATHAMA, CETHA-TE-D, Edgewood Area, APG, MD 21010-5401

Manufacturing activities at Army Ammunition Plants (AAPs) result in the production of organic wastewaters that contain both explosive residues and other organic chemicals. As a result of past waste practices at such plants, explosive residues may leach through the soil and contaminate groundwater. Two pilot studies were performed to evaluate the use of granular activated carbon (GAC) to treat groundwater contaminated with explosives at Badger AAP and Milan AAP. An additional goal of the Badger AAP study was to examine the potential discharge of explosives 2,4-DNT and 2,6-DNT from a packed column air stripper used to remove volatile organic compounds from groundwater. A laboratory method was developed for the BAAP study to permit lower detection levels for 2,4-DNT and 2,6-DNT (0.46µg/L and 0.017µg/L, respectively). The studies concluded that removal of explosives from groundwater using continuous flow GAC is feasible.

INTRODUCTION

The United States Army operates explosives plants that load, assemble, and pack (LAP) explosives into military ordnance. Activities at such plants produce process wastewaters that contain both explosives residues and other organic chemicals. Several treatment technologies have been developed and are currently in use to treat these wastewaters for final discharge.

Past waste handling practices at explosives LAP plants often utilized unlined lagoons or pits to contain process wastewaters. As a result of this practice, some explosives residues have leached through the soil to contaminate groundwater. Therefore, groundwater treatment may be required. Based upon process wastewater treatment experience, potentially applicable treatment technologies are available. However, the similarities and differences between process wastewaters and explosives-contaminated groundwater should be considered before transferring technologies from one application to an other.

Process wastewaters at explosives LAP plants are often treated by activated carbon adsorption. This treatment has been documented in the literature [1, 2, 3, 4]. Therefore, based

upon process wastewater treatment experience, activated carbon adsorption would likely work for the treatment of explosives-contaminated groundwater. However, because of the similarities and differences between process wastewaters and explosives-contaminated groundwater, the feasibility of using activated carbon adsorption for treatment of groundwater should be determined.

Hinshaw et al. [5] present a multiphase study providing quantitative data on the ability of activated carbon to remove the nitrobodies 2,4,6-trinitrotoluene (TNT), 2,4-dinitrotoluene (2,4-DNT), cyclotrimethylenetrinitramine (RDX), and cyclotetramethylenetetranitramine (HMX) from pink water. The study included preliminary activated carbon screening, isotherm tests of activated carbon, preliminary column tests of selected activated carbons, four-in-series column tests, and an economic analysis of activated carbon. Isotherm tests on five different carbons were conducted to select the best performing carbon for further testing using continuous-flow columns. Pilot-scale column tests were subsequently performed with four columns in-series using one of the five carbons for the treatment of actual ammunition plant pink water. Effluent criteria (40 μ g/L TNT, 30 μ g/L RDX, 30 μ g/L HMX, and 0.7 μ g/L 2,4 DNT) were generally met for RDX, HMX, and 2,4-DNT, but not for TNT. The TNT performance limitation was determined to be a physicochemical phenomenon. This phenomenon did not appear during the isotherm tests and points to the importance of performing actual column tests with the wastewaters to be treated.

Groundwater in the area of the Propellant Burning Grounds at Badger Army Ammunition Plan (BAAP) in Baraboo, Wisconsin, has been found to be contaminated with explosives compounds, including dinitrotoluene isomers, volatile organic compounds (VOCs), and related degradation products [6]. A barrier well network to intercept the advancing contaminant plume with associated treatment using air stripping and/or granular activated carbon (GAC) has been proposed.

Explosives-contaminated water has percolated from the Oline ponds at the Milan Army Ammunition Plant (MAAP) in Milan, Tennessee, into the upper and middle part of the Clairborne aquifer underlying the facility. Migration of these wastes is expected to continue in the groundwater flow system and thereby contaminate additional groundwater and possibly surface water in the area. The chemical wastes that are of major concern at MAAP are 2,4,6-trinitrotoluene (TNT), cyclotrimethylenetrinitramine (RDX), cyclotetramethylenetetranitramine (HMX), 2,4,6-trinitrophenylmethylnitramine (Tetryl), 2,4-dinitrotoluene (2,4-DNT), 2,6-dinitrotoluene (2,6-DNT), 1,3-dinitrobenzene (I,3-DNB), 1,3,5-trinitrobenzene (1,3,5-TNB), and nitro-benzene (NB).

The primary objective of these pilot studies was to determine the feasibility of using GAC to treat explosives-contaminated groundwater. The explosive contaminants studies were TNT, RDX, HMX, Tetryl, 2,4-DNT, 2,6-DNT, 1,3-DNB, 1,3,5-TNB, and NB. The studies included preliminary batch (isotherm) testing followed by column testing using a continuousflow pilot plant. Since groundwater contamination at BAAP includes VOCs, an air stripper was used in conjunction with GAC treatment. The secondary objective of the pilot study at BAAP was to examine the potential for discharge of explosive components from the air stripper.

ISOTHERM TEST PROGRAM

Laboratory GAC isotherm studies were conducted to evaluate the ability of activated carbon to remove 2,4-DNT and 2,6-DNT from groundwater at BAAP and TNT, RDX, HMX, Tetryl, 2,4-DNT, 2,6-DNT, 1,3,5-TNB, and NB from the explosives-contaminated groundwater at MAAP. Five different types of granular activated carbon were evaluated in both studies in terms of: (1) relative adsorbability of explosives; (2) adsorption capacities and associated exhaustion rates for explosives; (3) degrees of removal based on desired effluent objectives; and (4) preferential adsorption of component groups. The isotherm studies also evaluated the effect of an acidic pH (4.0) on adsorption capacities.

Isotherm testing consisted of a series of batch adsorption experiments in which multiple aliquots of groundwater from BAAP and MAAP were treated with varying dosages of GAC. The test containers were agitated to achieve equilibrium between the liquid phase and the solid phase. The GAC was then filtered out of the solution and the filtrate was analyzed to determine the equilibrium concentration of the parameters (or adsorbate) of interest. The data obtained from the analysis were interpreted by plotting the amount of adsorbate absorbed per unit weight of activated carbon versus the equilibrium concentration of adsorbate remaining in solution.

Groundwater samples from BAAP received additional treatment prior to isotherm testing. They were aerated for 1 hour and 15 minutes (to simulate air stripping) to strip off volatiles present in the groundwater. The pHs of the groundwater samples before aeration were in the range of 7.0 to 7.5. After aeration, the pH of the groundwater samples had increased to range from 8.3 to 8.7. Sulfuric acid (H_2SO_4) was added to the aerated groundwater samples to lower the pH to required levels prior to contacting the groundwater with carbon.

ISOTHERM TEST RESULTS

The Freundlich adsorption equation was used in presenting the carbon isotherm test results. Isotherms were developed for BAAP for 2,4-DNT and 2,6-DNT by plotting the adsorption data on logarithmic coordinates as carbon loading (q_e) versus the equilibrium concentration (C_e) of compound remaining in the groundwater sample. The empirical constants of the Freundlich equation for the five test carbon isotherms are presented in Table 1.

On the basis of adsorption capacities for 2,4-DNT and 2,6-DNT, Filtrasorb 300 and Hydrodarco 4000 were judged to be the best performing GACs for removing both contaminants from groundwater at BAAP. The maximum saturation capacities (theoretical maximum loading) for Filtrasorb 300 and Hydrodarco 4000 were estimated by extrapolating the isotherms to $(q_e)C_0$. The carbon loading thus obtained corresponds by definition to a condition when all the carbon is in equilibrium with the influent concentration (C_0) . In a carbon column treatment system, this equates to operating a GAC system until the concentration of a particular compound at the column effluent equals the influent concentrations. The equilibrium q_e values at different influent concentrations of 2,4-DNT and 2,6-DNT for the five GACs are presented in Table 2.

Therefore, as a result of this isotherm testing program, Filtrasorb 300 and Hydrodarco 4000 were selected for further testing, using continuous-flow GAC columns at BAAP.

The empirical constants of the Freundlich adsorption equation for the five test carbon isotherms for MAAP are presented in Table 3 for TNT and RDX. These constants for the three Filtrasorb carbons could not be determined for TNT because it was present at concentrations below its detection limit in all of the filtrates. For this same reason, the empirical constants of the Freundlich adsorption equation for all five carbons for the other seven explosives could not be determined.

The maximum saturation capacities (theoretical maximum loading) for all five carbons for RDX were estimated by ex-

Table 1 Empirical Constants of Freundlich Adsorption
Equation" for Five GACs Using Groundwater from Mon
itoring Well PBN82-02C at the Badger Army Ammunition
Plant

	Fiant			
Activated Carbon Type	2,4- Isotl	DNT	2,6- Isot	DNT herms
	K ^b	l/n ^c (slope)	K ^b	l/n ^c (slope)
Filtrasorb 200	0.085	0.077	0.03	0.022
Filtrasorb 300	0.075	0.067	0.09	0.086
Filtrasorb 400	$(0.9)^d$	(2.72)	0.09	0.086
Hydrodarco 3000	0.02	0.014	0.03	0.029
Hydrodarco 4000	0.2	0.263	0.035	0.024

 $^{a}q_{e}=X/M=KC_{e}^{l/n}.$

where.

 q_e = Carbon loading, mg compound/mg carbon, dimensionless.

 $\ddot{X} = C_0 - C_e$, the amount of compound adsorbed for a given volume of solution, mg/L.

M = Carbon dosage, mg/L.

 C_0 = Initial concentration of compound, mg/L.

 C_e = Concentration of compound remaining in solution, mg/L.

K = Freundlich constant (mg/L)^{-1/n}.

n = Empirical constant, dimensionless

^bIntercept at $C_e = 1.0 \text{ mg/L}$ on the isotherm line. This intercept was determined by extrapolation.

Slope of the line within the concentration range of 0.01-1.0 mg/L.

^dBy extrapolation from the maximum equilibrium concentration of 0.06 mg/L (obtained at the lowest carbon dosage of 10 mg/L) to a concentration of 1.0 mg/L.

Table 2 Maximum Saturation Capacities (q.) for Five GACs Tested for at the Badger Army Ammunition Plant

		Influent	Saturation
		Concentration	C_0 Capacity (q_e)
Carbon Type	Nitrobody	(mg/L) ^a	(mg/mg)
	2,4-DNT	2	0.1
T'll		10	0.21
Filtrasoro 300	2,6-DNT	1	0.09
	-	4	0.21
	2,4-DNT	2	0.28
11. 1 1		10	0.62
Hydrodarco 4000	2,6-DNT	1	0.03
	8	4	0.07
	2,4-DNT	2	0.10
E'll		10	0.22
Filtrasoro 200	2,6-DNT	1	0.03
		4	0.06
	2,4-DNT	- 2	b
E'L 1 400		10	b
Filtrasorb 400	2,6-DNT	1	0.09
		4	0.21
	2,4-DNT	2	0.02
M. J. J. J. J. 2000		10	0.03
Hydrodarco 3000	2,6-DNT	1	0.03
	1410 - 1419 - 1200-1400 - 1200	4	0.12

Influent concentrations correspond to those anticipated from previous work (10 mg/L for 2,4-DNT and 4 mg/L for 2,6-DNT) and to those actually found in the groundwater sample used for isotherm tests reported here (2 mg/L for 2,4-DNT and 1 mg/L for 2,6-DNT). ^bDue to an elevated detection limit caused by sample interference, the exact

value for the data point corresponding to this carbon dosage could not be determined.

trapolating their isotherms to C_0 . This gives a q_e that corresponds to a condition in which all the carbon is in equilibrium with the initial concentration of RDX. In a continuous-flow GAC column, this condition occurs when the RDX concentration in the effluent is the same as that in the influent. However, this does not occur in practice, because normally

Table 3 Empirical Constants for Freundlich Adsorption Equation[®] for Five Test Carbons Using Groundwater from Monitoring Well MI051 at the Milan Army Ammunition Plant

	TI Isoth	NT herms	RDX Isotherms	
Activated Carbon Type	K ^b	l/n ^c	K ^b	l/n ^c
Filtrasorb 200	ND ^d	ND ^d	0.052	0.535
Filtrasorb 300	ND^{d}	ND^{d}	0.031	0.413
Filtrasorb 400	ND^{d}	ND^{d}	0.049	0.555
Hydrodarco 4000	0.128	0.828	0.0012	0.100
Atochem, Inc.	0.136	0.642	0.045	0.630
GAC 830				

 $^{a}q_{e}=X/M=KC_{e}^{1/n}.$

where,

 q_e = Carbon loading, mg compound/mg carbon, dimensionless.

 $\ddot{X} = C_0 - C_e$, the amount of compound adsorbed for a given volume of solution, mg/L.

M = Carbon dosage, mg/L.

 $C_0 =$ Initial concentration of compound, mg/L.

 C_e = Concentration of compound remaining in solution, mg/L. K = Freundlich constant (mg/L)^{-1/n}.

n = Empirical constant, dimensionless

^bIntercept at $C_e = 1.0 \text{ mg/L}$ on the isotherm line. This intercept was determined by extrapolation.

'Slope of the line within the concentration range of 0.01-1.0 mg/L.

^dND = Not determinable.

the column service is terminated when the effluent concentration reaches a predetermined effluent limit. In addition, maximum saturation capacities for Hydrodarco 4000 and Atochem, Inc. GAC 830 for TNT were estimated by extrapolating their isotherms to C_0 .

Because of inconclusive results on the effect of pH on the activated carbon adsorption of the groundwater from monitoring well M1051 at MAAP, the continuous-flow GAC columns at MAAP were run at pH 7.0. This decision was supported by the results on the effect of pH described in the previous pilot study at Badger Army Ammunition Plant (BAAP) [7]. In that study, the results of the isotherm tests for Filtrasorb 400, using explosives-contaminated groundwater from monitoring well PBN82-02C at BAAP, showed that relatively higher adsorption capacities were obtained at neutral pH (7.0) than at acidic pH (4.0). Even though previous results in the literature [3] showed the opposite to be true for TNT and other nitroaromatics, greater weight was given to the BAAP report because it represented the most recent experience with explosives-contaminated groundwater.

PILOT STUDIES

Activated Carbon Pilot-Scale Test at BAAP

The continuous-flow column testing was conducted at BAAP using the two types of carbons selected from the batch (isotherm) testing. A schematic of the pilot plant, including the air stripper, is shown in Figure 1. Two continuous-flow column tests were performed, each using two carbon columns.

For each test, the total groundwater flow was split between the two test trains (A and B) at different proportions depending on the desired hydraulic loading rates. The two columns in Train A and Train B contained carbon types designated as A and B, respectively.

All of the tests were conducted inside the test building located at BAAP. The test area was maintained at a minimum temperature of 50°F. The hydraulic surface loading rates that were employed during the pilot tests were 3, 5, and 7 gpm/ft². These values are within the range of hydraulic surface loading rates that are normally used in full-scale operation of GAC systems.

Adsorption and breakthrough characteristics were studied in the first column of each parallel pair. The function of the second column was to maintain effluent (discharge) quality within acceptable limits while allowing contaminant leakage up to influent levels (total exhaustion of capacity) of the first column. Influent and effluent (first column as well as second column) were monitored for pH and temperature at regular time intervals.

An air stripper was used to remove volatile solvents from the groundwater prior to its entering the GAC pilot plant. The primary function of the air stripper in this project was to minimize any effects of these solvents on the evaluation of activated carbon adsorption of explosives. The potential for discharge of explosives into the air stream was examined by emissions testing of the exhaust stack. Sampling points were located approximately 6 feet above (downstream from) the



FIGURE 1. BAAP Pilot Treatment Plant Configuration

	Test One	1 1000	Test Two	1 1000	Test Three	
	Starting Date: 7 Sep	otember 1989	Starting Date: 19 S	eptember 1989	Starting Date: 16 O	ctober 1989
	Column Inner Dian	neter: 4.25 in.	Column Inner Diar	neter: 4.25 in.	Column Inner Dian	neter: 4.25 in.
	0.354 ft)	e.	0.354 ft)	U Support of the second s	0.354 ft)	
	Column Area: 0.098	85 ft ²	Column Area: 0.09	85 ft ²	Column Area: 0.09	85 ft ²
	Bed Volume: 0.394	ft ³ (2.94 gal.	Bed Volume: 0.394	ft ³ (2.94 gal.)	Bed Volume: 0.197	ft ³ (1.47 gal.)
	Column A1	Column B1	Column A1	Column B1	Column A1	Column B1
AC Used	Atochem, Inc.	Atochem, Inc.	Atochem, Inc.	Calgon	Atochem, Inc.	Calgon
	GAC 830	GAC 830	GAC 830	Filtrasorb 300	GAC 830	Filtrasorb 300
ow Rate	1.0 gpm	0.2 gpm	0.7 gpm	0.7 gpm	0.75 gpm	0.75 gpm
ydraulic Loading	10.15 gpm/ft ²	2.03 gpm/ft ²	7.11 gpm/ft ²	7.11 gpm/ft^2	7.6 gpm/ft^2	7.6 gpm/ft^2
ed Depth	4 ft	4 ft	4 ft	4 ft	2 ft	2 ft
mpty-Bed Contact Time	3.0 min.	14.7 min.	4.2 min.	4.2 min.	2.0 min.	2.0 min.
nal Effluent Levels (μ g/L)						
NT (Influent 433 µg/L)	ND	ND	ND	ND	88.6 µg/L	192 µg/L
DX (Influent 487 $\mu g/L$)	$7.05 \ \mu g/L$	ND	ND	ND	$315 \mu g/L$	344 µg/L
MX (Influent 3.4 μ g/L)	ND	ND	ND	ND	ND	IJ
etryl (Influent ND)	ND	ND	ND	ND	ND	ND
4-DNT (Influent 10.8 μ g/L)	ND	ND	ND	ND	$3.6 \mu g/L$	ND
6-DNT (Influent ND)	ND	ND	ND	ND	ND	ND
3-DNB (Influent 3.2 µg/L)	ND	ND	ND	ND	ND	ND
3,5-TNB (Influent 21.4 μ g/L)	ND	ND	ND	ND	9.91 µg/L	14.9 µg/L
B (Influent ND)	ND	ND	ND	ND	ND	ND
un Time	7.6 days	7.6 days	16.5 days	16.5 days	54.5 days	54.5 days

expansion point in order to eliminate any effects of flow disturbances caused by the expansion. Samples from the exhaust stack were analyzed for explosives components (2,4-DNT and 2,6-DNT). The adequacy with which the air stripper removed volatiles was verified by comparing GAC pilot-plant influent levels to those in the untreated groundwater.

Samples for explosives analysis were collected from the air stripper exhaust duct using an EPA Modified Method 5 (MM5) sampling train. The train was further modified by the inclusion of an additional XAD-2 resin trap (a total of two) to ensure complete collection of target explosives. Sampling was conducted along the horizontal axis of the 8-inch inner diameter duct. The number of sample points and the sampling duration were determined on-site, with a 4-hour sample time used at four traverse points. Sampling was isokinetic (90 to 110%). Volumetric flow rate was determined by EPA Methods 1 and 2 as part of the MM5 train. Moisture content was determined using the EPA MM5 sample train used for collection of explosives. Triplicate test repetitions were performed for each emission parameter.

A mobile field laboratory was used to provide rapid turnaround of the large number of samples generated during pilot testing. Samples were analyzed in the field laboratory for dinitrotoluenes by liquid/liquid extraction and electron capture gas chromatography (GC), utilizing USATHAMA Method UW01. The following modifications to Method UW01 were made for field use:

- 1. Use injection volume of 5 mL.
- 2. Calibration curves of $1 \times$, $5 \times$, $10 \times$, $50 \times$, and $100 \times$.
- 3. Daily QC of blank and $10 \times$ spike.
- Final 10× calibration standard.

Detection limits were determined by instrument sensitivity. In order to verify treatability at very low contaminant influent and effluent concentrations, WESTON's Analytics Division obtained certification from USATHAMA for a lowlevel DNT analytical method employing liquid/liquid extraction and electron capture detector QC analysis. The detection limits established in this certification effort were 0.46 μ g/L for 2,4-DNT and 0.017 μ g/L for 2,6-DNT. Following certification, this method was used for explosives analyses at the WESTON Analytics Division laboratory. The WESTON Analytics Division laboratory data were used to verify the field data.

Samples were analyzed by U.S. EPA Method 8010 for carbon tetrachloride, chloroform, trichloroethylene, and 1,1,1trichloroethane. Detection levels of 1 ppb were obtained with Method 8010.

Activated Carbon Pilot-Scale Test at MAAP

The continuous-flow column testing at MAAP was conducted using the two types of carbon selected from the batch (isotherm) testing. A schematic of the pilot plant is shown in Figure 2. The air stripper was not used at MAAP because volatiles were not detected in the groundwater used for the pilot test. During the field test program at MAAP, three continuous-flow column tests were performed, each using two carbon column trains. The operating and performance data for Tests One, Two, and Three are presented in Table 4.

Test One was run for a total of 7.6 days, at which time it was decided by WESTON and USATHAMA that breakthrough would not be reached in a reasonable amount of time for study purposes using the operating parameters in Table 4. Except for RDX, none of the contaminants of concern in this table were detected in the effluent samples during Test One. RDX was detected in column A1's effluent sample, but at a concentration that was approximately 1% of its influent concentration.

Test Two was run for a total of 16.5 days, at which time it



FIGURE 2. MAAP GAC Pilot Plant Configuration

was decided by WESTON and USATHAMA that breakthrough would not be reached in a reasonable amount of time for study purposes using the operating parameters in Table 4. None of the contaminants of concern in this table were detected in the effluent samples during Test Two.

Test Three was run for a total of 54.5 days, at which time, because of cold weather, it was decided by WESTON and USATHAMA that the unit should be shut down so that no damage would occur to the system. Of the contaminants of concern in Table 4, only TNT, RDX, 2,4-DNT, and 1,3,5-TNB were detected in the effluent samples during Test Three.

The analytical parameters for all tests were TNT, RDX, HMX, Tetryl, 2,4-DNT, 2,6-DNT, 1,3-DNB, 1,3,5-TNB, and NB. A total of 750 samples was analyzed by high-performance liquid chromatography (HPLC) for these parameters at the field laboratory at MAAP, utilizing USATHAMA Method UW01, Explosives in Waters. The samples included influent samples to the GAC unit and effluent samples from columns A1, A2, B1, and B2. The field laboratory was critical in providing rapid turnaround, facilitating operating decisions, and protecting effluent quality. In addition, 42 samples were sent to WESTON's Analytics Division for similar analysis. These samples represented 6% of the field laboratory samples analyzed and verified field laboratory performance.



FIGURE 3. 2,4-DNT Concentration Using Early Morning Data (0600–0800) from Field Laboratory Results for Run One (Starting 15 February 1989) at BAAP





DISCUSSION OF RESULTS

At the influent concentrations of groundwater from monitoring well PBN-82-02C at BAAP, influent surface loading rates of 3.0 to 7.0 gpm/ft² and an influent hydraulic loading rate of 1.5 to 3.5 gpm/ft³, GAC columns employing either Filtrasorb 300 or Hydrodarco 4000 can provide run lengths of at least 16,130 gallons (10,970 bed volumes) while providing 2,4-DNT and 2,6-DNT removals of greater than 90%. Under conditions employed in this study, explosives concentrations could be reduced below detection limits (in this study, approximately 0.46 $\mu g/L$ for 2,4-DNT and 0.017 $\mu g/L$ for 2,6-DNT) for approximately 4,120 gallons (2,800 bed volumes) at the highest loading rate for 2,4-DNT and for approximately 180 gallons (123 bed volumes) at an intermediate loading rate for 2,6-DNT. Figures 3 through 6 summarize experimental results.



FIGURE 5. 2,4-DNT Concentration Using Early Morning Data (0600-0800) from Field Laboratory Results for Run Two (Starting 8 March 1989) at BAAP



FIGURE 6. 2,6-DNT Concentration Using Early Morning Data (0600-0800) from Field Laboratory Results for Run Two (Starting 8 March 1989) at BAAP

Table 5 shows that generally good agreement was obtained between the mobile (field) laboratory analyses and those performed by WESTON's Analytics Division, particularly considering the difference in detection limits between the analytical methods.

Tables 6 and 7 summarize operating characteristics for runs one and two. Based upon the flow rate, observed (average) influent concentrations, and total operating time, the total mass loading of each explosive to each column was calculated. Based upon the total amount of carbon in each column and the specific adsorption capacity of the carbon for the explosive (as determined from the isotherm tests), the total column capacity for each explosive was calculated. Finally, from the mass of explosive applied and the capacity of the column, the percent utilization of column capacity was calculated. As noted previously, the effluent (that is, not adsorbed) fraction of the explosives did not exceed 10%, and most of the time was substantially lower. Therefore, for this calculation the difference between the total mass of explosive applied and that adsorbed (as would be estimated by integration under the respective concentration curves in Figures 3 through 6) was considered minor.

Table 6 shows that after 16 days of operation at the indicated loading, a relatively small fraction of each carbon's capacity for each explosive was utilized. Table 7 suggests the same conclusions for all three loading rates using Hydrodarco 4000. It should be noted that, even though the maximum hydraulic loading in run two was higher than in run one, the explosives concentrations in the groundwater had fallen as compared to run one, resulting in a lower mass loading rate of explosives to the column. Longer activated carbon column contact times (which in this study did not exceed 5 minutes) may provide even longer column lives for a given influent concentration.

These data clearly show that the use of new granular activated carbon in continuous-flow columns produced very low effluent explosives concentrations, generally in the low partsper-billion (ppb) range during the early portions of each run. In most test runs effluent explosives levels rose gradually through the duration of the experiment. However, in no run did the effluent concentration of either 2,4-DNT, or 2,6-DNT exceed 10% of the respective (average) influent concentration before it became necessary to terminate the run. In the first run, which examined both selected carbons at an identical influent rate of 0.5 gpm, effluent levels of 2,4-DNT reached 40 $\mu g/L$ (Filtrasorb 300) and 14 $\mu g/L$ (Hydrodarco 4000) at the end of 16 days of operation (11,520 gallons—7,835 bed

Table 5 WESTON's Analytics Division Results Versus WESTON Field Laboratory Results for Run Two at the Badger Army Ammunition Plant

		WESTON	Analytics		
		Laboratory Results Concentration		Field Labora	tory Results
	Hours			Concen	tration
	into	(μg.	/L)	(μg.	/L)
Sample ID	Test	2,4-DNT	2,6-DNT	2,4-DNT ^a	2,6-DNT
Column A1 (Hydrodarco 4	000 at 0.3 gpm)			
8 March 1989, 16:00	2	18.10	9.80	0.7^{b}	0.6
9 March 1989, 16:00	26	< 0.46	0.30	0.4	0.4
10 March 1989, 16:00	50	< 0.46	0.27	_	
11 March 1989, 16:00	74	< 0.46	0.26	0.3	0.3
12 March 1989, 16:00	98	< 0.46	0.20	0.25	0.25
Column B1 (Hydrodarco 4	000 at 0.7 gpm)			
8 March 1989, 16:00	2	0.50	0.19	0.2^{b}	0.2^{b}
9 March 1989, 4:00	14	< 0.46	0.14	ND ^c	0.1 ^c
9 March 1989, 16:00	26	< 0.46	0.08	ND	0.1
10 March 1989, 4:00	38	< 0.46	0.07	ND ^c	0.1 ^c
10 March 1989, 16:00	50	< 0.46	0.06	-	_
11 March 1989, 4:00	62	< 0.46	0.05	ND^{d}	0.1^{d}
11 March 1989, 16:00	74	< 0.46	0.05	ND	0.1
12 March 1989, 4:00	86	< 0.46	0.05	ND^{d}	0.1^{d}
12 March 1989, 16:00	98	< 0.46	0.04		
$^{a}ND = Below detection limit.$					
^b Sample taken at 14:00 hours on 8 March 1989).				
Samples taken at 6:00 hours.					
"Samples taken at 8:00 hours.					

volumes). Effluent levels for 2,6-DNT at this point were 24 μ g/L (Filtrasorb 300) and 15 μ g/L (Hydrodarco 4000).

The second run employed a higher flow rate through one carbon train (0.7 gpm). Since the output of the monitoring well was limited to approximately 1.0 gpm, the other column train operated at 0.3 gpm. In this run, both column trains employed Hydrodarco 4000 carbon, which had provided the lower final effluent concentrations for both contaminants in run one. After a throughput volume of 16,130 gallons (10,970 bed volumes) for the higher (0.7 gpm) flow rate train, effluent 2,4-DNT was 10 μ g/L and effluent 2,6-DNT was 1.6 μ g/L. Effluent levels were 0.48 μ g/L for 2,4-DNT and 0.12 μ g/L for

2,6-DNT for the lower (0.3 gpm) flow rate train. As with run one, complete breakthrough was not observed, and with the concurrence of USATHAMA, run two was terminated at this point.

It should be noted that a major contributing factor to the inability to obtain breakthrough within available time periods was the low influent concentration of the contaminants of concern, at approximately two orders of magnitude lower than anticipated. The average influent 2,4-DNT concentration for the second run was 50% lower than the first run. Similarly, the average influent 2,6-DNT concentration for the second run was 33% lower than the first run.

Table 6 Activated Carbon Column Operating and Performance Data for Run One at the Badger Army Ammunition Plant Hydrodarco 4000 Versus Calgon Filtrasorb 300

ter = 5 inches (0.42 ft)		
er = 4.25 inches (0.354 ft)		
$= 0.0985 \text{ ft}^2$		
$epth) = 0.197 \text{ ft}^3 (1.47 \text{ gallons})$		
Hydrodarco 4000	Filtrasorb 300	
0.5 gpm	0.5 gpm	
5.0 gpm/ft^2	5.0 gpm/ft^2	
3.0 min.	3.0 min.	
2.50 gpm/ft^3	2.50 gpm/ft^3	
0.42 mg/L	0.42 mg/L	
0.3 mg/L	0.3 mg/L	
0.12 lb/lb	0.05 lb/lb	
0.016 lb/lb	0.044 lb/lb	
4.8 lb	5.8 lb	
0.576 lb	0.290 lb	
0.077 lb	0.255 lb	
0.045	0.045	
0.029	0.029	
7.8%	15.5%	
37.7%	11.4%	
16	16	
	er = 5 inches (0.42 ft) er = 4.25 inches (0.354 ft) = 0.0985 ft ² epth) = 0.197 ft ³ (1.47 gallons) Hydrodarco 4000 0.5 gpm 5.0 gpm/ft ² 3.0 min. 2.50 gpm/ft ³ 0.42 mg/L 0.3 mg/L 0.12 lb/lb 0.016 lb/lb 4.8 lb 0.576 lb 0.077 lb 0.045 0.029 7.8% 37.7% 16	er = 5 inches (0.42 ft) er = 4.25 inches (0.354 ft) = 0.0985 ft ² epth) = 0.197 ft ³ (1.47 gallons) Hydrodarco 4000 Filtrasorb 300 0.5 gpm 0.5 gpm 5.0 gpm/ft ² 5.0 gpm/ft ² 3.0 min. 3.0 min. 2.50 gpm/ft ³ 2.50 gpm/ft ³ 0.42 mg/L 0.42 mg/L 0.3 mg/L 0.3 mg/L 0.12 lb/lb 0.05 lb/lb 0.016 lb/lb 0.044 lb/lb 4.8 lb 5.8 lb 0.576 lb 0.290 lb 0.077 lb 0.255 lb 0.045 0.045 0.029 0.029 7.8% 15.5% 37.7% 11.4% 16 16

*From isotherm data.
Table 7 Activated Carbon Column Operating and Performance Data for Hydrodarco 4000 from Runs One and Two at the Badger Army Ammunition Plant

Column Outer	Diameter = 5 inches (0.42 ft)	
Column Inner	Diameter = 4.25 inche	es (0.354 ft)	
Column Area	$= 0.0985 \text{ ft}^2$		
Bed Volume (a	t 2-ft depth) = 0.197 ft^3 ((1.47 gallons)	
Flow Rate	0.3 gpm (Run Two)	0.5 gpm (Run One)	0.7 gpm (Run Two)
Surface Loading Rate	3.0 gpm/ft^2	5.0 gpm/ft^2	7.0 gpm/ft^2
Contact Time	5.0 min.	3.0 min.	2.2 min.
Hydraulic Loading	1.50 gpm/ft^3	2.50 gpm/ft^3	3.50 gpm/ft ³
2,4-DNT Influent Concentration (avg.)	0.17 mg/L	0.42 mg/L	0.17 mg/L
2,6-DNT Influent Concentration (avg.)	0.2 mg/L	0.3 mg/L	0.2 mg/L
2,4-DNT Capacity (rate)*	0.06 lb/lb	0.12 lb/lb	0.06 lb/lb
2,6-DNT Capacity (rate)*	0.013 lb/lb	0.016 lb/lb	0.013 lb/lb
Weight of Carbon in Column	4.8 lb	4.8 lb	4.8 lb
Column 2,4-DNT Capacity (wt)	0.288 lb	0.576 lb	0.288 lb
Column 2,6-DNT Capacity (wt)	0.062 lb	0.077 lb	0.062 lb
Total 2,4-DNT Loading to Column (lb)	0.0098	0.045	0.023
Total 2,6-DNT Loading to Column (lb)	0.012	0.029	0.027
2,4-DNT Capacity Utilized (lb)	3.4%	7.8%	8.0%
2,6-DNT Capacity Utilized	19.4%	37.7%	43.5%
Run Time (days)	16	16	16

The explosives emissions evaluation during this study consisted of three separate tests during which exhaust gases from the stripper were sampled and analyzed. All three air stripper tests were conducted during the second GAC test run. Stack gases were near saturation at approximately 72% relative humidity for all three runs. Explosives emission data from all three test runs are presented in Table 8.

Table 8 indicates that explosives were detected in the exhaust of the stripper. The feed rate to the stripper was approximately 4 gpm of explosives-contaminated water for each time period for each test; therefore, over a 4-hour period approximately 960 gallons of water were processed through the stripper. The contaminated groundwater influent contained 0.18 mg/L of both 2,4-DNT and 2,6-DNT for emissions tests one and two, and 0.34 mg/L of 2,4-DNT and 0.19 mg/L of 2,6-DNT for test three. These concentrations equate to 3.62×10^{-4} lb/hr for 2,4-DNT and 2,6-DNT for tests one and two, 6.84×10^{-4} lb/hr for 2,4-DNT test three, and 3.82×10^{-4} lb/hr for 2,6DNT test three. When comparing input concentrations to output air emissions, on the average 99.8% of 2,4-DNT and 99.5% of 2,6-DNT remained in the liquid phase and did not exit through the stripper air exhaust.

It was found that the "U" connector at the base of the stack did accumulate contaminated condensate through the course of each test. This water was analyzed and found to contain almost identical levels of explosives as the feed water. Therefore, it can be assumed that this condesnate is a result of mist carryover through the air stripper mist eliminator and that this water should be returned to the feed tank for reprocessing.

Samples from the GAC influent tank indicated that the GAC influent contained very low concentrations of volatiles, as compared to the raw groundwater. Therefore, the air stripper was effective for its intended purpose of removing volatiles and minimizing their potential impact on the GAC test program.

Based upon the data obtained in this study, the preferred carbon for removing 2,4-DNT and 2,6-DNT from pretreated

Test Data		Air Stripper Stack						
Test Location	1	2	- 3					
Test Date	3/13/89	3/13/89	3/14/89	Series Average				
Test Time Period	0902-1310	1352-1758	0807-1342					
Air Stripper Feed Rate (gpm)	4	4	4					
EXPLOSIVES EMISSIONS								
2,4-Dinitrotoluene								
Concentration, lbs/dscf	7.26E-11	9.61E-11	1.26E-10	9.23E-11				
Concentration, $\mu g/m^3$	1.16	1.54	2.02	1.57				
Concentration, ppm/v	1.23E-04	1.63E-04	2.14E-04	1.67E-04				
Mass Rate, lb/hr	5.47E-07	7.36E-07	9.59E-07	7.48E-07				
2,6-Dinitrotoluene								
Concentration, lb/dscf	1.97E-10	2.90E-10	2.11E-10	2.32E-10				
Concentration, $\mu g/m^3$	3.15	4.64	3.38	3.72				
Concentration, ppm/v	3.34E-04	4.92E-04	3.58E-04	3.95E-04				
Mass Rate, lb/hr	1.48E-06	2.22E-06	1.60E-06	1.77E-06				

Table 8 Summary of Explosives Test Data and Test Results from Air Stripper Stack Testing at the Badger Army

Note: Standard Conditions = 68°F (20°C) and 29.2 inches Mg (760 mm Hg).



FIGURE 7. TNT Concentration vs. Column Throughput Volume for Test Three at MAAP

groundwater at BAAP is Hydrodarco 4000.

Influent and effluent concentrations for Tests One, Two, and Three at MAAP showed that granular activated carbon in continuous-flow columns produced very low effluent concentrations for all nine explosives, generally in the low partsper-billion (ppb) range during the early portions of each test. These concentrations for RDX, TNT, 2,4-DNT, and 1,3,5-TNB are presented graphically in Figures 7 through 10.

In Test One, columns A1 and B1 both contained Atochem, Inc. GAC 830 but were operated at different influent flow rates (see Table 4). Only RDX exceeded its detection limit of 0.63 $\mu g/L$ in column A1's effluent during Test One. This occurred after approximately 8,000 gallons (2,715 bed volumes) of influent, containing an average RDX concentration of 487 $\mu g/L$, had passed through the column. When Test One was terminated at a column A1 throughput volume of 10,920 gallons (3,705 bed volumes), the RDX concentration in this column's effluent was 7.05 $\mu g/L$.

Test Two had Atochem, Inc. GAC 830 in column A1 and Calgon Filtrasorb 300 in column B1. Both columns were operated at the same influent flow rate (see Table 4). Even after 16,632 gallons (5,644 bed volumes) of influent had passed through both columns, the effluent concentrations of all nine explosives were below their detection limit, which were all approximately 1.0 ppb.

In Test Three, column A1 had Atochem, Inc. GAC 830, and column B1 had Calgon Filtrasorb 300. Both columns were operated at the same influent flow rate (see Table 4). Since the maximum pumping rate for the system was 1.5 gpm, the



FIGURE 9. 2,4-DNT Concentration vs. Column Throughput Volume for Test Three at MAAP

flow was split evenly between two sets of columns. Only TNT, RDX, 2,4-DNT, and 1,3,5-TNB exceeded their detection limits in columns A1 and B1 effluents. HMX, Tetryl, 2,6-DNB, 1,3-DNT, and NB were all below their detection limits, even after 56,160 gallons (38,112 bed volumes) of influent had passed through both primary columns (A1 and B1). However, Tetryl, 2,6-DNT, and NB were all below their detection limits in the influent, and HMX's and 1,3-DNB's influent concentrations were low, 3.8 and 4.2 $\mu g/L$, respectively.

In Test Three, the average TNT influent concentration was 734 μ g/L, and final effluent levels in columns A1 and B1 were 88.6 and 192 µg/L, respectively (see Figure 7). Even though these levels indicate that Atochem, Inc. GAC 830 performed slightly better than Calgon Filtrasorb 300, both carbons performed about the same (see Figure 7) for most of the test. In particular, the apparent breakthroughs for both carbons occurred at approximately the same column throughput volume of 30,000 gallons (20,359 bed volumes). Apparent breakthrough is the point on the breakthrough curve where the concentration of the column effluent first begins to rise above its initial column leakage concentration. The average RDX influent concentration was 549.1 µg/L, and its final effluent levels in columns A1 and B1 were 315 and 344 µg/L, respectively (see Figure 8). As with TNT, the final effluent levels indicate that Atochem, Inc. GAC 830 performed slightly better than Calgon Filtrasorb 300. However, both carbons performed about the same for most of the test; the apparent breakthroughs for both carbons occurred at approximately the same column throughput volume of 12,000 gallons (8,144 bed volumes).



FIGURE 8. RDX Concentration vs. Column Throughput Volume for Test Three at MAAP



FIGURE 10. 1,3,5-TNB Concentration vs. Column Throughput Volume for Test Three at MAAP

Table 9 Activated Carbon Column Operating and Performance Data for Test Three at the Milan Army Ammunition Plant Atochem, Inc. GAC 830 Versus Calgon Filtrasorb 300

Starting Date: 16 October 1989 Ending Date: 15 December 1989 Column Inner Diameter: 4.25 in. (0.354 ft) Column Area: 0.0985 ft² Bed Volume: 0.197 ft³ (1.47 gal.)

	Atochem, Inc. GAC 830	Calgon Filtrasorb 300
Flow Rate	0.75 gpm	0.75 gpm
Hydraulic Loading	7.6 gpm/ft^2	7.6 gpm/ft^2
Bed Depth	2.0 ft	2.0 ft
Empty-Bed Contact Time	2.0 min.	2.0 min.
TNT Influent Concentration (avg.)	734 μg/L	734 µg/L
RDX Influent Concentration (avg.)	549.1 µg/L	549.1 µg/L
TNT Capacity (rate)*	0.112 lb/lb	0.124 lb/lb
RDX Capacity (rate)*	0.031 lb/lb	0.024 lb/lb
Weight in Carbon in Column	5.1 lb	5.1 lb
Column TNT Capacity (wt)	0.571 lb	0.632 lb
Column RDX Capacity (wt)	0.158 lb	0.122 lb
Total TNT Loading to Column	0.338 lb	0.330 lb
Total RDX Loading to Column	0.191 lb	0.199 lb
TNT Capacity Utilized	59.2%	52.2%
RDX Capacity Utilized	121%	163%
Run Time	54.5 days	54.5 days
Calculated using isotherm data and average influent concentration.	•	

Figures 9 and 10 present the results for 2,4-DNT and 1,3,5-TNB.

Good agreement was found between the analyses performed by WESTON's Analytic Division and those performed by the WESTON field laboratory for all three tests. Except for TNT and RDX, the analytical results from both WESTON's Analytics Division and the WESTON field laboratory show all explosive concentrations in columns A1 and B1 below their detection limits in all three tests. WESTON's Analytics Division's analytical results for TNT and RDX in columns A1 and B1 were for the most part slightly higher than the WES-TON field laboratory's in all three tests. However, the WES-TON field laboratory's analytical results for TNT and RDX in the influent were slightly higher than WESTON's Analytics Division's in Tests Two and Three.

Table 9 summarizes the operating characteristics for Test Three, and Table 10 summarizes the operating characteristics for Atochem, Inc. GAC 830 for Tests One and Two by flow rate. Table 9 shows 59.2% and 52.2% utilization for TNT in columns A1 and B1, respectively. Table 9 also shows 121% and 163% utilization for RDX in columns A1 and B1, respectively. These utilizations of greater than 100% may be attributable to microbiological activity on the carbon surfaces. This activity has been documented in the literature [8-13]. In addition, earlier studies [14] have shown that RDX can be biodegraded, even though it and most explosives are relatively

Table 10 Activated Carbon Column Operating and Performance Data for Atochem, Inc. GAC 830 from Tests One and Two at the Milan Army Ammunition Plant

Column Inner Diameter: 4.25 in. (0.354 ft) Column Area: 0.0985 ft² Bed Volume: 0.394 ft³ (2.94 gal.)

Flow Rate	0.2 gpm (Test One)	0.7 gpm (Test Two)	1.0 gpm (Test One)
Hydraulic Loading	2.03 gpm/ft^2	7.11 gpm/ft ²	10.15 gpm/ft ²
Bed Depth	4.0 ft	4.0 ft	4.0 ft
Empty-Bed Contact Time	14.7 min.	4.2 min.	2.9 min.
TNT Influent Concentration (avg.)	433 µg/L	508 µg/L	433 μg/L
RDX Influent Concentration (avg.)	487 µg/L	536 µg/L	487 µg/L
TNT Capacity (rate)*	0.080 lb/lb	0.088 lb/lb	0.080 lb/lb
RDX Capacity (rate)*	0.029 lb/lb	0.030 lb/lb	0.029 lb/lb
Weight of Carbon in Column	10.2 lb	10.2 lb	10.2 lb
Column TNT Capacity (wt)	0.816 lb	0.898 lb	0.816 lb
Column RDX Capacity (wt)	0.296 lb	0.306 lb	0.296 lb
Total TNT Loading to Column	0.0076 lb	0.0722 lb	0.0379 lb
Total RDX Loading to Column	0.0086 lb	0.0751 lb	0.0431 lb
TNT Capacity Utilized	0.9%	8.0%	4.6%
RDX Capacity Utilized	2.9%	24.5%	14.6%
Run Time	7.6 days	16.5 days	7.6 days

*Calculated using isotherm data and average influent concentration.

Table 11 Activated Carbon Bed Volumes to Reach TNT and RDX Effluent Levels of Approximately 1 μg/L, 10 μg/L, and 100 μg/L from Test Three at the Milan Army Ammunition Plant

Columns A1 and A2: Atochem, Inc. GAC 830 Columns B1 and B2: Calgon Filtrasorb 300 Hydraulic Loading: 7.6 gpm/ft² Bed Volume: 0.197 ft³ (1.47 gal.)

TNT

Average Influent Concentration to Columns A1 and B1: 734 μ g/L Range of Influent Concentration to Column A2: <0.78 to 232 μ g/L Range of Influent Concentration to Column B2: <0.78 to 397 μ g/L

Column	41	Column B1			
Effluent Level Bed (µg/L) Volumes		Effluent Level (µg/L)	Bed Volumes		
0.769	733	1.30	733		
10.9	12,093	3.82	12,093		
92.1 30,783		92.3	30,050		
Column	42	Column	B2		
Effluent Level	Bed	Effluent Level	Bed		
(µg/L)	Volumes	(µg/L)	Volumes		
< 0.78 ^a	37,745	< 0.78°	37,745°		

"Effluent level and bed volume at end of test.

^bEffluent level and bed volume for first sample taken on 10 November 1989.

Notes: <0.78 = Detection limit of TNT.

<0.63 = Detection limit of RDX.

resistant to biodegradation. However, no attempt was made in the current study to assess this possibility.

Based on the percent utilizations in Table 9, Calgon Filtrasorb 300 appears to be slightly better for TNT removal. However, since the calculations of these utilizations are based on the extrapolation of limited isotherm test data, one cannot conclude that one carbon is definitely better than the other for removal of TNT and RDX.

Table 10 shows the percent utilizations of Atochem, Inc. GAC 830 at three different hydraulic loadings for TNT and RDX. Even after 7.6 days at the maximum hydraulic loading of 10.15 gpm/ft², percent utilizations for TNT and RDX were only 4.6% and 14.6%, respectively.

Table 11 shows the activated carbon bed volumes required to reach effluent RNT and RDX levels of approximately $1 \mu g/L$, $10 \mu g/L$, and $100 \mu g/L$ from Test Three. Based on these results, both Atochem, Inc. GAC 830 and Calgon Filtrasorb 300 appear to be equivalent in meeting the three effluent levels for both TNT and RDX. The results also show that if either columns A1 and A2 or columns B1 and B2 were used as two columns in-series instead of as single columns, a better effluent would be obtained over a longer period of time. The following conclusions were drawn from the two pilot studies:

- Granular activated carbon was capable of removing explosives from contaminated groundwater at two Army Ammunition Plants to 1 μg/L.
- The concurrent removal of 2,4-DNT and 2,6-DNT from groundwater at BAAP using continuous-flow granular activated carbon columns is feasible.
- Based on testing performed in the BAAP study, there is little potential for airborne emissions of 2,4-DNT or 2,6-DNT in the exhaust of an air stripper used to remove volatile compounds.
- The concurrent removal of TNT, RDX, HMX, Tetryl, 2,4-DNT, 2,6-DNT, 1,3-DNB, 1,3,5-TNB, and NB from groundwater at MAAP using continuous-flow granular activated carbon is feasible.

ACKNOWLEDGEMENT

Support for these projects by USATHAMA under Contract DAAA15-88-D-0010, Tasks 3 and 7. Wayne Sisk was the USA-THAMA Project Officer.

LITERATURE CITED

- Schulte, G. R., R. C. Hoehn, and C. W. Randall, "The Treatability of a Munitions-Manufacturing Waste with Activated Carbon," *Proceedings of the 28th Industrial Conference, Purdue University*, (1973).
- Shapira, N. I., J. Patterson, J. Brown, and K. Noll, Stateof-the-Art Study: Demilitarization of Conventional Munitions, Phase I Study, American Defense Preparedness Association, (August 1977).
- USATHAMA Report, Granular Activated Carbon (GAC) System Performance Capabilities and Optimization, (February 1987).
- Patterson, J. W., N. I. Shapira, and J. Brown, "Pollution Abatement in the Military Explosives Industry," Proceedings of the 31st Industrial Waste Conference, Purdue University, (1976).
- Hinshaw, G. D., Fanska, C. B., Fiscus, D. E., and Sorenson, S. A., "Granular Activated Carbon (GAC) System Performance Capabilities and Optimization," Report AMXTH-TE-GR87111, U.S. Army Toxic and Hazardous Materials Agency, Aberdeen Proving Ground, MD, 27 February 1987.
- Energy and Environmental Systems Division, Argonne National Laboratory, "Master Environmental Plan for the Badger Army Ammunition Plant Volume 1: Main Text and Appendix A(U)." Report Number AMXTH-IR-CR-87131, Prepared for USATHAMA, (January 1988).
- 7. USATHAMA Report, Use of Activated Carbon for Treat-

ment of Explosives-Contaminated Groundwater at the Badger Army Ammunitions Plant (BAAP), (August 1989).

- Weber, W. J., M. Pirbazari, and G. L. Melson, "Biological Growth on Activated Carbon: An Investigation by Scanning Electron Microscopy," ENVIRONMENTAL SCI-ENCE AND TECHNOLOGY, 12(7), p. 817, (1978).
- Benedek, A., "Simultaneous Biodegradation and Activated Carbon Adsorption—A Mechanistic Look," in Activated Carbon Adsorption of Organics from the Aqueous Phase, Vol. II, ed., M. J. McGuire and I. H. Suffet, Ann Arbor Scientific Publications, Ann Arbor, MI, (1980).
- Latosek, A., and A. Benedek, "Some Aspects of the Microbiology of Activated Carbon Columns Treating Domestic Wastewater," ENVIRONMENTAL SCIENCE AND TECHNOLOGY, 13(10), p. 1285, (1979).
- 11. Martin, R. J., and K. O. Iwugo, "Studies in Residual

Organics in Biological Plant Effluents and Their Treatment by the Activated Carbon Adsorption Process," in Activated Carbon Adsorption of Organics from the Aqueous Phase, Vol. II, ed. M. J. McGuire and I. H. Suffet, Ann Arbor Scientific Publications, Ann Arbor, MI, (1980).

- Van der Kooij, D., "Some Investigations into the Presence and Behavior of Bacteria in Activated Carbon Filters," in Translation of Reports on Special Problems of Water Technology, Volume 9-Adsorption, ed. H. Sontheimer, U.S. Environmental Protection Agency Report No. EPA-600/9-76-030, (1976).
- 13. Cairo, P. R., J. McElhaney, and I. H. Suffet, Pilot Plant Testing of Activated Carbon Adsorption Systems, JOUR. AWWA, 71: 11, p. 660, (1979).
- 14. USATHAMA Report, Composting of Explosives-Contaminated Soil Technology, (October 1989).

Evaluation of a Hydraulically-Installed Suction Lysimeter to Obtain Representative Soil Water Samples

Mark Chaimberg, Ronald H. Carty

Institute of Gas Technology, 3424 S. State St., Chicago, IL 60616

Joseph A. Scroppo

Biadon International, Inc., 880 Lee St., Des Plaines, IL 60016

The existing technology for obtaining liquid from the vadose zone involves drilling a bore hole, introducing a silica slurry packing, inserting a sampling device, and backfilling the hole. The high cost of drilling and backfilling contribute to the considerable expense of this procedure. Moreover, drilling through potentially contaminated soil increases the possibility of expanding the area of contamination. In this study, a novel small diameter suction lysimeter with a porous ceramic section was designed and tested. The stainless steel ram-tipped lysimeter was designed to be installed into the ground using a hydraulic ram without the need of drilling a bore hole, thereby reducing the expense and potential for cross-contamination. Simulated field testing was employed to evaluate the performance of the hydraulically-installed lysimeter as compared to a lysimeter installed using the standard silica slurry packing technique. The decrease in sample volume uptake due to plugging in the pores of the ceramic section was investigated for the lysimeters installed both hydraulically and with the silica slurry packing, in three types of soil: medium sand, silty clay, and a clayey topsoil-medium sand blend. The soil moisture operating range for the lysimeters was ascertained in a test chamber filled with a blend of clayey topsoil and medium sand. An investigation of the effect of sampling on the concentration of solutes was performed with aqueous solutions of ethanol and phenol.

INTRODUCTION

The vadose zone is defined as the unsaturated area located between ground level and the groundwater table. The vadose zone is characterized by the presence of solid, liquid, and vapor phases. Chemical spills [1], underground storage tank leaks [2], and improper burial of materials [3] have the potential to cause extensive vadose zone and groundwater contamination. Liquids are transported downward through the soil by percolation and gravitation forces, and upward and laterally as a result of capillary action. Vapors are conveyed through the soil due to concentration and temperature gradients. Lateral movement of liquids and vapors is generally limited. The transport of both liquids and vapors is facilitated, however, by the presence of fractures or channels.

To ascertain the extent of underground pollution and the potential for groundwater contamination requires monitoring in the vadose zone. Samples of the vapor phase are usually collected by a vapor probe, a long tube which is installed vertically to a particular depth [2]. Suction lysimeters are used to collect liquid samples from the vadose zone. The lysimeter sampling apparatus consists of a hollow porous section that is maintained at vacuum. Liquid is transported into the lysimeter due to the pressure differential across the porous section. The flow of liquid through the porous section is sustained by maintaining a pressure differential which is finite yet below the entry capillary pressure. The performance of suction lysimeters is restricted by the physical and chemical properties of the porous section. The porous section may be categorized by the type material, pore size distribution (or permeability), surface area, degree of hydrophilicity, and surface adsorption characteristics.

The existing technology for obtaining liquid from the vadose zone involves drilling a bore hole (usually four inches in diameter), injecting a silica slurry packing into the bottom of the hole, inserting a lysimeter sampling device, and backfilling and plugging the hole [4]. This procedure is expensive due to the high cost of drilling and backfilling the well. Furthermore, the potential for cross-contamination is exacerbated by drilling a conduit through potentially contaminated soil. In this work, a new lysimeter design, approximately one inch in diameter, capable of being hydraullically installed at desired depth without drilling, was evaluated. The results of this study will be used to improve the operation and lower the cost of sample collection in the vadose zone using suction lysimeters.



FIGURE 1. Schematic diagram of suction lysimeter.

DESIGN AND OPERATION

The suction lysimeter was designed to be installed into the vadose zone with a hydraulic ram. This design eliminates the need to drill and backfill a bore hole. A schematic diagram of the hydraulically-installed suction lysimeter is shown in Figure 1. The lysimeter is comprised of a stainless steel ram point and body, one or more ceramic porous sections, and one or more extraction tubes. The stainless steel ram point was designed to promote the installation of the lysimeter into the test soils by a hydraulic ram. Liquid present in the vadose zone may be drawn into the lysimeter through the porous section by application of negative pressure on the extraction tubes. The liquid samples are transported to collection vessels at the surface via the extraction tubes.

The porous section of the suction lysimeter can be modeled as a collection of capillary tubes operating in parallel with each other. In saturated soil, liquid enters each capillary tube and is transported to the interior of the lysimeter. In unsaturated soils, however, soil vapor may infiltrate the capillary tube and displace the liquid. If vapor occupies the entire capillary tube, the lysimeter will preferentially collect vapor and severely limit the sampling of liquids. The pressure at which vapor enters the pores is known as the bubble pressure or capillary entry pressure, P_{E} , and may be described by the following equation for a porous medium of uniform pore size [5].

$$P_E = \frac{4\gamma \cos\theta}{D_t} \tag{1}$$

where γ is the surface tension of the liquid, θ is the angle of contact between the liquid and the surface of the solid porous material, and D_t is the pore diameter. The capillary entry pressure limits the operating range of the suction lysimeter by establishing a maximum suction pressure for a particular pore diameter. For example, to obtain an aqueous sample at 20°C ($\gamma = 72.8$ dynes/cm) from unsaturated soil by a hydrophilic porous section ($\theta = 0^\circ$) it is essential that the maximum pore diameter of the porous section be less than 2.87 μ m.

EXPERIMENTAL

Experimental testing of the hydraulically-installed suction lysimeters was performed at the Institute of Gas Technology's Vadose Zone Testing Facility. The lysimeters were evaluated in terms of bubble pressure, pore plugging, soil moisture operating range, and representative sampling.

Table 1 Soil Characteristics								
Soil Type	Median Particle Size (μm)	Specific Gravity	Organic Content (%)					
Median Sand	500	2.65	0.0					
Silty Clay	6	2.72	6.1					
Clayey Topsoil	4	2.70	6.3					

Bubble Pressure Tests

Bubble pressure tests were performed on each lysimeter to determine the capillary entry pressure of the porous section. To test for leaks, the lysimeter was submerged into a transparent container filled with distilled water and was pressurized with nitrogen. Suction was then applied to the extraction tube and 0.5 L of water was drawn into the lysimeter. Suction was discontinued and the extraction tube was gradually pressurized with nitrogen. The bubble pressure, defined as the pressure at which bubbles begin to form on the exterior surface of the porous section of the lysimeter, was recorded.

Pore Plugging Tests

Pore plugging tests were performed using four lysimeters, two installed hydraulically and two using a standard silica slurry packing, in saturated test soils. Three soil types were obtained for use in the pore plugging and soil moisture operating range tests. STS Consultants Ltd. (Northbrook, IL) performed a combined sieve hydrometer analysis following ASTM D-422 procedures and an organic content test following ASTM D-2974 for the three soils described as medium sand, silty clay, and clayey topsoil. A summary of the median grain size, specific gravity, and organic content is given in Table 1, and the grain size distribution is shown in Figure 2. Upon consultation with soil scientists at the U.S. Department of Agriculture (USDA), it was recommended that a blend of 75% clayey topsoil with 25% medium sand would yield a soil with intermediate permeability representative of "typical" topsoil. Therefore, based on the soil analyses and USDA recommendation, the composition of the three test soils were: (1) 100% medium sand; (2) 100% silty clay; and (3) a 75% clayey topsoil-25% medium sand blend.

Three 200 L polyethylene (PE) drums, each filled with one of the three different test soils, were used for evaluating pore plugging in the suction lysimeters. The system configuration



FIGURE 2. Grain size distribution for soil types



FIGURE 3. Schematic diagram of pore plugging test apparatus.

consisted of a large capacity vacuum pump connected to each of four collection flasks fabricated from translucent, 10 cm diameter, polyvinychloride (PVC) pipes with sealed PVC end caps. The four 12 L collection flasks, graduated in 1 L intervals to facilitate volume measurement, were connected to the four suction lysimeters (Figure 3). The moisture content of the test soil was measured by a tensiometer (Model 2725 Jet Fill, Soilmaster Equipment Corp., Santa Barbara, CA) inserted into the center of the test chamber.

The following protocol was used for preparing the soil in the test chambers. The soil was packed into each test chamber to a depth of 30 cm in intervals of 15 cm, and the soil was hand tamped after each interval. To accommodate the silica slurry packing, two 10 cm diameter PVC casings were placed on top of the soil at the desired location in the sample chamber (Figure 3). The silica slurry, composed of 500 g of 200 mesh silica flour (Soilmaster Equipment Corp.) in 0.15 L distilled water, was poured into the PVC casing to a depth of approximately 10 cm. For the standard installation, each of two lysimeters was inserted into the silica slurry in the center of the PVC casing. Additional silica slurry to cover the porous cup section of the lysimeter was poured into the casing. The remainder of the soil was added to the test chamber to the same depth as the silica slurry at 15 cm intervals. The PVC casings were carefully removed and the sample chamber was completely filled with soil, hand tamped every 15 cm. Two additional lysimeters were installed using a hydraulic ram into the test soil to the same depth as the standard lysimeters. To measure the soil suction, a tensiometer was installed at the center of the test chamber at the same depth as the lysimeters. The test soil was thoroughly saturated with water and allowed to equilibrate overnight.

Pore plugging was evaluated by applying suction to the four 12 L collection flasks, each of which was connected to a lysimeter. After five hours the suction was discontinued and the water drawn from the soil by each lysimeter was measured. The test was repeated until the incremental volume of water obtained from each lysimeter reached a plateau value.

Soil Moisture Operating Range Tests

The soil moisture operating range test chamber consisted of a PE drum (1100 L) filled with test soil. The configuration of the vacuum and collection apparatus was similar to that of the pore plugging system, consisting of the large capacity vacuum pump and four 12 L collection flasks. The operating range apparatus included a series of extraction plates (Model 1950, Soilmaster Equipment Corp.) and extraction tubes (Model 1900, Soilmaster Equipment Corp.) used to reduce the moisture content of the test soil (Figure 4). Soil moisture operating range was evaluated for the clayey topsoil-medium sand blend using four lysimeters; two installed hydraulically and two with a standard silica slurry packing. Preparation of the test soil in the test chambers and installation of the lysimeters followed the procedure previously described for the pore plugging tests. The test soil was thoroughly saturated with water and allowed to equilibriate for two weeks with the periodic addition of water. To control the suction pressure in the test soil, two extraction plates were buried near the bottom of the test chamber and five extraction tubes were inserted into the test soil (Figure 4). Soil suction was measured by a series of five tensiometers, installed in the test chamber to the same depth as the lysimeters.

The lysimeter operating range was evaluated by applying suction to four 12 L graduated, collection flasks, each of which was connected to a lysimeter. After one hour, suction was discontinued and the volume of water collected by each lysimeter was measured. The soil moisture content was subsequently decreased by removing water from the test chamber via the extraction plates and tubes. The soil suction pressure was permitted to equilibriate over a period of twenty-four hours and the one hour operating range test was repeated for progressively higher soil suction pressures (lower soil moisture content).

Representative Sampling Tests

Sample representativeness was evaluated for aqueous solutions of ethanol and phenol, varying from 0.5 to 10 g/L. A suction lysimeter was inserted into each standard solution and was used to transfer a sample of approximately 0.5 L to a collection flask. The hydrocarbon concentration of the extracted sample was compared to that of the original solution by total organic carbon (TOC) analysis.

RESULTS AND DISCUSSION

Bubble pressure tests were performed on each suction lysimeter to detect hairline cracks in the porous section and to ensure that the capillary entry pressure had a tolerance range of $\pm 5\%$. After replacing several of the porous ceramic sections, sixteen lysimeters with bubble pressures equal to 117 ± 5 kPa were obtained. The relationship between capillary entry pressure and pore diameter (Eq. 1) establishes that a pressure of 117 kPa corresponds to a pore diameter of 2.5 μ m. This



FIGURE 4. Schematic diagram of soil moisture operating range test apparatus.



FIGURE 5. Results of pore plugging test in medium sand test soil.

value correlates with the maximum pore diameter of 2.4 μm derived from mercury porosimetry of the porous ceramic section.

Pore plugging experiments were carried out in three different saturated test soils: medium sand, silty clay, and a 75% clayey topsoil-25% medium sand blend. Figure 5 shows the volume of liquid collected during each 5-hour test over the complete test period for the medium sand test soil. At the start of the experiment, the volume of water collected by the hydraulicallyinstalled lysimeters (labeled C and D) was approximately 10 times that collected from the lysimeters (labeled A and B) installed by the standard method using the silica slurry packing. Pore plugging in the ceramic section of the lysimeters resulted in a decrease in the volume of water collected during subsequent 5-hour test periods. At the conclusion of the 21-day test period, the volume of water collected by the lysimeters had reached a plateau value, with the exception of lysimeter B which exhibited anomalous results. For the hydraulically-installed lysimeters C and D, the volume of water was reduced by an average of 65% over the complete test period. The lysimeter A installed with the silica slurry packing showed a decrease of 55% over the same time period. The larger sample volumes collected by the lysimeters C and D demonstrate the advantage of hydraulic installation for soils similar to the medium sand test soil.

The results of the pore plugging experiments in the silty clay test soil are shown in Figure 6. Contrary to the results in the medium sand test soil, the hydraulically installed lysimeters collected a significantly smaller volume of water than the lysimeters installed using the standard silica slurry packing. The



FIGURE 7. Results of pore plugging test in 75% clayey topsoil-25% medium sand test soil.

smaller uptake of sample by the hydraulically-installed lysimeters may be a consequence of the relatively small particle size of the silty clay test soil and thus, its increased ability to plug the pores of the ceramic section. The silica slurry packing shields the porous ceramic section from intimate contact with the soil and therefore, these lysimeters operate at a higher efficiency. All four of the lysimeters, however, exhibited a marked decrease in the sample volume collected over the period of the test.

Results similar to the pore plugging in the silty clay test soil were obtained using the clayey topsoil-medium sand test soil (Figure 7). The relatively small particle size of the topsoil produced a larger degree of pore plugging in the ceramic section of the hydraulically-installed lysimeters. Furthermore, the volume of water collected in the clayey topsoil-medium sand was significantly less than the volume collected in the silty clay. This data further corroborates the concept that the particle size of the soil strongly influences the degree of pore plugging.

A comparison of the collection rate at the completion of the pore plugging test in the three test soils is shown in Figure 8. The rates of collection at the completion of the test illustrate the differences in the sample volume which could be obtained from different soils over a prolonged sampling period. In terms of a 24-hour daily collection rate at the completion of the plugging test, the hydraulically installed lysimeters collected liquid at a rate of 14.7, 1.2, and 0.5 liters per day for the medium sand, silty clay and clayey topsoil-medium sand test soils, respectively. For comparison, the lysimeters installed with the silica slurry collected liquid at a rate of 4.0, 4.4, and



FIGURE 6. Results of pore plugging test in silty clay test soil.



FIGURE 8. Comparison of pore plugging tests in three test soils.

Environmental Progress (Vol. 11, No. 3)



FIGURE 9. Results of soil moisture operating range test in 75% clayey topsoil-25% medium sand test soil.

1.2 liters per day. Although the hydraulically-installed lysimeters collected smaller volumes in two of the three test soils, these volumes were more than adequate to meet the need for monitoring and analysis on a daily basis.

Soil moisture operating range experiments were performed in the clayey topsoil-medium sand test soil to determine the extent of any restrictions in the function of the lysimeters in unsaturated soils. The soil suction pressure is the conventional mode of indicating the degree of soil pressure, with 0 kPa corresponding to a completely saturated soil. Unfortunately, the soil suction pressure is not linearly proportional to the moisture content in the soil. Furthermore, the relationship varies depending on the composition of the soil. Moisture analysis of the completely saturated clayey topsoil-medium sand blend by weighing and drying yielded a soil water content of 27.0%.

The volume of water collected by the lysimeters as a function of the soil suction pressure is shown in Figure 9. The data indicate that the hydraulically-installed lysimeters collected a negligible volume of liquid when the soil suction pressure was greater than 5 kPa. The lysimeters installed with the silica slurry packing continued to extract liquid samples from the soil at a soil suction pressure of 17 kPa (17.7% moisture). It is suggested that the limited capacity of the hydraulically-installed lysimeters is a result of a possible break in contact of the porous ceramic section with the soil as the soil gets progressively drier [0], as well as a decrease in the efficiency of the porous section due to pore plugging. It is possible that contact would be maintained and samples obtained at high soil moisture pressures if the lysimeter were hydraulically installed into a soil of low moisture content.

Representative sampling tests were performed to determine whether the concentration of organic compounds extracted by the suction lysimeter was representative of the original solution. These tests were used to assess the degree of adsorption onto the porous ceramic, as well as the possibility of loss through the vacuum system. The results of the representative sampling tests for single extractions of aqueous ethanol and phenol solutions are given in Table 2. The data indicate that for the compounds and their concentration range used in this study the lysimeter collected a representative sample of the solution. Additional analysis using lower concentration solutions in which the lysimeter is inserted into a test soil would be required to further corroborate these results.

CONCLUSIONS

A novel, hydraulically-installed suction lysimeter for obtaining liquid samples from the vadose zone was designed and

Table 2 Lysimeter Sampling of Aqueous Solutions									
Solution Concentration (g/L) Percer									
Ethanol	10.00	9.14	91						
	5.00	4.84	97						
	2.50	2.47	99						
	1.00	0.97	97						
	0.50	0.59	118						
Phenol	10.00	8.97	90						
	5.00	3.53	71						
	2.50	2.65	106						
	1.00	1.06	106						
	0.50	0.44	88						

tested. It was demonstrated that the suction lysimeter could be installed using a hydraulic ram, thus eliminating bore hole drilling and silica slurry packing. The results of pore plugging experiments demonstrate a relationship between the pore size of the ceramic section and the particle size of the soil. The larger size sand did not plug the pores of the ceramic section and the flow rate into the hydraulically-installed lysimeter was higher than into the lysimeter with the silica slurry packing. Conversely, the topsoil-sand and clay soils markedly decreased the flow of liquid into the hydraulically-installed lysimeter. However, these volumes obtained from all three test soils by the hydraulically-installed lysimeters were far in excess of the amount required for monitoring and analysis on a daily basis. The soil moisture operating range tests in topsoil-sand blend showed that the hydraulically-installed lysimeters were limited in operation to a soil suction pressure of approximately 5 kPa, while those lysimeter installed with the silica slurry packing collected samples at pressures up to 17 kPa. Current work is being done to improve the pore size distribution of the ceramic section for operation in low moisture environments.

ACKNOWLEDGMENTS

This work was supported by the U.S. Department of Energy, Office of Technology Development, under contract W-31-109-ENG-38.

LITERATURE CITED

- Crow, W. L., Anderson, E. P., and Minugh, E. M., "Subsurface Venting of Vapors Emanating from Hydrocarbon Product on Ground Water," *Ground Water Monitoring Review* 7(1), 1987, pp. 51-57.
- Schlender, M. H. and Rowe, J., "Soil Gas Techniques for the Investigation of Underground Storage Tanks," *Envi*ronmental Progress 8, 1989, pp. 231-234.
- Hilton, J. and Colarullo, S., "Monitoring in the Vadose Zone at Two Inactive Uranium Mill Tailing Sites," *Geo*technical and Geohydrological Aspects of Waste Management, A. A. Balkema, Ed., Colorado State University, pp. 439-446 (1987).
- Everett, L. G., McMillion, L. G., and Eccles, L. A., "Suction Lysimeter Operation at Hazardous Waste Sites," Ground-Water Contamination: Field Methods, ASTM STP 963, A. G. Collins and A. I. Johnson, Eds., American Society for Testing and Materials, Philadelphia, pp. 304-327 (1988).
- Adam, N. K., The Physics and Chemistry of Surfaces, Oxford University Press, London, pp. 189-192 (1941).
- Ball, J. and Coley, D. M., "A Comparison of Vadose Zone Monitoring Procedures," in Proceedings of the Sixth National Symposium and Exposition on Aquifer Restoration and Ground Water Monitoring. Columbus, Ohio. May 19-22, 1986. National Well Water Association, Dublin, Ohio, pp. 52-61 (1986).

Ultrasonic Destruction of Chlorinated Compounds in Aqueous Solution

Jiann M. Wu, H. S. Huang, C. D. Livengood

Energy Systems Division, Argonne National Laboratory, Argonne, Illinois 60439

The application of ultrasonic-wave energy for the destruction of low concentrations of carbon tetrachloride into nonhazardous end products is described, and the results of experimental work are presented. Process parameters studied included steady-state temperature, concentration, pH, irradiation time, and the intensity of the applied ultrasonic-wave energy. Greater than 99% destruction efficiencies were achieved through this process, and the irradiation time required for a given degree of destruction decreased with increasing intensity of the ultrasonic energy. In addition, a detailed chemical reaction mechanism for the destruction of carbon tetrachloride in water was formulated. The agreement between the model and experimental results is generally good.

INTRODUCTION

Chemical ultrasonics began in 1927 when Richards and Loomis reported the acceleration of conventional reactions and the redox process by ultrasound [1]. Since then, a number of chemical reactions have been observed in an ultrasonic field [2-4]. In recent years, attention has focused upon the application of ultrasonic energy to problems associated with water pollution, especially in removing toxic and hazardous organic compounds from contaminated water [5, 6]. Removal of these compounds may be difficult and costly if very low concentration levels must be achieved. For many contaminants, the ultrasonic process has the advantage of completely destroying or converting these organics, not simply transferring them to another medium.

The chemical effects of ultrasound are due to the phenomenon of acoustic cavitation [7]. Sound is transmitted through any fluid as a wave consisting of alternating compression and rarefaction cycles. If the rarefaction wave is sufficiently powerful, it can develop a negative pressure large enough to overcome the intermolecular forces binding the fluid. As a result, the molecules are torn apart from each other and form tiny microbubbles. These microbubbles gradually grow during the compression/rarefaction cycles until they reach a critical size. Subsequent compression then causes the microbubbles to collapse almost instantaneously, thereby releasing a large amount of energy. Temperatures of the order of 5000 K have been experimentally obtained [8], and pressures of the order of 1000 atmospheres have been calculated [9].

The primary chemical reactions result, therefore, from the transient state of these high pressures and temperatures, both during and immediately after collapse of the microbubbles [10, 11]. Solvent and solute vapors in the cavity undergo direct thermal dissociation to yield CO_2 , H_2O , and radicals, such as hydroxyl and atomic hydrogen. Some of these radicals may recombine to form new compounds. In the bulk liquid phase, secondary reactions between solute molecules and radicals generated in the cavities also take place. These two reaction schemes together are responsible for the chemical effect of ultrasonics upon the solution.

In this paper, we present the results of an investigation of the ultrasonic irradiation of carbon tetrachloride in laboratorysimulated groundwater at various pH values, temperatures, and power intensities. Kinetic data and selected chemical mechanisms are discussed and proposed. To study oxidant efficiency, chemical oxidants, such as hydrogen peroxide, are also considered. This work is part of a project entitled "Ultrasonic Process for Detoxification of Groundwater and Soil," sponsored by the U.S. Department of Energy, Office of Technology Development, to develop an innovative process for the effective destruction of chlorinated organics in soil and groundwater.

Environmental Progress (Vol. 11, No. 3)



FIGURE 1. Experimental Apparatus

EXPERIMENTAL CONSIDERATIONS

Reaction Apparatus

Irradiations were carried out with an ultrasonic power supply (Sonics & Materials, VC 600) with a continuously variable output from 0 to 600 W, operating at 20 kHz. Figure 1 shows the schematic diagram of the experimental apparatus, and Figure 2 shows a close-up of the reaction cell. The ultrasonic intensity at the titanium tip of the sonication probe was about 27 W/cm^2 . The sonication vessel was a borosilicate glass cell with a stainless steel collar, which could be screwed onto the horn of the ultrasonic probe so that the probe could be immersed into the sample solution as sonication proceeded. The reaction vessel had side arms so that an oxidant, such as hydrogen peroxide, could be introduced into the reactor, and gas samples could be directed to a gas analysis system.

The temperature inside the reaction vessel was kept relatively constant by circulating cooling water in a constant-temperature bath, which was continuously monitored by a thermocouple probe and temperature readout. An alternative ultrasonic power supply (Sonics & Materials, VC 60) was used to study the effect of power intensity upon the destruction of organics. The major improvement of this apparatus is the addition of a wattmeter, which displays the actual, instantaneous ultrasonic power delivered into the sample solution. In this way, converter and probe losses can be automatically deducted so that the effect of power intensity can be accurately investigated.

Materials

Carbon tetrachloride (CCl₄), sulfuric acid (certified ACS



FIGURE 2. Sonication Cell

grade, Mallinckrodt, Inc.), hydrogen peroxide (30%, analytical grade, Mallinckrodt, Inc.) and n-hexane (certified for trace environmental analysis by capillary GC & GC-MS, Bardick & Jackson) were used as received by the suppliers. Sodium sulfate (certified ACS grade, Mallinckrodt, Inc.) was washed by nhexane to remove trace impurities and was dried in an oven (104°C) overnight before it was used. A standard CCl₄ solution was prepared by stirring the neat liquid with laboratory deionized water overnight and was equilibrated for at least 24 hours. This standard solution was then used to prepare all of the sample solutions for the subsequent experiments.

Analytical Method

Analysis of CCl₄ was carried out by a gas chromatograph equipped with an electron capture detector. A glass column packed with a 80/120 Carbopack B/3%, SP-1500 stabilizer was used after it had been conditioned at 230° C for more than 15 hours. Immediately following the experiment, irradiated sample solutions were extracted with n-hexane and were dehydrated by sodium sulfate before being analyzed. The pH value of the sample solution in the pH-effect study was determined by using a Cole-Parmer Chemcadet pH/ion/mv meter (model 5986-50), which was calibrated with standard solutions of pH 4, 7, and 10 before use.

Experimental Procedures

A 15-mL sample solution, which was diluted from the standard CCl₄ solution, was prepared for each experiment and was irradiated for a desired length of time. The power intensity delivered into the solution was regulated by a controller on the power supply panel. The temperature inside the reactor was monitored continuously throughout the experiment. After sonication ceased, the irradiated solution was extracted, dehydrated, and then stored in a Teflon-sealed sampling bottle for future treatment or analysis.

CHEMICAL KINETIC MECHANISM AND MODEL DEVELOPMENT

The chemical mechanism presented in Table 1 was developed by systematically considering the major plausible elementary reactions of water, dissolved air (nitrogen and oxygen), organics (CCl₄), and their related reactions in the field of ultrasonics [12-14]. The reactions are initiated by the dissociation of water molecules in the collapsing cavitation holes (the 'Hot Spot') (Eq. 1). Atomic hydrogen (H) and hydroxyl radicals (OH) are formed through this reaction. These two radicals recombine to form water or hydrogen molecules (H₂), or they react with oxygen molecules (O2) to yield hydroperoxide radicals (HO₂) and hydrogen peroxide (H₂O₂) (Eqs. 2 to 6). These radicals further react with the H2O2 produced in these reactions to form H₂ molecules, water, and other radicals (Eqs. 7 to 10). In order to simplify the treatment, we will not consider here the secondary reactions of H and OH with the H₂O₂ that is formed in the system.

In the presence of dissolved nitrogen, nitrogen molecules (N_2) decompose in the collapsing cavitation holes (Eq. 11) to form atomic nitrogen (N). Following a series of reactions, N then reacts with the OH radicals produced from the dissociation of water and related reactions to yield nitrate and nitrite (Eqs. 12 to 15). Atomic nitrogen also reacts with H and O_2 molecules to regenerate N_2 and H_2 molecules and to produce nitric-oxide (NO) and O_2 radicals (Eqs. 16 to 18).

Oxygen molecules dissolved in the water also decompose in the cavitation holes (Eq. 19) to produce atomic oxygen (O) and compete with H in Eq. 2, leading to the formation of

Table 1 Proposed Chemical Kinetic Mechanism

A. Water Dissociation:
1. $H_2O_{})))H + OH$
2. $H + H \rightarrow H_2$
3. $H + O_2 \rightarrow HO_2$
4. $HO_1 + HO_2 \rightarrow H_2O_2 + O_2$
5. $OH + OH \rightarrow H_2O_2$
6. $H + OH \rightarrow H_2O$
7. $H + H_2O_2 \rightarrow OH + H_2O_2$
8. $H + H_2O_2 \rightarrow H_2 + HO_2$
9. $OH + H_2O_2 \rightarrow HO_2 + H_2O_2$
10. $OH + H_2 \rightarrow H_2O + H$
B. In the Presence of Nitrogen:
11. N_2 —)))2N
12. $N + OH \rightarrow NO + H$
13. NO + OH \rightarrow HNO ₂
14. $NO + OH \rightarrow NO_2 + H$
15. $2NO_2 + H_2O_2 \rightarrow HNO_2 + HNO_3$
16. $N + H \rightarrow NH$
17. $NH + NH \rightarrow N_2 + H_2$
18. $N + O_2 \rightarrow NO + O$
C. In the Presence of Oxygen:
19. O ₂ —)))2O
20. $H + O_2 \rightarrow OH + O$
21. $O + H_2 \rightarrow OH + H$
22. $O + HO_2 \rightarrow OH + O_2$
23. $O + H_2O_2 \rightarrow OH + HO_2$
D. In the Presence of Organics (CCl ₄):
24. $CCl_4 + OH \rightarrow products$
25. $CCl_4 + H \rightarrow products$
26. $CCl_4 + HO_2 \rightarrow products$
27. $CCl_4 + O \rightarrow products$
28. CCL—)))products

hydroxyl radicals (Eq. 20). Atomic oxygen then reacts with H_2 molecules, H_2O_2 , and HO_2 to form O_2 molecules and other radicals, such as H and OH (Eqs. 21 to 23).

The target contaminant, CCl_4 , present in the water is either directly decomposed in the cavities (Eq. 28) or oxidized by radicals (Eqs. 24 to 27). If the reaction does not proceed to completion, the final products of the irradiation of CCl_4 by ultrasound are water, carbon dioxide, and/or some other chemical compounds.

On the basis of the chemical reaction mechanism proposed above, the primary pathway appears to be the thermal dissociation in the collapsing cavitation holes, such as those represented in Eqs. 1, 11, 19, and 28. The high temperatures and pressures created during cavitation provide the activation energy required for the bond cleavage. Water, nitrogen, oxygen, and carbon tetrachloride molecules decompose in these cavities to directly form the radicals. These radicals then either react with each other to form new molecules and radicals or diffuse into the bulk liquid to serve as oxidants. The secondary reaction seems to be in the bulk-liquid phase, where CCl_4 and other molecules are oxidized by oxidants, such as the radicals generated in the cavities.

Therefore, to better clarify these reactions, the whole system can be divided into two major areas: (1) the collapsing cavitation holes, in which molecule vapors may reach supercritical conditions and decompose, and (2) the bulk-liquid phase under normal operating temperatures and pressures, in which molecules are oxidized by radicals to form new products. In a batch reactor with reactions undergoing the bulk-liquid phase, we assume a second-order rate constant, in which the destruction rate of CCl_a can be represented by the following equation:

 $-d[CCl_4]/dt = k_{24}[OH][CCl_4] + k_{25}[H][CCl_4]$

 $+ k_{26}[HO_2][CCl_4] + k_{27}[O][CCl_4]$ (1)

The rate of disappearance of CCl_4 equals the sum of the rates of all of the related reactions in the mechanism. If the reaction takes place in the cavitation holes, it is reasonable to assume a second-order rate constant again. In this case, the destruction rate of CCl_4 can be expressed by a rate formula:

$$-d[CCl_4]/dt = k_c * k_{28}[CCl_4][M]$$
(2)

where M is any collision partner and k_c is the system adjustment coefficient, which is a function of bubble concentration, bubble radius, mixing extent of the system, etc. This coefficient is assumed to be a constant if all of the experimental conditions, such as reaction vessel size, steady-state temperature, and power intensity, are unchanged. This coefficient can be obtained by a best fit of the experimental data into the model. Because the reaction occurs in the collapsing cavitation holes, [CCL₄] in rate formula (2) represents the concentration of CCL₄ in the vapor phase. If we assume an ideal gas and Rault's law to hold, this value can be estimated by Henry's law.

Hence, in addition to CCl_4 , the net reaction rates for all the other compounds in the system can also be expressed in similar rate formulas, or differential equations, that describe the decreasing or increasing rates in the field of ultrasonics. If all of the reaction rate constants are available, these differential equations could be solved simultaneously to obtain the individual concentration profile for each species as a function of irradiation time.

RESULTS AND DISCUSSION

Exposure of CCl₄ solutions to ultrasound in the presence of dissolved air results in a decrease of the [CCl₄] in the solutions. Greater than 99% removal efficiency was obtained in the current experiment. Figures 3, 4, and 5 show the plot of [CCl₄] versus sonication time at various initial [CCl₄]. Initial [CCl₄] as high as 130 ppm (Figure 3) dropped to about 5 ppm after





FIGURE 4. [CCI₄] Decrease Versus Sonication Time; [CCl₄] = 47 ppm

6 minutes of irradiation, while initial [CCl₄] of 1.6 ppm (Figure 5) dropped to about 0.05 ppm at the same irradiation time. The concentration of residual CCl4 decreased exponentially with sonication time. With higher initial [CCl₄] within the same sonication period, greater decreases in residual CCl4 were observed; however, the destruction efficiency was about the same. First-order plots of ln[CCl₄] versus sonication time for various initial [CCl₄] are shown in Figure 6. The ultrasonic degradation of [CCl₄] apparently followed first-order kinetics within the current experimental concentration range. An average firstorder rate constant of $k = 0.7 \text{ min}^{-1}$ was determined from the slopes of these plots. It has been reported [15] that at low [CCl₄], CCl₄ destruction should follow a first-order reaction, but at higher concentrations this should become a zero-order reaction.

Temperature control in the reaction vessel is an important factor in maintaining a high destruction rate of CCl₄ in the solution. In sonochemistry, it has been reported [7] that one should not attempt reactions in a solvent (here, water) that is near its boiling point (100°C), because the rarefaction cycle causes the water to boil as a result of the reduced pressure generated; consequently, any cavitation bubbles formed will fill with water vapor almost instantaneously. This water vapor could reduce the extremes of temperature and pressure generated and thus decrease the direct destruction efficiency of the organics. On the other hand, the secondary reactions occurring in the liquid phase may be enhanced by operating the system at higher temperatures.

The effect of a steady-state temperature on the destruction efficiency of CCl₄ is shown in Figure 7. About 80% removal efficiency was observed for 4 minutes of irradiation; removal efficiency remained unchanged within a temperature range of 20-60°C. These results illustrate that, with the temperature within this range, increasing the steady-state temperature of the irradiation solutions seems to have little effect on the CCl4 destruction efficiency. In other words, manipulating the system at the optimum temperature range allows high removal effi-





FIGURE 5. [CCI₄] Decrease Versus Sonication Time; [CCl₄] = 8 ppm, 1.6 ppm, 0.53 ppm



FIGURE 6. First-Order Plot of In[CCI₄] Versus Sonication Time



FIGURE 7. Effect of Steady-State Temperature on CCI₄ Sonication

ciencies within reasonable operation times.

Ultrasonic power intensity is also an important factor affecting the CCl₄ destruction rate. In general, any increase in intensity will increase the sonochemical effect [16]. However, it must be realized that intensity cannot be increased indefinitely [17, 18]. With an increase in power intensity, the bubbles



FIGURE 8. Effect of Power Intensity on CCI₄ Sonication



FIGURE 9. Dependence of CCI₄ Destruction Rate on Power Intensity

may grow so large during rarefaction that the time available for their collapse is insufficient; consequently, the effective coupling of the ultrasonic energy to the system is reduced [7].

Figure 8 shows the effect of power intensity upon the destruction of CCl_4 . After 1 minute of irradiation, the residual [CCl_4] decreases with the increasing power intensity. More Ccl_4 molecules are destroyed at higher power intensities. The Ccl_4 destruction rate versus power intensity is shown in Figure 9. Within the current experimental range, the destruction rate seems to have a linear relationship to the power intensity. Although threshold intensity was not determined, it is estimated to be lower than the 0.95 W/cm² shown in this figure. The destruction rate is negligible if the power delivered into the solution is lower than the threshold intensity.

The relationship between initial pH value and CCl₄ destruction is shown in Figure 10. Residual [CCl₄] decreases with increasing pH value between pH 3 to pH 9 within the same sonication period. Better sonication efficiency was observed at higher initial pH values; however, such improvement increasingly diminishes as the pH is raised above 6. Since most of the irradiations were conducted at near neutral conditions for groundwater treatment, the initial pH value seems to have a minimal effect upon CCl₄ removal.

Modeling of ultrasonic chemical kinetics was also carried out on the basis of the proposed mechanism described above. If second-order kinetics are assumed, all of the reaction rate constants in the model can be obtained either from published data (reactions 1 to 27 [19-22]) or from experiments (reaction 28). Once determined, these constants are unchangeable and cannot be used as variables. Therefore, no adjustable parameters are included in the model. Reverse reactions were not considered significant because no other chemicals have been added so far and because the concentrations of the products for each reaction were assumed to be much less than those of the reactants.

All of the elementary reactions proposed in the mechanism were transformed into a set of differential equations that de-



FIGURE 10. Effect of Initial pH Value on CCI₄ Sonication

scribes the time variance of species concentrations on the basis of rate formulas (1) and (2), depending on whether these reactions take place in the bulk-liquid phase or in the cavitation holes. The k_c value in rate formula (2) was obtained by a best fit of the experimental data to the model, which was determined to be 2.5 * 10⁻¹¹ for the system. With the initial concentration



FIGURE 11. Effect of H₂O₂ Dosage on CCI₄ Sonication



FIGURE 12. Effect of H₂O₂ Dosage on CCl₄ with No Sonication

specified, these equations can be solved readily. Calculated results showed that the model fits the experimental data relatively well at low [CCl₄] (below 10 ppm). Correlation coefficients greater than 0.99 were obtained. However, deviations were observed at higher [CCl₄] (46.7 ppm). The predicted destruction rate was greater than that of the experiment. This might be because the organic vapor pressure in the bubble increases with increasing [CCl₄], and Henry's law no longer applies at this high concentration. In these cases, the activity for each compound in the bubble may need to be taken into consideration.

The sensitivity of each reaction in the model was also tested. Results reveal that the dissolved nitrogen concentration in the system has a minimal effect upon the CCl₄ destruction rate, an effect that is neglected. The major reactions affecting the CCl₄ destruction rate were found to be the dissociation of water molecules and the decomposition of CCl₄ molecules in the cavities. In the bulk-liquid phase, the reaction between OH radicals and CCl₄ was predominant, followed by the reactions of the atomic hydrogen, atomic oxygen, and finally HO₂ radicals.

Hydrogen peroxide (H_2O_2) , used as an oxidant, was added to the sample solution to study the effect of oxidant in the ultrasonic field. With sonication, H_2O_2 is known to decompose in the collapsing cavitation bubbles to yield OH radicals. These radicals diffuse into the bulk liquid and increase the radical concentrations in the solution, thus enhancing the destruction rate of organics.

Figure 11 shows the CCl₄ destruction versus H_2O_2 dosages (as $[H_2O_2]/[CCl_4]$ ratio), with 2 minutes of irradiation. Initial amounts of $[CCl_4]$ as high as 6.5 ppm were reduced to around 0.9 ppm without adding any oxidant and were slowly reduced to about 0.7 ppm as the H_2O_2 dosage increased to 20:1. This decreasing amount is insignificant when compared with the addition of H_2O_2 dosages.

Figure 12 shows CCl_4 destruction versus time with the addition of only H_2O_2 , without ultrasonics. Only slightly de-

creasing amounts of [CCl₄] were observed after 60 minutes reaction time, and the effect was negligible.

These results illustrate that the addition of the H2O2 oxidant has almost no effect upon the CCl4 destruction rate, both with and without ultrasonics. This can be explained by comparing the CCl4 reaction rate constants between the bulk-liquid phase (less than $10^7 \text{ M}^{-1} \text{ min}^{-1}$) and the cavitation hole (around 10^{12} M^{-1} min⁻¹). The CCl₄ is relatively inactive to radicals because all of the four free electrons are captured by chloride ions and form only single bonds (C-Cl) in the molecule. However, the single bond between C and Cl provides low dissociation energy (only about 80 kcal/mole), thus making it easier for the bonds to cleave in the cavitation holes. Therefore, the bulk-liquid reaction rate constant for CCl4 is about five orders of magnitude smaller than that in the cavities. This difference makes the reactions in the cavitation holes predominant; therefore, the addition of oxidants has only a minimal effect upon the whole system.

CONCLUSIONS

This study demonstrates that the ultrasonic process is a promising and effective method for the destruction of low concentrations of chlorinated organic compounds such as CCl4 in water. Removal efficiencies of greater than 99% were achieved through this process. For the treatment of water containing CCl₄, the major reactions may be the bond-cleavage of water and of CCl4 in the cavitation hole. Under the current experimental conditions, physical operating conditions, such as steady-state temperature and pH value of the irradiated solution, were found to have little effect upon the CCl4 destruction rate. However, the CCl4 destruction rate was found to be significantly affected by the intensity of the ultrasonic energy, with the destruction rate increasing proportionally to the intensity. By contrast, adding hydrogen peroxide as an oxidant had only a negligible effect upon the destruction rate, with or without ultrasonic irradiation.

Finally, mathematical modeling of ultrasonic chemical kinetics via a series of elementary reactions appears to be both a possible and valuable predictive tool for further exploring the means to control the complex ultrasonic process for the detoxification of groundwater.

ACKNOWLEDGMENT

This work was supported by the U.S. Department of Energy, Director of the Environmental Restoration and Waste Management, Office of Technology Development, under contract W-31-109-Eng-38.

LITERATURE CITED

- Richards, W. T., Loomis, A. L., J. Am. Chem. Soc., 49, p. 3086, (1927).
- 2. Suslick, K. S., Science, 247, p. 1439, (1990).
- Mason, T. J., Lorimer, J. P., "Sonochemistry: Theory, Applications and Uses of Ultrasound in Chemistry," John Wiley & Sons, New York, (1988).
- Weissler, A., Cooper, H. W., Snyder, S., J. Am. Chem. Soc., 72, p. 1976, (1950).
- Agranonik, Ya. R., Karpukhin, V. F., Fayngold, Z. L., "Antibiotiki i Khimioterapiya," Russ., 35, p. 48-51 (1986).
- Cheung, H. M., Bhatnagar, A., Jansen, G., Environmental Science & Technology, 25, (8), p. 1510, (1991).
- Mason, T. J., "Chemistry with Ultrasound," published for the Society of Chemical Industry, Elsevier Science Publishers Ltd., England, (1990).
- Suslick, K. S., Hammerton, D. A., Cline, D. E., Jr., J. Am. Chem. Soc., 108, p. 5614, (1986).
- Shutilov, V. A., "Fundamental Physics of Ultrasound," Gordon & Breach Science Publishers, New York, (1988).
- Kotronarou, A., Mills, G., Hoffmann, M. R., J. Phys. Chem., 95, p. 3630-3638, (1991).
- 11. Fitzgerald, M. E., Griffing, V., Sullivan, J., J. Chem. Phys., 25, p. 926, (1956).
- 12. Margulis, M. A., Didenko, Yu. T., Russian J. of Phys. Chem., 58, (6), p. 848, (1984).
- Glaze, W. H., Kang, J., Ind. Eng. Chem. Res., 28, p. 1573-1580, (1989).
- Margulis, M. A., Russian J. of Phys. Chem., 48(11), p. 1653, (1974).
- Chen, J. W., Chang, J. A., Smith, G. V., Chemical Engineering Progress Symposium Series, 67(109), p. 18.
- 16. Chen, J. W., Kalback, W. M., Ind. Eng. Chem. Fund., 6, p. 175, (1967).
- Rozenberg, L. D., Soviet Physics-Acoustics, 11(1), p. 110, (1965).
- Fogler, H. S., Barnes, D., Ind. Eng. Chem. Fund., 7, p. 222, (1968).
- Tsang, W.; Hampson, R. F., "Chemical Kinetic Data Base for Combustion Chemistry," J. Phys. Chem. Ref. Data, 15(3), p. 1087-1279, (1986).
- "Rate Constants of Gas Phase Reactions," U.S. Department of Commerce, National Bureau of Standards, COM-72-10014.
- Baulch, D. L., et al., "Evaluated Kinetic Data for High Temperature Reactions," J. Phys. Chem. Ref. Data, (1981).
- Johnston, H. S., "Gas Phase Reaction Rate Theory," Ronald Press Company, New York, (1966).

Remediation of Dichloromethane (DCM) -Contaminated Ground Water

Paul E. Flathman, Douglas E. Jerger, and Patrick M. Woodhull

Bioremediation Department, OHM Remediation Services Corp., Findlay, Ohio 45839-0551

This case history describes the physical and biological treatment of dichloromethane (DCM)-contaminated ground water following the rupture of an underground pipeline which contaminated an estimated 11,000 m³ (14,000 yd³) of soil and ground water in the early fall of 1983. Air stripping DCM from recovered ground water was initiated and provided an estimated 97% reduction in the ground water concentration of DCM. When it became evident that physical treatment alone would no longer be effective in removing residual DCM from the ground water environment, the practice of air stripping DCM from recovered ground water was terminated. Biological treatment was initiated and provided greater than a 500,000-fold reduction in the ground water concentration of DCM. Biological treatment had far exceeded the ability of physical treatment alone to remediate a ground water environment contaminated with a biodegradable contaminant.

INTRODUCTION

On August 16, 1983, an investigation was undertaken by a client to examine a water main pressure loss at one of its plants. A ruptured water main was discovered in soil saturated with DCM. The DCM had caused deterioration of a polyvinyl chloride (PVC)-constructed water main. The discovery was made that a buried DCM line, located in close proximity to the water main, had ruptured and an undetermined amount of DCM had leaked into the soil and ground water.

An emergency response site investigation was initiated following the discovery of the DCM leak. Interceptor trenches were constructed to contain free-standing product. Free-standing product was collected using vacuum equipment and was staged in vessels for eventual on-site treatment. Trenches were also constructed to determine the extent of lateral migration. Coarse gravel lenses and utility lines in the area permitted DCM to move laterally with little resistance. Monitoring wells were installed. They established that the contamination was within a confining clay layer at the 6-meter (20-ft) level and that the contamination did not reach the drinking water aquifer located at 30 meters (100 ft). The soil type to the 6-meter (20-ft) clay lens varied from sand with some gravel and a little silt to silty sand and silty clay. Monitoring wells were utilized in the perched aquifer to identify the contaminant plume. Ground water pumping techniques were then employed to contain the spilled DCM.

The emergency response to the spill involved only containment. An investigative phase was then initiated to examine various alternatives for environmentally restoring the site. The spill area was judged to contain $11,000 \text{ m}^3$ ($14,000 \text{ yd}^3$) of DCM-contaminated soil. Following removal of the 120 m^3 (160 yd^3) of highly contaminated soil to a Class A secure landfill, which represented an estimated 1.1% of the contaminated soil in the spill area, positive placement/suction lift techniques were utilized for recovery and treatment of contaminated ground water.

PHYSICAL TREATMENT

Granular activated carbon adsorption and packed column air stripping were considered for physical treatment of recovered ground water. The selection of a remedial treatment alternative is based upon the volume and concentration of contaminated material to be treated, discharge criteria, total capital investment, and annual operating and maintenance costs. At this site, the anticipated DCM influent concentration



FIGURE 1. Schematic for the physical treatment of DCMcontaminated ground water.

was the deciding factor in the determination of the most appropriate treatment technology.

Based upon a projected influent DCM concentration of 1,000 ppm following phase separation, an estimate was made of the theoretical carbon exhaustion rate to achieve a DCM effluent concentration in the low ppm range. Carbon isotherm data were derived from the USEPA's Treatability Manual [1] and were extrapolated to the higher concentration anticipated. The result was an estimate of 10% for the concentration loading on carbon which was equivalent to the adsorption of 100 mg DCM/g carbon. From previously performed pumping and drawdown tests, the predicted flow rate through the treatment system was 10 to 15 gpm. That prediction yielded a carbon exhaustion rate of 150 lb/day. Furthermore, any slug of DCM to the system would expend the activated carbon at an even faster rate. The additional cost of transporting and disposing of the spent carbon, as well as the additional cost of replacing fresh carbon in the contactor cells, led to the decision not to use granular activated carbon for on-site treatment.

Air stripping was selected as the preferred treatment technology for removal of DCM from the recovered ground water. That decision was based on several important factors. The first factor was the documented ease of stripping DCM from contaminated recovered ground water in past emergency and remedial field operations. Second, the regulatory agencies did not require the implementation of vapor-phase scrubbing units to cleanse air emissions from aeration devices. Third, unlike granular activated carbon, the use of air stripping is generally maintenance-free and regeneration and/or disposal of contaminated media would not be necessary. Finally, several packaged air stripping units were available for immediate dispatch to the site.

To achieve DCM removal efficiencies exceeding 99.9%, the influent wastestream to the air stripper would require heating prior to stripping. Live plant steam was available. Only the addition of a heat exchanger with the necessary controls and instrumentation was required.

Feasibility Evaluation

Characterization and laboratory testing of recovered ground water were performed prior to the final selection of the packed column air stripping treatment alternative [2]. The results of those determinations would define the need for pretreatment prior to air stripping. A particular treatment process may be cost-effective, but the pretreatment costs may render that process cost-prohibitive.

At this site, the treatment system effluent could not exceed 20 ppm for discharge into the plant's wastewater treatment system. The design procedure used to estimate the expected DCM removal efficiency in a packed column air stripper utilizing a nine-foot packing depth has been previously described [2]. With an assumed maximum influent DCM concentration of 16,500 ppm (the estimated solubility at 10°C), the packed



FIGURE 2. Removal efficiency for air stripping DCM from ground water as a function of column operating temperature.

column would be required to achieve a removal efficiency of 99.98% to meet the established discharge criteria of 20 ppm. By fixing liquid and air flow rates and packing height, a predicted column operating temperature of 40° C would be required. In the absence of experimental data to validate the predicted efficiency, the air stripping column was designed for an operating temperature of 40 oC.

Field Implementation

Figure 1 is a schematic of the treatment system designed to treat recovered ground water at a rate of 38 to 57 L/min (10-15 gpm). For removal of sand and other particulates, the ground water was pumped through a downflow mixed-media filter containing anthracite, silica sand, and pea gravel. The media were layered in the filter. The graduation in media pore size from large to small in the direction of flow provided longer filter runs before particulate binding necessitated backwashing of the bed. With backwashing, trapped sediments were recycled back to an influent holding pool.

Ground water leaving the mixed-media filter was piped to a phase separation/clarification/flow equalization vessel. Hydraulic residence time (HRT) in the vessel was fourteen hours. The vessel was a rectangular holding tank that allowed the denser DCM to separate from the water by gravity. Sludge collection equipment removed accumulated solids that escaped the multimedia filter, and float controls in the last chamber maintained a constant flow through the remainder of the treatment system. Liquid from the flow equalization vessel was pumped through a skid-mounted shell-and-tube heat exchanger to raise the water temperature from 10°C to more than 40°C. The heated water was then pumped to the top of the air stripping column for the removal of soluble DCM.

To volatilize DCM in the stripping column, contaminantfree air was passed countercurrent to the wastestream flow. The contaminant-laden air was vented to the atmosphere through a 3.0-meter (10-ft) stack attached to the top of the stripping column. Treatment of the contaminated air with vapor phase carbon was not required. Treated water leaving the stripping column was piped to the plant's wastewater treatment system for further processing. The operation of the treatment system, consisting of mixed-media pressure filtration, phase separation/clarification/flow equilization, and heated volatile air stripping, continued for three months.

The data collected during the operation of the project demonstrated that DCM, as a pure product, settled in the phase separator together with particulate matter that was not trapped by the mixed-media filter. The total volume of pure product collected in that manner did not exceed more than a few liters (gallons). Volatilization of DCM was apparent at the air/water interface in the flow equalization vessel. The DCM concentration entering the air stripping column rarely exceeded 150 ppm. The mixed-media filter removed virtually all particulate matter in the wastestream, and backwashing of the filter was infrequent. The stripping column operating temperature was



FIGURE 3. Physical removal of DCM in ground water from monitoring well B-5.

maintained between 27 and 60 °C. Samples analyzed to evaluate the air stripper efficiency were collected from sample taps located close to the influent and effluent ports of the column. The removal efficiencies actually observed during the threemonth operation of the air stripper are presented in Figure 2. The theoretical removal efficiency was calculated from the mass transfer model of Onda and coworkers [3, 4].

First-order kinetics were assumed in calculating the removal rate of DCM from MW B-5 (Figure 3). Monitoring well B-5 was located 6 meters (20 ft) downgradient from the ruptured DCM line (Figure 4). A 50% reduction in the ground water concentration of DCM was observed within the first eleven days of physical treatment. Within two months of field operation, air stripping DCM from recovered ground water resulted in an estimated 97% reduction in the ground water concentration of DCM. By the middle of the third month, it became evident that the effectiveness of physical treatment had appreciably diminished in removing residual DCM from the ground water environment. At the end of the third month, the practice of air stripping DCM from recovered ground water was terminated.

BIOLOGICAL TREATMENT: 1983/1984

Biodegradation of DCM under aerobic conditions in the laboratory was well documented [5-8]. Prior to on-site biological treatment, however, it was necessary to establish the biodegradation potential of DCM in the site matrix.

Feasibility Evaluation

OXYGEN ELECTRODE

SWITCH

Screening analyses were performed on seventeen selected ground water samples collected from the site on November 17, 1983. These analyses included (1) DCM concentration, (2) aerobic heterotrophic bacterial population density, (3) pH, and

HYDROGEN





1987. FIGURE 5. Electrolytic respirometer treatment vessel.

(4) available mineral nutrients (i.e., NH₃-N, NO₃-NO₂-N, and PO₄-P). The analytical results for the screening analyses demonstrated that the DCM-contaminated ground water contained an indigenous bacterial population. The geometric mean (GM) for the seventeen samples analyzed was 2.0×10^4 colony-forming units (CFU)/mL. The GM for the pH of the well samples analyzed was 7.7 which was acceptable for bacterial growth.

The results of the screening analyses also established that NH_3 -N and PO_4 -P mineral nutrient additions would be necessary for effective biological treatment of the contaminated ground water; NH_3 -N, NO_3 -NO₂-N, and PO_4 -P were not detected in any of the ground water samples analyzed. The analytical limits of detection (LODs) for these mineral nutrients were respectively 0.2, 2.0, and 0.06 ppm (Hach Company; Loveland, CO). Nitrogen and phosphorus are essential elements required by bacteria to support enhanced growth on available organics. A deficiency of either of these elements would limit growth on DCM. These elements are generally required in a 100:10:11 ratio of available C:N:P [9]. At a DCM concentration of 300 ppm, a minimum of 4.2 and 0.42 ppm NH₃-N and PO₄-P would respectively be required.

The biodegradation potential of DCM in site matrix ground water was established using the electrolytic respirometer (A.R.F. Products; Raton, NM). DCM loss and chloride release, resulting from the biodegradation of DCM in treatment vessels were used as the test parameters (Figure 5). Chloride concentration was determined using an Orion 96-17B combination chloride electrode with an Orion 901 microprocessor ionalyzer (Orion Research; Cambridge, MA).

The water sample used for the biodegradation study was collected from an injection pool containing recovered ground water. Sludge collected from one of the plant's industrial wastewater treatment basins was used to inoculate respirometer treatment vessels. The sludge in the basin was reported to have been continuously exposed to low levels of DCM for years and was, therefore, thought to contain populations of bacteria acclimated to DCM.

Table 1 presents the experimental design for the biodegradation study. DCM was added to all treatment vessels to a final concentration of 100 ppm. A concentration of 100 ppm approximated the average DCM concentration found in ground water at the site. Vessels 1 and 2 were replicates which demonstrated the biodegradation potential of DCM. Any nonbiological loss of DCM from treatment vessels would be quantified with the abiotic control (that is, vessels 3 and 4). The abiotic control received the microbial poisons KCN and NaN₃ to final concentrations of 320 ppm [10]. Using procedures described by Young and Baumann [11], the respirometer treatment vessels were prepared by adding diammonium phosphate and sodium dihydrogen phosphate to each treatment vessel. Nitrification was not suppressed.

Aliquots (40 mL) were periodically removed from treatment vessels and analyzed for DCM and chloride. DCM loss and chloride release were similar in the biological treatment vessels. The analytical results for one of the biological treatment vessels (vessel 1) are presented in Figure 6. Neither DCM loss nor

Table 1 Experimental Design for Treatments in Electrolytic Respirometer Treatment Vessels Used to Establish the Biodegradation Potential of DCM in a Representative Ground Water Sample

Vessel Number	Treatment
	Natural Bacteria Flora
1,2	Groundwater + DCM + basal
	salts + sludge inoculum
	Abiotic Control
3,4	Groundwater + DCM + basal
	salts + sludge inoculum + poisons



FIGURE 6. Biodegradation of DCM with chloride release in an electrolytic respirometer treatment vessel containing a representative ground water sample.

chloride release was observed in the abiotic control. Bacterial acclimation to DCM occurred within one day. Bacteria from the plant's wastewater treatment system were later used for startup of an aboveground biological reactor for the treatment of recovered ground water prior to reinjection into the DCM-contaminated soil/ground water environment.

Prior to evaluating the biodegradation potential of DCM using a bacterial inoculum from the plant's wastewater treatment basin, bacterial inocula obtained from soil in the vicinity of the ruptured pipeline and from the activated sludge of a domestic wastewater treatment plant were used for the biodegradation study. Within the two week time period the inocula were evaluated, biodegradation of DCM was not observed. With no space available to set up additional respirometer treatment vessels, the decision was made to terminate that attempted acclimation to DCM and to immediately initiate an evaluation of the ability of bacteria from the plant's wastewater treatment basin to biodegrade DCM. The plant's wastewater treatment basin had been receiving low concentrations of DCM for years. With that approach, acclimation to DCM was demonstrated within one day (Figure 6).

The theoretical chloride release for mineralization of DCM was calculated from the following relationship:

$$CH_2Cl_2 + O_2 = CO_2 + 2HCl$$

For each milligram of DCM mineralized, 0.835 milligrams of chloride would be released. DCM was biodegraded stoichiometrically to chloride in replicate respirometer treatment vessels (Figure 6). Using the following relationship, 130 and 120% of the theoretical chloride release for mineralization of DCM was met in respirometer treatment vessels 1 and 2, respectively:

 $\frac{\text{Actual chloride release}}{\text{Theoretical chloride release}} \times 100$

Analytical results from the biofeasibility evaluation indicated that the DCM was completely oxidized to carbon dioxide with chloride release. The lack of an appreciable lag period indicated that naturally occurring bacteria from the plant's wastewater treatment basin were adapted to DCM.

Field Implementation

A ground water recovery and treatment system, similar to that described for other biological remediation projects [12], was put into immediate operation. The injection/recovery wells used for the air stripping operation were now utilized for biological treatment. Air stripping of DCM from recovered ground water was replaced with biological treatment utilizing a suspended growth system for aboveground treatment.

In addition to providing efficient biological treatment, this system permitted wasting of adapted bacteria into the contaminated soil/ground water environment. The injection system was used to introduce adapted bacteria into that environment and to provide the chemical environment necessary to support enhanced bacterial growth. The recovery system was used to withdraw DCM-contaminated ground water for aboveground treatment. Effluent from the treatment system was then reinjected into the subsurface environment. Biodegradation of DCM took place in the soil/ground water environment as well as aboveground in the biological treatment system.

On day 82, batch treatment of DCM-contaminated ground water was initiated. To maintain a DCM-adapted bacterial population, air containing DCM vapor was bubbled into the bioreactor. Between days 82 and 87, an 80% increase in chloride concentration in the bioreactor was observed as a result of chloride release from the biodegradation of DCM (Figure 7). Between days 87 and 90, the contents of the bioreactor were drained into an injection pool and then the bioreactor was refilled. Injection of DCM-adapted bacteria into the soil/ ground water environment began on day 89.

Batch treatment of DCM-contaminated ground water was again performed in the bioreactor between days 90 and 92. A 110% increase in chloride concentration was observed as a result of DCM biodegradation. The bioreactor was again drained into the injection pool and then refilled. On day 95, continuous biological treatment was initiated. Over the tenday operational phase of continuous biological treatment, the effluent chloride concentration averaged 131 \pm 65% ($\bar{x} \pm s$, n = 9) greater than the influent concentration.

The removal of DCM from the ground water environment was rapid. In MW B-5 (Figure 8), a downgradient well located 6 meters (20 ft) from the DCM spill, a 50% reduction in DCM concentration was observed within eight days of biological treatment. The rate of decrease was assumed to be first-order [13]. With the onset of winter, biological treatment at the site was temporarily suspended. Winterization of the treatment system would have been required for continuation of active biological treatment.

At that time, a 97% reduction in DCM concentration was observed since the biological treatment was initiated. Theoretical chloride release for biodegradation of 186 ppm of DCM (that is, from 192 to 6 ppm DCM) is 155 ppm. The observed chloride release was 156 ppm; i.e., the chloride concentration in MW B-5 increased from 175 ppm on day 80 to 331 ppm on day 123. Biological treatment, therefore, was considered responsible for the removal of DCM from that ground water environment.

Active biological treatment was suspended on January 2, 1984, and did not resume until August 1987, when the residual



FIGURE 7. Chloride release in the bioreactor resulting from biodegradation of DCM.

DCM contamination remaining in the soil and ground water was addressed by the resumption of full-scale on-site biological treatment.



FIGURE 8. Biological removal of DCM in ground water from monitoring well B-5 as a function of time.

Table 2 Analytical Results of Screening Analyses Performed on DCM-Contaminated Ground Water Samples Collected from the Site in August 1985 and Used to Establish the Biodegradation Potential of DCM by the Indigenous Microflora

Location	DCM (ppm)	Chloride (ppm)	NH3-N (ppm)	PO₄-P (ppm)	pН	Aerobic Heterotrophic Bacteria (CFU/mL)
B-4	90	510	0.84	< 0.06	9.7	3.1×10 ⁵
B-5	9.5	200	32	< 0.06	8.6	8.0×10^{6}
B-6	14,000	490	3.9	< 0.06	7.2	5.4×10^{2}
B-17	73	450	4.5	< 0.06	6.6	1.4×10^{2}
WWTB ¹	20	990	2.2	< 0.29	8.6	3.1×10^{6}

BIOLOGICAL TREATMENT: 1987/1988

Feasibility Evaluation

Prior to renewal of full-scale treatment, a biofeasibility study was performed in August 1985 to determine the potential of the indigenous microflora in the DCM-contaminated soil/ ground water environment to biodegrade DCM. Ground water samples from MWs B-4, B-5, B-6, and B-17 and water from the plant's wastewater treatment basin were used for the study (Figure 4). Screening analyses were performed on those samples. The results of those analyses are presented in Table 2.

Analytical results of the screening analyses indicated that, with the exception of ground water from MW B-6, the environments tested did not appear toxic to bacterial growth. The bacterial plate count for B-6 was 5.4×10^2 CFU/ml. The DCM concentration in B-6 was 14,000 ppm. The plate counts for the remaining locations ranged from 3.1×10^5 to 8.0×10^6 CFU/ml. The DCM concentration in those locations ranged from 9.5 to 90 ppm. The high DCM concentration may have been responsible for the lower bacterial population density observed in B-6.

The analytical results of the screening analyses also indicated that mineral nutrient addition and/or pH adjustment would be required for enhanced biological treatment. Orthophosphate-phosphorus was not detected in any of the ground water samples analyzed. Only 50% of the ground water samples analyzed were within a pH range generally considered acceptable for enhanced bacterial growth.

The biodegradation potential of DCM in the collected samples was evaluated using the electrolytic respirometer (Figure 5). For the study, mineral nutrient concentrations and pH were maintained within acceptable ranges for enhanced bacterial growth. DCM loss and chloride release in treatment vessels were used as the test parameters. A two-phase study was performed. DCM-basal salts-yeast extract (5 ppm) medium [10], inoculated with washed (0.02 M phosphate buffer, pH 7.2) cell suspensions from each of the samples collected, was used for Phase 1. The objective of this phase of the study was to evaluate the potential of the indigenous microflora to biodegrade DCM. Adapted microorganisms from Phase 1 were used in Phase 2 as inocula to evaluate the biodegradation potential of DCM in the site matrix samples. Abiotic controls were prepared for the Phase 1 and Phase 2 studies to quantify nonbiological loss of DCM from treatment vessels.

Throughout the Phase 1 study, the environment remained favorable for bacterial growth (Table 3). Biodegradation of DCM was demonstrated in treatment vessels containing inocula from B-5, B-6, B-17, and the wastewater treatment basin after sixteen, twenty, seven, and twenty days, respectively. Biodegradation of DCM in a treatment vessel inoculated with a cell suspension from B-4 was not demonstrated after thirty-four days.

The biodegradation potential of DCM in site matrix samples was similarly determined in the Phase 2 study. Washed cell suspensions from Phase 1 treatment vessels were added to the treatment vessels containing the site matrix samples. Ground water from B-6 was diluted with laboratory-grade water to 10% (vol/vol) full strength since the high DCM concentration in the site matrix sample was suspected to be the cause of the low bacterial population density observed in ground water at that location. Biodegradation of DCM was demonstrated within seven days in all site matrix samples analyzed.

Based on the results of the feasibility evaluation, the conclusion was made that the indigenous microflora may be utilized for remediation of the DCM-contaminated ground water matrices analyzed. For enhanced biological treatment, oxygen and mineral nutrient addition and pH adjustment would be required.

Table 3	DCM, Chloride, NH ₃ -N, and PO ₄ -P Concentrations, pH, and Aerobic Heterotrophic Bacterial Population Density
	in Electrolytic Respirometer Treatment Vessels for the Phase 1 Study

Inoculum	DC (pp	CM m)	Chlo (pp	oride m)	NH (pp	3-N m)	PO (pp	₄-P m)	pl	н	Aerobic He Bac (CFU	eterotrophic teria /mL)
Source	Initial	Final	Initial	Final	Initial	Final	Initial	Final	Initial	Final	Initial	Final
B-4	130	81	1.6	1.6	180	200	470	450	7.5	7.5	1.2×10^{4}	5.9×10 ⁴
B-5	160	<1.0	1.9	170	220	180	520	440	7.0	6.3	1.9×10^{5}	1.7×10^{7}
B-6	210	<1.0	2.2	170	190	200	490	400	7.5	6.8	4.6×10^{1}	2.0×10^{6}
B-17	140	<1.0	2.4	200	190	180	480	420	7.1	6.3	8.7×10^{5}	1.1×10^{7}
WWTB ¹	200	<1.0	2.7	170	190	200	490	430	7.5	7.0	4.2×10^{4}	4.2×10^{6}
B-4 ²	210	99	500	500	170	200	490	430	8.3	8.4	$< 1.0 \times 10^{1}$	$< 1.0 \times 10^{1}$

Wastewater Treatment Basin

Abiotic Control





FIGURE 10. DCM concentration as a function of time in observation well OW-3 from August 5, 1987 to May 24, 1988 (day 295).

FIGURE 9. Site schematic for the 1987/1988 field operation.

Field Implementation

Prior to initiating the field operation, the analytical data from monthly analyses of ground water samples collected between June and December 1986 from wells B-4 through B-7, B-10, B-12, B-17 through B-20, MW-10 through MW-17, and recovery well RW-11 (Figure 4) were reviewed. Those samples were analyzed for DCM, chloride, and pH.

Based on the results of those analyses, a ground water injection/recovery system was installed in July 1987. The system focused on remediating the area enclosed by MWs B-4, B-6, B-12, and B-17. The DCM concentration in those wells in 1986 approximated 4,000, 10,000, 300, and 1,000 ppm, respectively. The size of the area requiring remediation was seven times smaller than that addressed in 1983. To effectively apply remediation within that area, additional observation wells (that is, OW-1 through OW-3), injection wells (that is, IW-1 through IW-4), and recovery wells (that is, RW-1 through RW-6) were installed (Figure 9).

A vacuum recovery unit was utilized for the recovery of DCM-contaminated ground water. From the vacuum recovery unit, ground water was pumped to a flow equalization chamber which was also used for nutrient addition, pH adjustment, and aeration. The treated water was then reinjected into the soil/ground water environment for enhanced *in situ* biological treatment. The flow through the aboveground system ranged from 7.5 to 15 L/min (2-4 gpm).

The DCM concentrations in monitoring wells at the initiation of the field operation in August 1987 and after eight months

Table 4	DC	A Con	centration	in	Monitori	ing	Wells	at the
Initiatio	n of	Field	Operation	in	August	198	7 and	After
Eigh	t Mo	nths e	of Enhance	d	Biologica	al Ti	reatm	ent

	DCM Concentration (ppm)			
Monitoring Well	August 1987	April 1988		
OW-1	11	< 0.005		
OW-2	<2	< 0.005		
OW-3	5,200	130		
B-4	3,200	0.35		
B-5	<2	< 0.005		
B-17	1,500	< 0.005		
B-18	<2	< 0.005		







FIGURE 12. DCM concentration as a function of time in monitoring well B-17 from August 5, 1987 to March 30, 1988 (day 238).

of field operation are presented in Table 4. Figures 10 through 12 present the DCM concentration in OW-3, MW B-4, and MW B-17 as a function of time beginning on August 5, 1987. The spikes of DCM observed in these wells are attributed to the flushing of pockets of DCM within the unsaturated zone (that is, vadose zone) into ground water by the injection system. Within eight months, the DCM concentration in the monitoring wells was reduced as much as 300,000 fold. Active biological treatment was suspended on May 24, 1988, when DCM was reduced to less than 2 ppm in all monitoring wells at the site. By June 1, 1988, DCM was reduced to less than 10 ppb in all monitoring wells at the site with the exception of MW B-4 [14]. Currently, the DCM concentration in MW B-4 ranges between 0.05 and 0.5 ppm. The state regulatory agency will consider the project completed when DCM remains less than 10 ppb in MW B-4 over a three-month period based on monthly analysis of ground water from that well.

ACKNOWLEDGMENTS

For preparation of this paper, the authors would like to thank Laura L. Duhigg, who prepared the figures, and Anne L. Hermiller, who did the typing.

LITERATURE CITED

- U.S. Environmental Protection Agency, Treatability Manual, Volume 1, Treatability Data, EPA-600/2-82-001a. Washington, DC: Office of Research and Development, U.S. Environmental Protection Agency (1982).
- Flathman, P. E., M. J. McCloskey, J. J. Vondrick, and D. M. Pimlett, *In Situ* Physical/Biological Treatment of Methylene Chloride (Dichloromethane) Contaminated Groundwater, Proceeding of the Fifth National Symposium and Exposition on Aquifer Restoration and Groundwater Monitoring, May 21-24, 1985, Columbus, OH, pp. 571-597.
- Bolles, W. L., and J. R. Fair, 1982, "Improved Mass-Transfer Model Enhances Packed-Column Design," *Chemical Engineering*, 89, pp. 109-116.

- Roberts, P. V., G. D. Hopkins, C. Munz, and A. H. Riojas, "Evaluating Two-Resistance Models for Air Stripping of Volatile Organic Contaminants in a Countercurrent, Packed Column," *Environmental Science and Technology*, 19, pp. 164–173 (1985).
- Klecka, G. M., "Fate and Effects of Methylene Chloride in Activated Sludge," *Applied and Environmental Micro*biology, 44, pp. 701-707 (1982).
- Tabak, H. H., S. A. Quave, C. I. Mashni, and E. F. Barth. "Biodegradability Studies with Organic Priority Pollutant Compounds," *Journal Water Pollution Control Federation*, 53, pp. 1503-1518.
- Brunner, W., D. Staub, and T. Leisinger, "Bacterial Degradation of Dichloromethane," *Applied and Environmen*tal Microbiology, 40, pp. 950-958 (1980).
- Rittman, B. E., and P. L. McCarty, "Utilization of Dichloromethane by Suspended and Fixed-Film Bacteria," *Applied and Environmental Microbiology*, 39, pp. 1225-1226 (1980).
- Mitchell, R., "Introduction to Environmental Microbiology," Englewood Cliffs, NJ: Prentice-Hall, pp. 172-187 (1974).
- Liu, D., W. M. J. Stracham, K. Thomson, and K. Kwasniewska, "Determination of the Biodegradability of Organic Compounds," *Environmental Science and Technology*, 15, pp. 788-793 (1981).
- Young, J. C., and E. R. Baumann, "The Electrolytic Respirometer. I. Factors Affecting Oxygen Uptake Measurements," *Water Research*, 10, pp. 1031-1040 (1976).
- Flathman, P. E., D. E. Jerger, and L. S. Bottomley, "Remediation of Contaminated Ground Water Using Biological Techniques," *Ground Water Monitoring Review*, 9, pp. 105-119 (1989).
- Larson, R. J., "Role of Biodegradation Kinetics in Predicting Environmental Fate," In A. W. Maki, K. L. Dickson, and J. Cairns, Jr. (eds.), Biotransformations and Fate of Chemicals in the Aquatic Environment. Washington, D.C.: American Society for Microbiology, pp. 67-86 (1980).
- Flathman, P. E., D. E. Jerger, P. M. Woodhull, and D. K. Knox, "Remediation of Dichloromethane-Contaminated Groundwater," pp. 25-37. H. M. Freeman and P. R. Sferra, Editors, Innovative Hazardous Waste Treatment Technology Series, Volume 3, Biological Processes. Lancaster, PA: Technomic Publishing Co. (1991).

Effect of Chemical Pretreatment on Anaerobic Biodegradation of Refractory Organic Compounds

Yi-Tin Wang

Department of Civil Engineering, University of Kentucky, Lexington, KY 40506

The potential for using chemical oxidation to enhance anaerobic biodegradability and reduce toxicity of two model phenolic compounds (o-cresol and 2,4-DNP) was evaluated. Batch bioassays were performed on the model compounds and their oxidation samples to determine biodegradability and toxicity in batch methanogenic cultures. Ozone, hydrogen peroxide, and potassium permanganate were the three oxidants examined in this study. A dose of approximately 5 moles hydrogen peroxide, in the presence of a ferrous iron catalyst (Fenton's reagent), 7 moles permanganate, or 25 moles of ozone per mole of o-cresol was needed to significantly enhance anaerobic biodegradability of oxidation samples. Approximately 2.5-4.5 moles of hydrogen peroxide or 7 moles of ozone per mole of 2,4-DNP reduced methanogenic toxicity by 50%.

INTRODUCTION

Phenolic compounds represent the major constituents present in a wide spectrum of industrial wastewaters such as coal conversion, coke preparation, petroleum refineries, pulp and paper, and photoprocessing [1]. Many of these compounds can cause adverse effects on human health and receiving waters if discharged without proper treatment. Effective treatment of these materials has been demonstrated by chemical oxidation using oxidants such as ozone [2-7], hydrogen peroxide [8, 9], and permanganate [10]. Reaction mechanisms of phenols with these oxidants have been proposed by several researchers. Permanganate oxidation in acid and neutral solutions may be accomplished by permanganate ions and several ions derived from the tri- and tetravalent states of manganese [11]. In basic solutions, the oxidation of phenols may be initiated by transfer of an electron from the phenolate ion to permanganate to form phenoxy free radicals. The phenoxy free radicals so formed are then involved in coupling and oxidation reactions resulting in formation of o- and p-diphenoquinones as the intermediates [10]. The strong oxidation power of hydrogen peroxide, however, is attributed to the formation of hydroxyl free radicals

in the presence of a ferrous iron salt [8]. The oxidation of phenol by this reagent may produce cathechol and hydroquinone which are further oxidized to muconic acid and the corresponding quinones [8]. The oxidation of phenol by molecular ozone, on the other hand, probably involves the electrophilic attack by ozone of the ortho, para-directive phenol molecular and 1,3 dipolar cyclo addition of ozone [12]. The electrophilic attack yields catechol and hydroquinone as the major intermediates while the dipolar cyclo addition causes direct ring cleavage to produce muconic acid or muconaldehyde. Ozonation of substituted phenols may involve oxidation of substituents groups. The methyl group of o-cresol may be oxidized to carboxylic acid upon initial ozonation and produce salicylic acid as the early ozonation intermediate. At pH above 8, ozone is unstable and tends to react with HOH to form highly reactive hydroxyl radicals which attack more rapidly and indiscriminately than ozone. Consequently, reaction mechanisms and intermediates formed during oxidation may differ depending on the media pH.

Chemical oxidation to achieve complete reduction in chemical oxygen demand (COD) or total organic carbon (TOC), however, would be exceedingly costly and time consuming. Although biological oxidation may be the most cost-effective treatment method for high-strength wastewaters, its effectiveness is generally limited to nontoxic and biodegradable wastes. Chemical oxidation pretreatment of wastewaters containing biorefractory or toxic compounds followed by biological processes seems to be a logical approach to treat this type of waste.

Anaerobic biological treatment offers the advantages of no oxygen requirement, low waste biomass production, and the generation of methane gas. However, very little attention has been paid to assess the potential for using chemical oxidation to enhance anaerobic biological treatment of biorefractory or toxic materials. In this study, anaerobic bioassay of two model phenolic compounds and their chemical oxidation samples was performed to determine the effect of chemical oxidation on their biodegradability and toxicity.

MATERIALS AND METHODS

Phenolic Compounds

o-Cresol and 2,4-diinitrophenol (DNP) were the model compounds studied. These compounds were selected because previous work has shown that they are biorefractory and/or inhibitory to methanogenic cultures [13, 14]. All phenolic solutions were prepared by adding a predetermined amount of phenolic compounds to a proper volume of distilled and deionized water.

Chemical Oxidation Pretreatment

Ozone was generated from pure oxygen using a Welsbach Model T-816 Ozonator. Ozonation of model compounds was conducted in a 500 mL gas-washing bottle (6 cm OD \times 24 cm) fitted with a ground glass joint. A coarse sintered glass dispersion tube delivered ozone to the bottom of the reactor at an average rate of 0.23 mmol/min. During ozonation, the reactor content was stirred vigorously with a magnetic stirrer to maintain good mixing. The reaction of phenols with hydrogen peroxide or permanganate was conducted in 500 mL batch stirred reactors. Ozonation and permanganate oxidation were conducted with both unbuffered and buffered solutions. The buffered solutions were prepared by adding 0.02 M boric acid and 0.02 M sodium borate and adjusting pH to 9 with NaOH. Ferrous iron (FeCl₂) was added as a catalyst for hydrogen peroxide oxidation at an optimum concentration ratio of phenolic compounds to iron of 100 to 1 and the reaction pH values were within the optimum range of 3-4 [8, 9, 15]. The reactors were covered by alumimum foil during the course of oxidation and samples were immediately stored in dark bottles and refrigerated prior to analysis. Oxidant doses were expressed as mole oxidant/mole compound initially present.

Methanogenic Culture

A phenol-enriched methanogenic culture was used as the seed inoculum for all biodegradability and toxicity tests. This culture had been developed for more than 3 years from seeds obtained from the Department of Civil Engineering, University of Illinois (courtesy of Dr. Makram T. Suidan) and digested municipal wastewater sludge (West Hickman Treatment Plant, Lexington, Kentucky) using a 14-L, continuous flow fermentor (Model MF-114, New Brunswick Scientific Co., Edison, NJ). The feed solution contained 4000 mg/L phenol as the primary substrate and 5 mg/L each of other organic compounds (ocresol, m-cresol, p-cresol, 2-ethylphenol, 3-ethylphenol, 4ethylphenol, catechol, resorcinol, hydroquinone, salicylic acid, m-hydroxybenzoic acid, and p-hydroxybenoic acid) along with a salt and vitamin solution. The fermentor was operated at 35°C and 30-day liquid detention time with pH maintained at 7 using sodium bicarbonate. Table 1 summarizes the performance of the fermentor throughout this study.

Table 1 Summary of Performance of the Methanogenic Fermentor

	Average	Range
Total gas production, mL/day	1820	1460-2190
Gas composition, % methane carbon dioxide	63.6 35.4	61.9-66.5 31.5-37.1
Volatile suspended solids, mg/L	260	164-396
Phenol and other feed compounds, mg/L	BDL*	BDL
[*] below detention limit (<1 mg/L)		

Batch Bioassay

Two batch test procedures, the biochemical methane potential (BMP) and the anaerobic toxicity assay (ATA), were used to evaluate the anaerobic biodegradability and toxicity of oxidant treated samples, respectively. The procedures were a modification of Owen et al. [16] and involved the placement in a 125-mL serum bottle of 50 mL sample and 50 mL methanogenic culture while under the purge of an oxygen-free 30% CO₂ and 70% N₂ gas mixture. Sodium sulfide (500 mg/L) and L-cystine hydrochloride (500 mg/L) were added to provide a reducing environment.

Before being transferred to serum bottles, oxidation samples were prepared by removing excess oxidants. Dissolved oxygen and residual ozone were removed by purging the samples with nitrogen gas for 15 min. Any excess hydrogen peroxide was removed by reacting with added ferrous iron [17]. Residual permanganate was removed by first acidifying the samples to form MnO_2 (4 $MnO_4^- + 4H^+ = 3O_2 + 2H_2O + 4MnO_{2(s)}$). The solid MnO_2 so formed was then removed from solutions by filtering the samples through a 0.45 μ m membrane filter.

To evaluate toxicity to the specific bacterial groups in the mixed methanogenic culture, two types of substrates (acetic acid and phenol) were used in the ATA tests. Acetic acid was used to assess sample toxicity toward methanogenic bacteria while phenol was selected to estimate toxicity toward acetogenic bacteria in the phenol-enriched mixed culture. Acetic acid or phenol was used as the test substrate to allow for an independent evaluation of toxicity toward the specific bacterial groups in the methanogenic cultures. A consortium of acetogenic and methanogenic bacteria is responsible for the degradation of phenol to methane while conversion of acetate to methane is achieved by the methanogens alone [18, 19]. Three distinct bacterial morphotypes were also observed in the phenolenriched methanogenic culture used in this study. The following steps in the overall conversion of phenol to methane were proposed before [20]:

Phenol-degrading acetogens:

 $C_6H_6O + 5H_2O - 3CH_3COOH + 2H_2$ Acetate-utilizing methanogens: $3CH_3COOH - 3CH_4 + 3CO_2$ Hydrogen-utilizing methanogens: $2H_2 + 0.5CO_2 - 0.5CH_4 + H_2O$

Sum:

 $C_6H_6O + 4H_2O \rightarrow 3.5CH_4 + 2.5CO_2$

Based on the above stoichiometrical relationship, acetate is the major precursor of methane and it accounts for approximately 86% of methane produced during anaerobic degradation of phenol. Therefore, using phenol and acetate as test substrates may provide information on which bacterial group is more susceptible to inhibition caused by oxidant treated samples.

All samples were buffered at pH 7.0 with sodium bicarbonate

before finally sealed with butyl rubber stoppers and placed on a continuous shaker in a 35° C constant temperature room. Duplicates were run for all samples, including the controls which contained no chemical oxidation samples.

ANALYTICAL PROCEDURE

Gas samples were withdrawn from the test bottles with syringes at proper intervals for the determination of methane content in the BMP and ATA tests. Gas production was monitored by displacement of the plunger in an appropriate sized wetted glass syringe. Gas composition was determined with a gas partitioner (Fisher Model 1200) using certified calibration standards. Chemical oxidant treated samples were analyzed for COD, TOC, initial compound, and organic acids. COD was determined by using Method 508 of Standard Methods [21]. TOC was measured by chemical ultraviolet oxidation and infrared detection using a carbon analyzer (Dohrman Model DC-80). o-Cresol, 2,4-DNP, and salicylic acid were analyzed by a liquid chromatograph (Varian Model 5020) equipped with a variable wavelength UV detector and a MicroPak MCH-5 C18 reverse phase column at ambient temperature. Detection wavelength was set between 270 and 310 nm corresponding to the absorbance maximum of the compound. All organic acids except glyoxylic acid and salicylic acid were analyzed by a modified procedure of Salanitro and Muirhead [22] in which the samples were subjected to a butyl-esterification process. BF₂-Butanol was used as the butylating agent, and the dry salts of the acids were heated at 60°C for 20 min in a temperature bath. The butyl esters were then analyzed by a gas chromatograph (Varian Model 3400) with a 50 m \times 25 mm SPB-1 fused silica capillary column and a FID detector. The carrier gas was helium, and a split ratio of 90:1 was used. Oven temperature was increased from 50°C to 170°C at 12°C/min, then 20°C/min to 230°C. The detector and the injector temperature were both set at 250°C. Glyoxylic acid was analyzed by measuring spectrometrically the ionized species of glyoxylic dinitrophenylhydrazine at 513 nm [4].

BIODEGRADATION AND TOXICITY PARAME-TERS

Biodegradability of reaction samples was indicated by the ratio of the net and the potential methane production. The net methane production was determined by subtracting methane produced in the seed-blank bottles from that measured in test



FIGURE 1. Anaerobic biodegradability of oxidant treated o-cresol

bottles after incubation for at least 30 days. Experimental errors of determining methane production were estimated to be within $\pm 10\%$. The methane data were corrected for moisture content and converted to 0°C and 1 atm. The methane partial pressure of the methane in the gas phase was also included in the data. The methane potential was calculated based on the measured COD of the sample and a conversion factor of 0.35 CH₄/g COD at 0°C and 1 atm [23]. Inhibition to the acetate-utilizing methanogens was indicated by a decreased rate of methane production relative to the acetate objection with phenol degradation was indicated by a decreased rate of methane production relative to the phenol control.

RESULTS AND DISCUSSION

BMP Tests

Figure 1 demonstrates the anaerobic biodegradability of oxidant pretreated samples of unbuffered 200 mg/L o-cresol under a range of the three oxidant doses. Figure 1 shows that net methane production from the untreated o-cresol (control, at 0 mole oxidant/mole o-cresol) was less than 8% of its potential, indicating that o-cresol was not significantly degraded. However, the biodegradability of the treated o-cresol increases with increasing dose of oxidants. Although o-cresol might be degraded at a very low level as the o-cresol concentration in the effluent from the methanogenic fermentor was less than 1 mg/l, 200 mg/l o-cresol was used in the BMP tests to assess

	able 2 Summary of Chemical Oxidation Pretreatment of o-Cresol				
o-Cresol concentration (initial, mg/l)	Oxidant and dose (mole/mole o- Cresol)	End pH at highest dose	Oxidant needed for 100% removal (mole/ mole o-cresol)	Product identified	
200	O ₃ : 0.6-55.9	2.7	6.1	Formic, acetic,	
600	O ₃ : 0.6-49.6	2.5	5.0	propionic, glyoxylic,	
600	O ₃ : 0.6-49.6	8.3	4.1	oxalic and salicylic acid	
200	H ₂ O ₂ : 0.1-10	3.1	4.0	Formic, acetic, and	
600	H ₂ O ₂ : 0.1-10	2.8	5.0	oxalic acid	
200	KMnO₄: 0.1-10	3.3	3.0		
600	KMnO ₄ : 0.1-10	4.7			

	Table 3 Summary of Chemical Oxidation Pretreatment of 2,4-DNP				
2,4-DNP concentration (initial, mg/l)	Oxidant and dose (mole/mole 2,4- DNP)	End pH at highest dose	Oxidant needed for 100% removal (mole/ mole 2,4-DNP)	Product identified	
100	O ₃ : 10.6-190.4	2.8	21.1	Formic,	
100	O ₃ : 10.6-190.4	8.3	14.9	acetic, glyoxylic, and oxalic	
100	H ₂ O ₂ : 0.1-5	2.8	5.0		
100	KMnO₄: 0.1-10	2.7	> 5.0	_	
100	KMnO ₄ : 0.1-10	8.7	3.0		

the effect of chemical oxidation on a moderate strength wastewater.

The data in Figure 1 indicated that the observed increase in biodegradability of treated samples was faster with hydrogen peroxide or permanganate than with ozone based on oxidant doses. For example, for 1 mole of o-cresol, about 5 moles hydrogen peroxide (highest dose applied) or approximately 7 moles permanganate were needed to obtain 40% biodegradable products while approximately 25 moles ozone were needed to achieve the same biodegradability. The ozone dose far exceeded the observed requirement of 6.1 moles/mole for complete removal of nonbiodegradable 200 mg/L o-cresol by oxidation (Table 2), indicating slow formation of biodegradable products during ozonation. The data in Table 2 and Figure 1 also suggested that complete or near complete oxidation of o-cresol was needed to obtain significant methane production from the o-cresol carbon. In the case of 2,4-DNP, chemical oxidation (see Table 3) was ineffective to enhance biodegradability. All three oxidant treated samples were not degraded as no significant net methane production was noted in any of these sample bottles.

Several organic acids were identified as the oxidation products during the reaction with ozone and peroxide oxidation. An example of organic acids formation is shown in Figure 2 with ozone as the oxidant. Identified organic acids are summarized in Tables 2 and 3. These acids accounted for 21-85%and 18-33% TOC present after complete removal of o-cresol by ozonation and peroxide oxidation, respectively. For 2,4-DNP, organic acids identified accounted for 42-77% TOC after the disappearance of the initial compound during ozonation. Salicylic acid was detected during ozonation of o-cresol under relatively lower doses of ozone (≤ 3 moles/mole) but its level was never higher than 1 mg/L. Although permanganate treated samples were not analyzed for organic acids, formation

FIGURE 2. Organic acids formation during ozonation of o-cresol (unbuffered)

of these acids was suggested by a rapid drop in pH during the reaction as were the cases with ozone and peroxide in unbuffered solutions. The anaerobic biodegradability of these identified acids (except the negligible salicylic acid) was examined by the BMP tests and results indicated that formic acid, acetic acid, and propionic acid were biodegradable to methane while oxalic acid and glyoxylic acid were not significantly degraded in the phenol-enriched methanogenic culture. During ozonation of 200 mg/l o-cresol, the nonbiodegradable glyoxylic acid and oxalic acids accounted for only 5.4% of the measured TOC while the biodegradable acetic acid, formic acid, and propionic acid constituted 15.2% of the measured TOC under an ozone dose of 55.9 moles/mole o-cresol. Therefore, the near complete conversion to methane from ozone treated 200 mg/l o-cresol at the highest doses as observed in Figure 1 may be largely due to formation of other unidentified, biodegradable products. However, the observation that no net methane was produced in any of the bottles containing treated, 2,4-DNP suggested that biodegradation of biodegradable organic acids was inhibited by oxidation reaction products in the treated samples.

ATA Tests

In order to evaluate the inhibitory nature of the treated samples, the ATA tests were conducted using phenol and acetate as the test substrates, respectively. Figures 3 and 4 show the cumulative methane production in sample as well as control bottles with 600 mg/L o-cresol as the initial compound and permanganate as the oxidant. Figure 3 was obtained with acetic acid as the test substrate while the data in Figure 4 were collected with phenol as the test substrate. The controls were



FIGURE 3. Methane production from added acetic acid in permanganate oxidation products



FIGURE 4. Methane production from added phenol in permanganate oxidation

spiked with a test substrate only and received no samples (either treated or untreated). The acetate control received only 500 mg/L acetate, whereas the phenol control received only 200 mg/L phenol.

The data in Figure 3 indicate that both the untreated o-cresol and all of the permanganate treated o-cresol were slightly inhibitory to the acetate-utilizing methanogens during the initial 44 hours of incubation. The untreated o-cresol and o-cresol treated with doses of 1 mole permanganate/mole or less also exhibited inhibition to the phenol degraders during the first 44 hours of incubation as the rates of methane production were lower than those of controls (Figure 4). However, methane production in sample bottles approximated or even exceeded those produced in the control bottles at the end of incubation, indicating acclimation or biodegradation of some products in the treated samples. A similar pattern was also noted for o-cresol with ozone or hydrogen peroxide as oxidant (data not shown). Ozonation products formed in the unbuffered o-cresol solutions were inhibitory even under doses as high as 24 moles/mole whereas products formed in the buffered basic solutions did not inhibit methanogenic activity under doses of 12 moles/mole or more. Lower doses of peroxide (0.3 mole/mole or less) also formed inhibitory products which exhibited inhibition during the early stages of incubation.

For 2,4-DNP, the initial compound was more inhibitory than the treated samples and progressive decrease in inhibition was observed with increasing doses of either ozone or hydrogen peroxide (Figures 5 and 6). The removal of 2,4-DNP during the course of oxidation may contribute to the observed reduction of toxicity. However, no significant reduction in toxicity was observed with permanganate oxidation in both buffered (at pH 9) and unbuffered solutions. The reduction of toxicity of 2,4-DNP by ozone or peroxide was quantified by defining relative toxicity, T_r , as

where

 $T_r = T/T_o$

- T = toxicity of samples to methane production = 1 - v/V
- $T_o =$ toxicity of initial compound to methane production = $1 - v_o/V$

Where

- v = total volume of methane produced from samples, mL
- v_o = total volume of methane produced from initial compound, mL, and
- V = total volume of methane produced from control, mL.





FIGURE5. Toxicity reduction of 2,4-DNP with ozonation

Figures 5 and 6 show the toxicity of reaction products relative to the initial compound, 2,4-DNP. The reduction in toxicity by ozonation did not differ significantly between the acetateutilizing methanogens and the phenol-degrading bacteria nor was it affected appreciably by the pH during the ozonation reaction. However, toxicity reduction by hydrogen peroxide was faster for acetate-utilizing methanogens than phenol-degraders under the same doses, indicating that the phenol-degrading bacteria were more susceptible to inhibition of the reaction products than were the acetate-utilizing methanogens (Figure 6). The doses for 50% toxicity reduction were about 2.5 moles and 4.5 moles per mole of 2,4-DNP for the methanogens and the acetogens, respectively. A dose of about 7 moles of ozone per mole of 2,4-DNP was needed to reduce toxicity to 50%. Doses beyond 5 moles H2O2 or 10 moles O3 did not seem to result, appreciably, in further toxicity reduction. However, no significant toxicity reduction was noted with permanganate even under a high dose of 10 moles/mole 2,4-DNP.

Based on the oxidant doses, ozone appeared to be less efficient than peroxide or permanaganate in terms of enhancing anaerobic biodegradability and performed less effectively than peroxide in reducing methanogenic toxicity. Although the gas transfer efficiency in the ozonation reactor was not determined, a low efficiency was expected, especially under the high dose conditions employed in this study. Others [24] have showed a stoichiometric yield of 2.03 moles of ozone consumed per mole of o-cresol consumed which is much lower than our values (4.1-6.1 moles O₂/mole o-cresol). o-Cresol was much less toxic toward the methanogenic cultures than 2,4-DNP. At 300 mg/ L, o-cresol was not toxic but only caused slight inhibition during the early stage of incubation. 2,4-DNP, on the other hand, ceased methanogenic activity completely even at 50 mg/ L (data not shown). The observation that the phenol-degrading acetogens were more susceptible to toxicity caused by peroxide



FIGURE 6. Toxicity reduction of 2,4-DNP with hydrogen peroxide

oxidation products than the acetate-utilizing methanogens (as indicated in Figure 6) is in agreement with our recent finding using substituted phenols as the test inhibitors [25]. Furthermore, ATA tests conducted on the identified, nonbiodegradable oxalic acid and glyoxylic acid revealed that these acids did not inhibit phenol degradation nor affect acetate utilization at the detected levels. Thus, the inhibition of the methanogenic cultures by oxidant treated samples may be attributed to unidentified oxidation products.

CONCLUSIONS

Chemical oxidation by ozone, Fenton's reagent, or permanganate enhanced the anaerobic biodegradability of o-cresol by forming biodegradable products such as formic acid, acetic acid, and propionic acid. However, complete or near complete oxidation of o-cresol was needed to form a significant portion of biodegradable products. The biodegradability of chemical treated o-cresol increased with increasing doses of oxidants. Chemical oxidation by ozone or peroxide also reduced the toxicity of 2,4-DNP to a phenol-enriched methanogenic culture. Based on oxidant doses, approximately 4 moles hydrogen peroxide, in the presence of a ferrous iron catalyst, 7 moles permanganate, or 25 moles ozone per mole of o-cresol is needed to significantly enhance biodegradability of o-cresol. Relatively lower doses of peroxide or ozone are required to reduce methanogenic toxicity of 2,4-DNP. The observed high doses of oxidants needed to enhance anaerobic biodegradability and reduce methanogenic toxicity may limit chemical pretreatment to small streams of industrial wastes unless oxidation efficiency can be improved through process modification or application of appropriate catalysts. However, chemical oxidation pretreatment followed by anaerobic biological processes may be applied to treat polluted groundwater in which the level of refractory and/or inhibitory contaminants are typically not high.

ACKNOWLEDGMENTS

The author wishes to thank James L. Latchaw and Pin-Chieh Pai for conducting the laboratory work.

LITERATURE CITED

- Patterson, J. W., "Wastewater Treatment Technology," Ann Arbor Science, Ann Arbor, MI (1975).
- Baillod, C. R., Faith, B. M., and Masi, O., "Fate of Specific Pollutants During Wet Oxidation and Ozonation," *Environ. Prog.*, 1(3), pp. 217-227 (1982).
- Eisenhauer, H. R., "The Ozonation of Phenolic Wastes," J. Water Pollu. Control Fed., 40(11), pp. 1887-1899 (1968).
- Gould, J. P., and Weber, W. J., "Oxidation of Phenols by Ozone," J. Water Pollu. Control Fed., 48(1), pp. 47– 60 (1976).
- Neufeld, R. D., and Spinola, A. A., "Ozonation of Coal Gasification Plant Wastewater," *Environ. Sci. Technol.*, 12(4), pp. 470-472 (1978).
- Singer, P. C., and Gurol, M. D., "Dynamics of the Ozonation of Phenol—I. Experimental Observations," Water Res., 17(9), pp. 1163-1171 (1983).
- 7. Yamamoto, Y., Niki, E., Shiokawa, H., and Kamiya, Y., "Ozonation of Organic Compounds. II. Ozonation of

Phenol in Water," J. Org. Chem., 44(13), pp. 2137-2142 (1979).

- Eisenhauer, H. R., "Oxidation of Phenolic Wastes: I. Oxidation with Hydrogen Peroxide and a Ferrous Salt Reagent," J. Water Pollu. Contr. Fed., 36(9), pp. 1116-1128 (1964).
- Keating, E. J., Brown, R. A., and Greenberg, E. S., "Phenolic Problems Solved with Hydrogen Peroxide," Proc. 33rd Purdue Ind. Waste Conf., pp. 464–470 (1979).
- Lee, D. G., and Sabastian, C. F., "The Oxidation of Phenol and Chlorophenols by Alkaline Permanganate," *Can. J. Chem.*, 59, pp. 2776-2779 (1981).
- Ladbury, J. W., and Cullis, C. F., "Kinetics and Mechanism of Oxidation by Permanganate," Chem. Rev., 58, pp. 403-438 (1958).
- Bailey, P. S., "Organic Groupings Reactive Toward Ozone Mechanisms in Aqueous Media," in Ozone in Water and Wastewater Treatment, Evans, F. L., Ed. Ann Arbor Science, Ann Arbor, Mich., pp. 29-59 (1977).
- Wang, Y. T., Suidan, M. T., Pfeffer, J. T., and Najm, I., "Effects of Some Alkyl Phenols on the Methanogenic Degradation of Phenol," *Appl. Environ. Microbiol.*, 54(5), pp. 1277-1279 (1988).
- Boyd, S. A., Shelton, D. R., Berry, D., and Teidje, J. M., "Anaerobic Biodegradation of Phenolic Compounds in Digested Sludge," *Appl. Environ. Microbiol.*, 46(1), pp. 50-54 (1983).
- Bowers, A. R., Gaddipati, P., Eckenfelder, W. W., and Monsen, R. M., "Treatment of Toxic or Refractory Wastewaters with Hydrogen Peroxide," *Wat. Sci. Tech.*, 21, pp. 477-486 (1989).
- Owen, W. F., Struckey, D. C., Healy, J. B., Young, L. Y., and McCarty, P. L., "Bioassay for Monitoring Biochemical Methane Potential and Anaerobic Toxicity," *Wat. Res.*, 13, pp. 485-492 (1979).
- Wang, Y. T., and Latchaw, J. L., "Anaerobic Biodegradability and Toxicity of Hydrogen Peroxide Oxidation Products of Phenols," *Res. J. Wat. Pollu. Cont. Fed.*, 62(3), pp. 234-238 (1990).
- Smith, P. H., and Mah, R. A., "Kinetics of Acetate Metabolism During Sludge Digestion," *Appl. Microbiol.*, 14(3), pp. 368-371 (1966).
- Dwyer, D. F., Krumme, M., Boyd, S. A., and Tiedje, J. M., "Kinetics of Phenol Biodegradation by an Immobilized Methanogenic Consortium," *Appl. Environ. Microbiol.*, 52(2), pp. 345–351 (1986).
- Sheridan, W. G., Jones, W. J., Wolfe, R. S., and Suidan, M. T., "Fundamentals Associated with Biodegradation of Phenols and Polycyclic N-Aromatic Compounds," Report to the Advanced Environmental Control Technology Research Center, University of Illinois, Champaign, IL. U.S. EPA Cooperative Agreement CR 806819 (1985).
- Am. Public Health Assoc., "Standard Methods for the Examination of Water and Wastewater," 16th ed., Washington, D.C. (1985).
- Salanitro, J. P., and Muirhead, P. A., "Quantitative Method for the Gas Chromatographic Analysis of Short-Chain Monocarboxylic and Dicarboxylic Acids in Fermentation Media," *Appl. Microbiol.*, 29(3), pp. 374-381 (1975).
- McCarty, P. L., "Anaerobic Waste Treatment Fundamentals—Part I," Public Works, pp. 107-112, Sept. (1964).
- Beltran, F. J., Encinar, J. M., and Garcia-Araya, J. F., "Ozonation of o-Cresol in Aqueous Solutions," Wat. Res., 24(11), pp. 1309-1316 (1990).
- Wang, Y. T., Pai, P. C., and Gabbard, H. D., "Effects of Aromatic Concentration on Methane Fermentation. Final Report to Water Resources Research Institute, University of Kentucky, Report No. 174 (1990).

Soil Decontamination via Microwave and Radio Frequency Co-Volatilization

Clifford E. George, George R. Lightsey, Inchul Jun, and Jiayi Fan

Department of Chemical Engineering, Mississippi State University, P.O. Drawer CN, MS State, MS 39762

Microwave and radio frequency heating techniques have proven to be suitable on the laboratory scale and for small scale field studies as energy sources for thermal processing of solvent contaminated hazardous waste and contaminated soils. The process described here is a technique that could be used to remove contaminates from soils or sludges on-site, collect the contaminate in an activated carbon absorption tower, and move the activated carbon off site for regeneration. The data presented show that destruction and removal efficiencies (DRE) of near 100% can be achieved for phenanthrene in simulated API separator sludge and 60% for pentachlorophenol in contaminated soil. A technique to enhance microwave absorption by the addition of carbon particles to the soil or sludge sample to be treated is discussed.

INTRODUCTION

Thermal treatment of solids contaminated with volatile hazardous constituents is a popular treatment method. Thermal treatment includes the practices of incineration, pyrolyzation coupled with incineration, or volatilization without combustion and recovery of the volatile compounds. The discussion here is limited to the noncombustion thermal treatment methods.

A commercially viable process used to remove volatile compounds, such as a mixture of petroleum derived hydrocarbons, from a contaminated solid substrate is to treat the mixture in a heated oven or rotary dryer for a period of time. This method is economical for some applications. However, regulations may prohibit direct release of contaminate laden air or exhaust gases into the atmosphere. In this case, the ability to clean the exhaust gases in order to remove contaminates becomes the limiting factor for the process.

A secondary but non-trivial problem is the heating step required to raise the temperature to a sufficient level to volatilize the contaminate. Once heated, the contaminate must diffuse through the solid particle to the surface to be removed by a sweep gas. Often, as in the case of working with heat sensitive substrates such as polymeric materials [1], it is difficult to maintain these elevated temperatures to remove the last trace of contaminate without burning or otherwise thermally damaging the substrate. Problems occur when the outside of the substrate particle must be heated to an elevated temperature in order to effect conduction heat transfer to the center of the particle. Since microwave and radio frequency mechanisms heat from the inside out, polymeric materials can be heated without experiencing excess skin temperatures at the outside surface temperature of the particle. These techniques have been the subject of several studies [2].

The volatilization process reduces to the following operations:

- heating the material to a temperature sufficient to remove the volatile contaminate without combustion or thermally cracking the contaminate;
- 2. removing the contaminate from the substrate;
- 3. transporting the contaminate away from the bulk of the substrate;
- 4. recover the substrate from an exhaust gas;
- 5. and dispose of the contaminate in an environmentally acceptable manner.

When dealing with materials that are considered to be haz-



FIGURE 1. Proposed on site treatment scheme

ardous wastes, it is desirable to remove contaminates without combustion. Otherwise, the process will be classified as a hazardous waste incinerator. In the United States, incinerators have become unpopular with many environmental and political groups because of the potential for release of contaminates or by-products of combustion via the flue gas stream. Some of the by-products, such as dioxin, are perceived to be more hazardous then the original contaminates. As a result, obtaining the necessary permits to operate an incinerator is an extremely expensive and time consuming process. On the other hand, a process that does not include combustion with the associated flue gas stream will be much less controversial and should have fewer problems during the permitting process.

Dauerman and Windgasse first suggested that an inert gas could be used to protect the contaminate from exposure to oxygen while the mixture was being heated with microwave energy in a fluidized bed [3]. By eliminating oxygen, subsequent oxidation cannot take place at the elevated temperatures necessary for desorption of volatile contaminates from soils. The inert gas then acts as a carrier gas to remove volatilized compounds for recovery. The inert gas used was Argon.

Chemical Waste Management has developed a commercial process that uses conventional heating techniques for thermal desorption and recovery of volatile organic compounds. In order to avoid combustion that occurs with high temperature processing, the patented XTRACT process uses nitrogen as an inert gas to blanket the heating chamber and serve as a carrier to move the volatile compounds from the heating chamber to a cooling and recovery section [4]. The nitrogen is reheated and recycled back to the heating section and only a small amount of makeup nitrogen is needed. Nitrogen serves the purpose well as it is available commercially at a relatively low cost.

George and others [5, 6] proposed that if microwave heating is a viable technology, then the process could be carried out in a reduced pressure condition to exclude oxygen in order to prevent combustion of the contaminate. Figure 1 shows the process flow diagram proposed for the application of this microwave treatment technique. The addition of carbon particles is shown and is a technique that was used to enhance the heating of the contaminated substrate. The mechanism of this technique will be discussed later. The objective of the treatment scheme is to capture all volatile contaminates in a carbon absorption tower for off site disposal.

This technique is expanded to semi-volatile compounds in this paper. The semi-volatile compounds of interest are those commonly found in API separator sludges and contaminated soils from wood treatment sites. API separators are used in the oil refining industry to separate solids, oil, and water from the various waste water streams generated in the refinery. These streams can be generated by a refining process, a wash down operation, tank bottom draining, or storm water runoff collection. The solid residue, API separator sludge (APISS), is a listed hazardous waste and must be disposed of as a hazardous waste unless the refinery can recycle the material back to a refining process such as a coking furnace. The sludge contains a mixture of volatile and semivolatile compounds [7]. For the purpose of this study, phenanthrene (normal boiling point 340°C), which is found in APISS, was chosen as representative semi-volatile compound.

Additionally, tests were performed using a simulated pentachlorophenol (PCP) contaminated soil. PCP contamination is typically found in soils at wood treating sites [8].

MATERIALS AND METHODS

Figure 2 shows the apparatus used during the study. A 30 ml quartz crucible used as a sample container. The container was placed in a microwave wave guide connected to a Gerling Laboratory 1.5 kW 2450 MHz power supply. Connections were provided for nitrogen sweep gas inlet and outlet. The outlet was connected to route the entire sweep gas stream through a series of cold traps.

Contaminated substrates were placed in the crucible and heated by microwave power for a preset time interval. During the tests, a vacuum of 500 torr was maintained and sweep gas was introduced to the container at a rate of 3.3 standard liters per minute. A microwave power of 540 watts was used throughout the experiment. After heating, the substrate was removed and analyzed for contaminate.

The efficiency of microwave heating to the decontamination of the substrate was evaluated by determination of the destruction and removal efficiency (DRE) of the considered contaminate. To make this determination, the contaminate concentration in the substrate was determined before and after treatment and DRE calculated by the following equation:

$$DRE = \frac{c_i - c_i}{c_i} \times 100\% \tag{1}$$

where, c_i is the initial concentration of contaminate in the sample and,

 c_i is the final concentration of contaminate in the microwave treated sample.

Analysis was made in accordance with EPA standard method 8270. Direct solvent extraction was used to prepare the GC sample from the microwave treated substrate and the internal standard method was used for GC quantitative analysis. A Hewlett Packard Model 5890 Series II GC equipped with a DB-5 capillary column was used in the research.

The course of the experimental work led to the conclusion that the soil/sludge materials did not possess the dielectric characteristics necessary to absorb sufficient microwave power



FIGURE 2. Sketch of microwave processing equipment



FIGURE 3. Temperature profiles of API separator sludge blank samples with carbon particles added

at a rate sufficient to maintain the temperatures required or contaminate removal. To increase the power absorption properties of the substrate, a series of experiments were performed using solid carbon particles of an average diameter of 3 mm as an admixture to the materials being treated. This technique takes advantage of the ability of carbon to absorb microwave energy by ionic conduction [9]. By using this technique, the heating of the substrate was enhanced by the carbon particle being heated by microwave energy and the substrate being indirectly heated by conduction of heat from the carbon to the substrate particle. Thus, the substrate is heated internally by microwave energy and externally by conduction from a heated carbon particle.

RESULTS AND DISCUSSION

Figure 3 illustrates the temperature rise of soil mixed with the various amounts of carbon particles added. The steady state substrate temperature without carbon particles was less than 100°C whereas the addition of a 40% admixture of carbon particles brought the test sample to a temperature in excess of 1000° C. Further power addition at temperatures above 1000° C caused physical changes in the carbon particles which decreased



FIGURE 4. DRE of PCP in 300 PPM contaminated soil by microwave heating

218 August, 1992

its ability to absorb energy resulting in a steady state substrate temperature below 700°C. The presence of carbon particles significantly enhances the ability of the mixture to attain temperatures needed for decontamination of the soil.

A series of tests were performed to depict the effectiveness of this treatment technique to PCP removal from soils at wood treatment sites. Soil samples were spiked with 300 ppm PCP to simulate this type of contaminated soil. Figure 4 shows the PCP destruction or removal efficiency (DRE) at various operating times. About 60% of the PCP can be removed from the soil without combustion in the case of 40% added carbon particles. The DRE is rather low primarily because of the high normal boiling point of PCP (310°C).

Addition of a certain amount of water to the contaminated soil can efficiently enhance the ability of the soil to absorb microwave energy and enhance the diffusion of the volatile contaminate [6]. But this enhancement method failed for the semi-volatile contaminated soil sample shown in Figure 4 due to the fact that PCP has a much higher boiling point than water. The effect of the presence of additional volatile components is seen in the results obtained with the API separator sludge.

API separator sludge was simulated by the addition of 110 ppm of phenanthrene to a mixture of 90% 40 mesh sand and 10% heptadecane. Phenanthrene is a specific semi-volatile component of APISS whereas heptadecane is added as a control compound to simulate the presence of light oils common to APISS. The mixture was then heated in the microwave apparatus and DRE was determined. Figure 5 shows that with the addition of carbon particle, a DRE of over 99% was achieved. These results show the advantages of contaminate removal with carbon particles addition when compared to the 20% contaminate removal without carbon particle addition. Phenanthrene removal, 99%, appears to be much easier than the PCP removal, 60%, at the same conditions. Differences in normal boiling point, 340°C for phenanthrene compared to 310°C for PCP does not explain the difference in removal. The presence of the heptadecane in the phenanthrene sample probably enhanced the diffusion of the phenanthrene from the sand particles due to vapor pressure effects inside the porous substrate. Further work is planned using the addition of heptadecane to see its effect on DRE of other semi-volatile compounds in soil.

CONCLUSIONS



FIGURE 5. DRE of phenanthrene in API separator sludge (110 PPM) by microwave heating

The data collected show that microwave heating can be used

to heat soil for decontamination without the use of incineration and with limited emissions of contaminates. The fact that no combustion takes place means that the chances of forming products of incomplete combustion are remote. Microwave heating has an advantage over conventional heating techniques in that it can deliver the heat energy to the inside of the heated substrate without developing excessive temperatures at the outside surface of the substrate. This is important if the substrate or the contaminate is heat sensitive.

It was found that carbon particles can effectively enhance the absorption of the heat energy. For this technique to be practical on a commercial scale, a scheme would need to be devised for the process depicted in Figure 1 whereby carbon pellets could be mixed in the soil for the heating process, recovered after the soil had been treated, and reused for the next batch of soil.

The results presented here are preliminary in nature and further study is needed. However, these preliminary results indicate that the presence of longer chain hydrocarbons, such as heptadecane, similar to those found in API separator sludge tends to enhance the removal of semi-volatiles from the sludge. This may be important if microwave heating were adopted as a commercial treatment technique.

ACKNOWLEDGMENTS

This project has been funded in part with Federal Funds as part of the program of the Gulf Coast Hazardous Substance Research Center which is supported under cooperative agreement R 815197 with the United States Environmental Protection Agency (EPA) and in part with funds from Mississippi Power and Light Company (MP&L). The contents do not necessarily reflect the views and policies of the U.S. EPA or MP&L nor does the mention of trade names or commercial products constitute endorsement or recommendation for use.

LITERATURE CITED

- Gibson, C., I. Matthews, and A. Samuel, "Microwave Enhanced Diffusion in Polymeric Materials," *Journal of Microwave Power and Electromagnetic Energy*, 23(1), pp. 17-28 (1988).
- Wehr, A. G., C. E. George, and G. R. Lightsey, *Journal* of Microwave Power and Electromagnetic Energy, 25(2), pp. 100-103 (1990).
- Dauerman, L. and G. Windgasse, "Microwave Treatment of Hazardous Waste," Proceedings of the International Conference on High Frequency/Microwave Processing and Heating, p. 6-6-1, Arnhem, The Netherlands (1989).
- EPA, The Superfund Innovative Technology Evaluation Program: Technology Profiles, EPA/540/5-90/006, pp. 34– 35 (1990).
- George, C. E., Journal of Hazardous Materials, 24(2+3), p. 280 (1990).
- George, C. E., Inchul Jun, and Jiayi Fan, "Application of Microwave Heating Techniques to the Detoxification of Contaminated Soil," Ceramic Transactions, Vol. 21, *Microwaves: Theory and Applications in Materials Processing*, pp. 459-466 (1991).
- Leemann, J. E., Journal of the Air Pollution Control Association, 38(6), pp. 814–823 (1988).
- Borazjani, H., and others, "Evaluation of Wood-Treating Plant Sites for Land Treatment of Creosote and Pentachlorophenol Contaminated Soils," ACS Symposium Series 422, pp. 252-266 (1989).
- 9. Microwave Power in Industry, EPRI EM-3645, Electric Power Research Institute, Palo Alto, CA (1984).

Supercritical Extraction of Polynuclear Aromatic Hydrocarbons from Soil

S. Kothandaraman, R. C. Ahlert, E. S. Venkataramani and A. T. Andrews

Department of Chemical and Biochemical Engineering, Rutgers University, Piscataway, NJ 08855-0909

Adsorption and desorption of polynuclear aromatic hydrocarbons from soil in the presence of Supercritical CO_2 is studied. The soil adsorption isotherms are described by the Freundlich and BET models. Polar modifiers like methanol and water are found to exhibit a strong desorptive influence. A simple, singleparameter model using linear desorption kinetics is used to fit the desorption data.

INTRODUCTION

Supercritical fluid extraction is one of the most promising emerging technologies in the remediation of contaminated soils. Supercritical fluids can achieve high efficiencies in the extraction of organic contaminants from soils. Supercritical CO₂ has been demonstrated to effectively remove many species including polycyclic aromatic hydrocarbons (PAHs) [1, 2], pesticides, phenol, and polychlorinated biphenyls (PCBs) [3]. A pilotscale extraction system has been shown to remove PCBs from contaminated sediment using liquefied propane [4]. The primary disadvantage of supercritical fluid processing is the higher equipment costs of pressure operations. However, the advantages like low operating temperature, enhanced mass transfer, and little or no solvent residue on the product can justify the capital costs, especially in smaller scale/high product value operations. Andrews [1] conducted adsorption and solubility studies focusing on four PAH compounds: anthracene, phenanthrene, triphenylene, and perylene and some desorption experiments with anthracene and phenanthrene from a sandy loam-type soil using supercritical CO₂. In this paper a simple, single-parameter model assuming linear desorption kinetics was attempted on the desorption data and found to fit with a good correlation factor.

EXPERIMENTAL METHODS

The sandy loam-type soil was packed in a pressure vessel and the system was pressurized with CO_2 using a piston pump. Recirculation of SC CO_2 fluid phase was accomplished by a magnetically coupled gear pump. On-line measurements of temperature and pressure was made using thermocouples and pressure transducers, respectively. The entire system was placed in a temperature controlled oven. On-line samples of the fluid phase were taken in a closed, 2 mL chamber using a six-port sample valve. Methanol flushing of the sample loop and transfer lines assured quantitative and representative sample collection. The methanol solution containing the PAH was then analyzed by high performance liquid chromatography. Solid PAH added to the system dissolved in the circulating SC CO2. In adsorption studies, the PAH solid was introduced below the soil bed, downstream from the recirculating flow, thus avoiding contact of undissolved solute with the soil. Adsorption took place from the solution phase only. The system volume was carefully calibrated to allow calculation of the soil PAH concentrations from the fluid phase concentrations and the mass of soil in the system. In flow desorption experiments, the soil was pre-spiked with PAH prior to desorption. The system was allowed to equilibrate with circulation for several hours. Circulation was then stopped and CO₂ was introduced at a constant flow rate. Exit flow was controlled by a heated discharge metering valve, totalized by a dry test meter, and flow rates measured by a rotameter. Samples of the postextractor fluid were taken prior to depressurization, at frequent intervals, and then analyzed by HPLC. Complete details of the experimental procedure are given in the earlier paper [2].

PAH ADSORPTION [2]

The solubility of the PAHs in SC CO₂ was studied by using the Peng-Robinson equation of state to calculate the fluid density, and the solubility (S) behavior was shown to follow the typical van't Hoff relation, where ln S is linear with 1/Tfor constant density plots. The adsorption data were obtained at three fluid densities, and two temperatures, 45° C and 65° C. Anthracene isotherms at 45° C are shown in Figure 1. It can be seen that with increasing densities, the adsorption levels are reduced. The nonlinear, sigmoid shaped curves are nicely represented by regression to the Brunauer-Emmett-Teller (BET) isotherm equation. The Freundlich model is not able to predict the sigmoidal upswing at concentrations near the saturation limit. SC CO₂ is not a good solvent because it provides little


FIGURE 1. Anthracene adsorption at 45°C

polar or dispersive interaction with PAHs. However, the addition of polar modifiers can have two beneficial effects. Significant increases in PAH solubility can be achieved through the addition of a modifier such as methanol; in addition, polar modifiers can become strong competitors for polar adsorption sites in the non-aqueous SC system. To ascertain the importance of polar sites to PAH adsorption, low levels of water and methanol were added to the adsorption system during anthracene adsorption experiments. The soil was pre-equilibrated with a known concentration of anthracene, by recycling the system for several hours. After an initial sampling, water (or methanol) was added to the system in small increments and the system sampled again after re-equilibration. Figure 2 describes the effect of water or methanol on K_{ads} , the ratio of soil concentration to fluid concentration. A strong desorptive influence of both water and methanol is apparent, with the bulk of the effect occurring in the first addition of 0.2%, and a rapid decline in the desorptive effect at higher concentrations. Above 0.6%, further additions of water or methanol have only a minor effect on the adsorption of anthracene.

PAH DESORPTION

A "plug flow" desorption of the PAHs was observed. The desorption curve shapes for phenanthrene and anthracene were observed to be quite similar, with the less strongly bound phenanthrene eluting faster relative to anthracene. Tan and Liou proposed a linear desorption kinetics model [5] and found that it represented their experimental data quite well. For this work, the same model with minor modifications was utilized. The single-parameter model for describing desorption, can be



FIGURE 2. Effect of polar modifiers on anthracene adsorption SC density 0.7 g/cc



FIGURE 3. Desorption of anthracene from soil

written for the bulk fluid phase.

$$\epsilon \frac{\partial C}{\partial t} + u \frac{\partial C}{\partial z} = -\rho_s (1 - \epsilon) \frac{\partial q}{\partial t}$$
(1)

$$t = 0, \quad C = 0$$
 (2)

$$z=0, \quad C=0 \tag{3}$$

and for the solid phase:

$$\frac{\partial q}{\partial t} = -kq \tag{4}$$

$$t=0, \quad q=q_0 \tag{5}$$

The concentration at the exit of the bed can be expressed by:

$$C_{\epsilon} = \rho_{s} \frac{(1-\epsilon)}{\epsilon} q_{0} \left\{ \exp\left[-k\left(t-\frac{\epsilon L}{u}\right)\right] - \exp(-kt) \right\}$$
(6)

The desorption rate constant k, was estimated by fitting Equation (6) to the experimental data. Figures 3 and 4 illustrate the observed and predicted desorption behavior of anthracene and phenanthrene, respectively. The temperature and pressure were held constant at 45°C and 2500 psig, respectively; density of SC CO₂ was 0.79 g/mL under these conditions. It can be seen that the model adequately explains the experimental observations. The desorption behavior is well correlated with the



FIGURE 4. Desorption of phenanthrene from soil

Environmental Progress (Vol. 11, No. 3)

model, especially at high and low PAH concentrations, with minor deviations at intermediate levels. Mass transfer resistances could possibly be responsible for the observed deviations.

CONCLUSIONS

Sandy loam adsorption isotherms for anthracene are nonlinear and can be described using the BET equation. Water and methanol, added as polar modifiers in adsorption experiments, dramatically reduce the PAH adsorbed on the soil surface. A single-parameter model was used and found to explain the desorption behavior of anthracene and phenanthrene from a sandy loam-type soil, quite well. Mass transfer considerations are the focus of a current experiments and a modeling effort.

NOTATION

- C = concentration of the PAH in the supercritical fluid phase,mg/L
- C_e = exit concentration of PAH, mg/L
- k = desorption rate constant, 1/min
- L = bed length, cm
- q = loaded PAH on soil, mg/Kg soil
- q_0 = initially loaded PAH on soil, mg/Kg soil

- t = time, min
- u = superficial velocity, cm/min
- z = axial position in bed, cm $<math>\epsilon = void fraction in bed$
- $\rho_s = \text{particle density of soil, g/cm}^3$

LITERATURE CITED

- Andrews, A. T., Ph.D. Thesis, Rutgers, The State University of New Jersey, 1990.
- Andrews, A. T., R. C. Ahlert, and D. S. Kosson, "Supercritical Fluid Extraction of Aromatic Contaminants from a Sandy Loam Soil," *Environmental Progress*, 9(4): p. 204– 210, 1990.
- Dooley, K. M., D. Ghonasgi, and F. C. Knopf, "Supercritical CO₂—Cosolvent Extraction of Contaminated Soils and Sediments," *Environmental Progress*, 9(4): p. 197-203, 1990.
- Staley, L. J., R. Valentinetti, and J. McPherson, "SITE Demonstration of the CF Systems Organic Extraction Process," *Journal of Air and Waste Management Association*, 40(6): p. 926-931, 1990.
- Tan, C.-S. and D.-C. Liou, "Modeling of Desorption at Supercritical Conditions," *AIChE Journal*, 35(6): p. 1029– 1031, 1989.

The B.E.S.T.[®] Solvent Extraction Process Treatment of Soil, Sediment, and Sludges

George R. Jones

Resources Conservation Co., Bellevue, WA 98004

Many treatment technologies are being used or proposed to remediate hazardous wastes in sites throughout the U.S. These technologies can be broken down into four general treatment categories: destructive, fixation, biological, and separation. This paper will discuss a separation technology: the B.E.S.T. (Basic Extractive Sludge Treatment) process.

INTRODUCTION

Following is a brief description of the four general treatment categories:

- Destructive technologies include incineration, in-situ vitrification, and selective chemical destruction (such as dechlorination of PCBs). In these types of processes the waste contaminant and other components of the waste are irreversibly changed.
- Biological processes are "destructive" to contaminants but since natural processes are involved, biological processes are a preferred approach for many cleanup actions if practical and economical.
- Fixation technologies presume that hazardous compounds in waste materials can be "locked" by mixing additives such as Portland cement that completely bind or encapsulate the waste.
- Separation technologies include solvent extraction, soil washing, physical separation processes (centrifugation, filtration, sedimentation and the like), and low temperature thermal desorption (drying). Thermal desorption is generally regarded as a separation process although at medium (>500°F) and high (>1000°F) operating temperatures significant irreversible changes can occur in product solid characteristics. Cracking or decomposition of organic compounds can occur.

SOLVENT EXTRACTION AS A SEPARATION PROCESS

Figure 1 is a simplified illustration of waste separation using solvent extraction. Three distinct products are produced: solids free of water and hydrocarbons, oil free of solids and water, and water without residual oil and grease or suspended solids. In the real world, none of these products will be completely free of residuals. Many design decisions are made based on the limits of the residuals allowed in each product stream. In addition, waste feeds range from watery sludges to soils highly contaminated with hydrocarbons. This further complicates design engineering decisions.

A solvent extraction process must meet the following criteria in order to process a wide variety of contaminated wastes:

- Meet treatment goals established for all decontaminated product streams
- · Process a wide variety of feed compositions
- Effectively remove a broad range of organic contamination
- Meet permitting guidelines for performance and air emissions

The B.E.S.T. solvent extraction technology developed by Resources Conservation Company has been demonstrated an effective separation process for a wide variety of organically contaminated wastes in bench, pilot, and commercial scale



FIGURE 1. The B.E.S.T.® process.

operation. As a separation process it has been shown to effectively remove polychlorinated biphenyls (PCBs), polyaromatic hydrocarbons (PAHs), pesticides, and many other organics from soil, sediment, and sludge. Current design work has focused on process design to handle a wide variety of wet and dry feeds.

THE B.E.S.T. SOLVENT EXTRACTION PROCESS

The B.E.S.T. process is a patented solvent extraction technology using triethylamine (TEA) as the solvent. TEA is a aliphatic amine that is produced by reacting ethyl alcohol and ammonia. TEA is an excellent solvent for extraction of organic contaminants from various wastes.

- TEA is easily recovered within the process due to a high vapor pressure and formation of a low-boiling azeotrope with water.
- TEA has one-seventh the latent heat of water, which allows solvent recovery at very low energy levels.
- TEA biodegrades in the environment in the presence of Aerobactor soil bacteria. (EPA document EPA-600/2-82-001a)
- TEA is inversely miscible with water; that is, soluble in water below 60°F but insoluble in water above 90°F.

The inverse miscibility property makes the B.E.S.T. process unique. A block diagram of the process is presented in Figure 2. The first extraction of the waste is conducted at 40° F. At this temperature, triethylamine is soluble with both the organics and water. The advantage of a miscible solvent is the ability to break difficult oil-water emulsions, allowing the organics to be extracted from the waste without being "protected" by the water film barrier in emulsions. The effluent from the first extraction contains most of the water in the feed material. If the first stage effluent contains sufficient water to



FIGURE 2. B.E.S.T. process concept.

allow a phase separation of the solvent and water, the effluent is heated to a temperature well above the miscibility limit $(>100^{\circ}F)$. At this temperature, the effluent separates into two distinct phases, a solvent/organic phase and a water phase. The two phases are separated by gravity and decanted.

Once the water is removed, subsequent extraction stages can be conducted at elevated temperature to improve extraction efficiencies. Effluent from all stages is recovered by solvent evaporation/oil stripping and the water phase is stripped of residual solvent in a live steam stripper. Operation on waste with significant light end organics (boiling point < 200°F) such as refinery waste may require separations equipment to limit light end accumulation in the recycled solvent stream. Triethylamine is removed from the treated solids by indirect steam heating. A small amount of steam may be added directly to the dryer to improve residual stripping of solids and return a small quantity of water to the solids for dust control.

The B.E.S.T. process operates near ambient pressure and temperature and at a mildly alkaline pH. Temperatures of the liquid streams vary from about 40–170°F. The system runs at virtually atmospheric pressure. A low pressure nitrogen blanket creates a small positive pressure on tanks and vessels. Since the process operates in a closed loop, with one small vent for removal of non-condensible gases, there are virtually no air emissions. An advantage of this type of system is that it uses standard off-the-shelf processing equipment.

SEPARATION RESULTS

The B.E.S.T. solvent extraction process has been proven effective on contaminated soils, river and harbor sediment and

	Table 1 PCB-contaminated soils and sediments									
Test	Feed PCBs (mg.l)	% Oil	% Water	% Solids	PCBs in Solids (mg/l)	% Removal				
Α	250	.06	9.4	91	2.2	99.2				
В	120	.06	13	87	6.4	95.4				
С	4300	1.0	15	84	6.4	>99.9				
D	5300	1.0	19	80	11	99.8				
E	190	.07	16	84	1.6	99.3				
F	3000	1.5	30	69	0.8	>99.9				
G	15000	1.8	29	69	6.9	>99.9				
н	760	2.6	22	75	4.1	99.6				
I	1900	.22	14	86	5.5	99.8				
J	19	.09	16	84	0.7	96.9				
K	1500	1.5	11	88	19	98.6				
L	820	0.2	8	92	9.4	98.9				

			BDAT
Constituent	Feed	Product Solids	Standards
Oil & Grease, wt. %	13	0.025	_
Solids, wt. %	33	99.5	-
Benzene	100	0.0	14
Ethyl Benzene	160	0.0	14
Toluene	1300	0.044	14
Total Xylenes	1600	0.044	22
Naphthalene	340	9	42
Anthracene	47	3.6	28
Phenanthrene	190	11	34
Benzo(a)anthracene	29	2.5	_
Chrysene	22	1.6	15
Pyrene	61	3.3	36
Benzo(a)pyrene	20	2	12
Bis(2-Ethylhexyl) Phthalate	< 190	na	7.3
Di-n-Butyl Phthalate	< 190	<2.6	3.6
o-Cresol	<190	<2.6	_
m- & p-Cresol	<190	<2.6	-
Phenol	<190	<2.6	3.6

various contaminated sludges. Table 1 summarizes results from many investigations of actual samples of PCB-contaminated soils from various sites. The extraction efficiencies reflected in this table are affected by the soil matrix (clay, sand, silt, etc.) and the particular process parameters (solvent to feed ratio and number of extraction stages) selected for each specific test. Achieving extraction efficiencies approaching 99% is typical.

Table 2 shows representative results from pilot tests on a difficult-to-treat refinery slop oil emulsion filter cake from the Houston area. Disposal of the solids from this material must meet strict EPA limits (BDAT standards for K048 through K052 wastes) on polyaromatic hydrocarbons (PAHs) prior to disposal in a regulated landfill. Water separated during treatment of refinery wastes is directed to the plant's waste water treatment facility while the oil separated from the waste stream is recycled back to the refinery.

Table 3 summarizes the treatment results of using the B.E.S.T. process on a soil contaminated with a wide variety of pesticide residues from a California Superfund site.

DESIGN FOR MAXIMUM FEED FLEXIBILITY

RCC has implemented several equipment design concepts in pilot and commercial scale equipment during B.E.S.T. process development. Initial pilot and commercial designs done in the mid-80s were multi-stage, countercurrent extraction trains. Each stage employed a mix tank for feed/solvent contact and a centrifuge for separating extracted solids from the solvent containing the extracted oil and water. These designs were primarily suitable when a sludge type feed was available. Performance was quite good for sludges. Drier feeds caused excessive loading and maintenance of the centrifuges, while soils and non-pumpable feeds required front-end mixing with water, greatly increasing the amount of material that had to be treated and adding additional waste streams to the process. As the scope of the hazardous waste remediation problems became clear, the market for soils processing equipment far exceeded that for pumpable type waste streams. RCC developed a new basic design for the B.E.S.T. process that would successfully process soils but also allow pumpable sludges to be processed.

A B.E.S.T. system configured for soils and pumpable sludges and sediments uses a dual extractor design. An extractor/dryer vessel (secondary extractor) is used for direct soil extraction, solids separation and solids drying. High liquid wastes are contacted initially in a mixer/settler type extractor (primary extractor) that allows gross dewatering and organic reduction prior to transferring of the settled solids to the extractor/dryer vessel for final extraction and solids drying. All extraction is performed in a batch mode. While less elegant than a continuous process, batch extraction allows better control over extraction variables such as retention time and solvent to feed ratio and more assurance that solids meet disposal requirements.

Table 3	Pesticide contaminated soils	s slop oil emulsion treated solid	ls
Constituent (mg/l)	Feed	Product Solids	Removal Eff. %
p,p'-DDT	500 mg/l	0.2 mg/l	99.96
p.p'-DDE	84	0.5	99.4
p,p'-DDD	190	0.05	99.97
Endosulfan-I	250	< 0.02	> 99.99
Endosulfan-II	140	< 0.02	>99.99
Endrin	140	0.02	99.99
Dieldrin	37	< 0.02	>99.95
Toxaphene	2600	0.9	99.97
BHC-Beta	< 30	< 0.13	_
BHC-Gamma (Lindane)	< 30	< 0.07	_
Pentachlorophenol	150	1.9	98.7



BY WEIGHT CELL



FIGURE 3(e). Solids settling and solvent draining.

FIGURE 3(a). Soils metering.



FIGURE 3(b). Washer/dryer charging.









In soil treatment, the extractor/dryer vessel allows the contaminated soil to be handled only once. The treated solids are not moved from the extractor until the treatment level is reached. Figure 3 illustrates the typical processing sequence of a load of contaminated soil. Feed front-end handling currently includes screening to remove material greater than 1/2" in size. Extraction equipment to handle material from 1/2" it about 1 1/2"-2" is being investigated. Feed material in excess of 2" must be crushed before processing. Screened material is loaded into special transfer hopper containing about three yards of soil and transferred to the extractor/dryer vessel. The various processing steps including solvent addition, extraction/ mixing, decantation, and solvent drying are identified in Figure 3.

FIGURE 3(c). Solvent addition.



FIGURE 3(d). Extraction.



FIGURE 4. B.E.S.T. solvent extraction process.

Figure 4 shows a schematic of a B.E.S.T. unit with the dual extractor design. After each extraction step (see Figure 3) the solids are allowed to settle and the solvent mixture and particulates are separated by simple decantation of the liquid phase from the extractor vessel. Carryover of fine particulates from this step can be controlled by using a centrifuge or filter if necessary.

After each extraction step, the spent solvent is discharged to a holding tank. The spent solvent, which contains the oil and water removed from the feed, is heated to about 140° F to separate water without the expense of evaporation. The solvent phase, after water decantation, is evaporated to recycle the solvent and permit the oil product to be discharged. The recovered oil is stored for subsequent treatment, recycle, or destruction. The water separated during decantation is steam stripped to remove residual solvent and discharged from the unit for disposal or reuse.

Upon final extraction of the solids, the extractor/dryer vessel is heated by direct and indirect addition of steam. Residual solvent is vaporized and steam stripped from the extracted solids product. Upon completion of drying the solids are discharged to a product hopper and sent to final disposal.

CONCLUSIONS

Solvent extraction using the B.E.S.T. process is a proven separation process for treatment of contaminated soils, sludges, and sediments. The B.E.S.T. solvent extraction process allows the hazardous portion of a waste to be separated and concentrated, minimizing the total waste volume and with less environmental impact than destructive technologies. The B.E.S.T. process has evolved into a highly flexible semi-batch process that can handle soil, sludges, and sediment feeds by incorporating two types of extraction concepts within a single process unit.

LITERATURE CITED

- 1. Tose, M. K., "PCB Waste Minimization Using Triethylamine (B.E.S.T.[®]) Solvent Extraction," Presented at the PCB Forum, Houston, Texas, August 1989.
- Weimer, L. D., "B.E.S.T.[®] Solvent Extraction Treatment of PCB Contaminated Soils from Natural Gas Pipeline Sites," Presented at Haztech International Conference, Houston, Texas, February 1991.

Pneumatic Pumping Test for Soil Vacuum Extraction

Jong Soo Cho and Dominic C. DiGiulio

Robert S. Kerr Environmental Research Laboratory, U.S. Environmental Protection Agency, Ada, Oklahoma 74820

In-situ pneumatic pumping tests were performed to estimate the pneumatic permeability at a site containing soils contaminated with aviation gasoline. Determination of pneumatic permeability was necessary to evaluate soil-air discharge or pore volume exchange rates. Pressure propagation was measured in clustered vapor probes during the application of vacuum and positive pressure. An analytical solution for soil-air pressure distribution with a non-linear data fitting algorithm was adopted to obtain the pneumatic permeability from soil-air pressure distribution. Pneumatic pumping tests indicated substantially higher air discharge rates in the immediate vicinity of wells. The air discharge rate dissipated rapidly as distance from the wells increased. Application of increased vacuum or injection pressure resulted in substantially increased air flow in the immediate vicinity with small changes at a distance. This fact indicates that effective design should be based on air flow fields near wells, and the site specific design criteria should be determined with carefully conducted tests.

INTRODUCTION

Contamination of soils and ground water by fuels and organic solvents from leaking underground storage tanks and pipelines is common in the United States. Fuels and organic solvents contain components having low water solubililties and high vapor pressures. After initial release, they move down through the unsaturated soil zone as a non-aqueous phase liquid (NAPL), leaving residually saturated blobs in soil pores [1]. Liquid blobs trapped inside soil pores do not mobilize easily by intruding water, and limited solubilities keep them as longterm contamination sources to soil and ground water. Therefore, remediation of soil and ground water should include the removal of residual NAPL blobs from soil pores. Soil vacuum extraction (also known as soil venting) has proven to be a cost effective technology for removal of volatile organic compounds (VOCs) from residual NAPL blobs in the unsaturated soil. This in situ remediation process has been widely used in the United States since early 1980. Prior to 1980, this technology had been used primarily for methane gas control around landfills [2].

The principles of soil vacuum extraction are simple. The

evaporation of VOCs from soil pores is enhanced by artificially driven air flow. Soil vacuum extraction produces a pressure gradient, which drives air flow continuously into contaminated soils. Contaminated air is removed from subsurface soil through extraction wells and either released into the atmosphere or treated prior to discharge. A typical soil vacuum extraction system consists of extraction wells, injection wells, manifold piping, vacuum pumps, measuring gauges and controllers, and a vapor treatment system. Major advantages in soil vacuum extraction are as follows: disturbance of the contaminated soil is minimal, large volumes of soil can be treated, and standard equipment is used. Though a large number of field applications of soil vacuum extraction have been reported since early 1980, design of the field system has been empirical due to the variable field conditions and complexities of interacting processes [3, 4, 5]. Since system design is site-specific, a prototype or pilotscale system is commonly used to obtain system parameters before full scale implementation [6]. A large number of physical and system variables are expected to impact the applicability, design, operation, efficiency, and success of the field system. Among these variables, air-filled porosity and pneumatic permeability are key parameters in controlling convective



FIGURE 1. Source of Contamination and Plume Boundary at Testing Site

air flow in the unsaturated soil. Pneumatic permeability is defined as a product of the intrinsic permeability and the relative air permeability. Intrinsic permeability is defined as solely a function of soil pore structures, while relative air permeability is a function of air saturation in soil pores. When air occupies more than 80% of total porosity, relative air permeability becomes close to one for sandy soils [7].

Measurement of pneumatic permeability is well documented in the oil and gas industries [8]. Weeks [9] develoed a method of determining the vertical air permeability in soils by monitoring soil-air pressure fluctuation at various depths. Jakubick [10] used a vacuum method to measure pneumatic permeability and Miller et al., [11] reported an air-injection technique to study fractures in volcanic rock. Kearl et al., [12] reported air permeability measurements of the unsaturated tuff. They used three different methods: vacuum, air injection, and laboratory testing of rock core samples. Recently, Sellers et al., [13] reported various measurement methods on soil-air permeability for soil vacuum extraction.

This paper presents a report of pneumatic pumping tests conducted at an aviation gasoline contamination site to obtain site-specific parameters for the design of a large-scale soil bioventing system. Test procedures and results are presented using two-dimensional analytical solutions of soil-air pressure distribution in a layered soil reported by Baehr and Hult [14] and Joss et al., [15].

PNEUMATIC PUMPING TEST

Site Characterization

Pneumatic pumping tests were conducted in April, 1990, at the US Coast Guard Aviation Station in Traverse City, Michigan, to obtain design parameters for a bioventing project. Contamination of soil and ground water was caused by an accidental release of aviation gasoline in the autumn of 1969. An estimated 35,000 gallons (132,000 liters) were released [16]. A plume of aviation gasoline about 260 ft (80 m) wide and 1000 ft (300 m) long existed in the ground-water flow (Figure 1). A large number of soil and ground-water samples had been collected and analyzed. Though more than twenty years had passed since the release, a sizeable fraction of the spilled fuel was found in the unsaturated zone. It had been a continuous source of ground-water contamination. A highly concentrated zone was confined in a narrow band above the water table. The surface soil at the site was categorized as a fine sand with an average grain size of 0.35 mm. Hydraulic permeability was reported as $4.6 \times 10^{-10} \text{ ft}^2 (4.22 \times 10^{-7} \text{ cm}^2)$ [*I*6]. This sand layer had a very low organic content below the top 4 inches (10 cm) of soil. The sand layer was about 47 ft (14 m) thick and was underlain by a thick clay layer. The water table was located 17 ft (5.2 m) below the soil surface when testing was conducted. Annual fluctuation of the water table was ± 2 ft (0.6 m).

Test System

Two wells, air injection and vacuum extraction, were installed 20 ft (6 m) apart in an area close to the plume. Both wells were made of schedule 40, 5-inch (12.5 cm) diameter PVC pipe with a slotted screen section at 13 to 14 ft (3.9 to 4.2 m) below the soil surface. A pressure gauge was installed at the top of each well, and those wells were connected to separate blower pumps. Ball valves were attached to connecting pipes between the wells and the blower pumps to control the pressure and air flow. All the PVC pipes were connected with glued joints. Clustered vapor probes for soil air pressure measurement were installed at various distances along the line connecting the two wells. Clusters which were at the same location with wells had to be installed 1 ft (0.3 m) away from the line. Each cluster had three probes made of 1/4-inch (6.35 mm) O.D. copper tubing with a quick release connector at the top and a 2-inch (5 cm) meshed stainless steel screen at the bottom. The depths of the probes were 3, 9, and 14 ft (0.9, 2.7, and 4.2 m) from the soil surface. A 6-inch (15 cm) diameter, hollowstem auger was used to drill the holes for wells and probe clusters. The schematic diagram of the test system is illustrated in Figure 2. Pressures at the probes were measured with Magnahelic gauges. After the pump was started, the well pressure was adjusted by controlling the valve on the bypass lines. Soilair pressures at the cluster probes and differential pressures on pitot tubes were measured after the well pressure was stabilized. The volumetric flow rate of air from each well was measured with a pitot tube and Magnahelic pressure gauge. For air injection, the maximum capacity of the blower pump was found to be 176 scfm (4.9 std. m³/min) at 52.3 inch H₂O (13 kPa). Due to the high permeability of the soil, pressure propagation reached steady state shortly after the pump was started. Pressure gauges in the range of less than 1 inch H₂O (0.25 kPa) were found to be very sensitive to the windy field conditions.



FIGURE 2. Schematic Diagram of Testing System

Environmental Progress (Vol. 11, No. 3)



MATHEMATICAL MODEL

Pressure Distribution in Cylindrical Coordinate System

Baehr and Hult [14] and Joss et al., [15] presented analytical solutions to soil-air pressure distribution assuming radial symmetry around an extraction well and homogeneity in soil properties. Soil-air pressure at steady state can be described by

$$k_r \frac{\partial^2(p^2)}{\partial r^2} + k_r \frac{1}{r} \frac{\partial(p^2)}{\partial r} + k_z \frac{\partial^2(p^2)}{\partial z^2} = 0$$
(1)

where k, and k, are pneumatic permeabilities in radial and axial directions (in this test, they are considered to be identical with horizontal and vertical directional values). The water table constitutes a lower boundary impermeable to air. A solution was obtained for the unsaturated zone with soil surface open to atmosphere as the upper boundary (Figure 3). The boundary conditions are as follows:

$$p = p_{\text{atm}}$$
 at $z = 0$ (1a)

$$\frac{\partial p}{\partial z} = 0$$
 at $z = b$ (1b)

(1c)

$$p = p_{\text{atm}}$$
 at $r \to \infty$

at $r = r_w$,

$$\frac{\partial p}{\partial r} = 0 \quad 0 < z < d$$

$$\frac{\partial (p^2)}{\partial r} = -\frac{Q}{\pi k_r (l-d) r_w} \quad d < z < l \quad (1d)$$

$$\frac{\partial p}{\partial r} = 0 \quad l < z < b$$

The solution is

where m = n - 1/2 and $M_m = (m\pi)/b$.

Specific Discharges

The specific discharges in radial and axial directions, V, and V_{z} , are evaluated with Darcy's law assuming that the slip flow (Klingenberg effect) is negligible for sandy soil [14].

$$V_r = -\frac{k_r}{\mu} \frac{\partial p}{\partial r}$$
(3)

$$V_z = -\frac{k_z}{\mu} \frac{\partial p}{\partial z} \tag{4}$$

in which

 $\frac{\partial p}{\partial r} = \frac{1}{2p} \frac{\partial (p^2)}{\partial r}$ (5)

$$\frac{\partial p}{\partial z} = \frac{1}{2p} \frac{\partial (p^2)}{\partial z} \tag{6}$$

With the relationship of the Bessel functions,

$$\frac{dK_o(r)}{dr} = -K_1(r) \tag{7}$$

the specific discharges, V_r , and V_z , can be derived

$$V_{r} = -\frac{Q}{\pi^{2} p (l-d) r_{w}} \left\{ \sum_{n=1}^{\infty} \frac{1}{m} \left[\frac{\cos\left(\frac{m\pi d}{b}\right) - \cos\left(\frac{m\pi l}{b}\right)}{K_{1}\left(M_{m} \frac{r_{w}}{a}\right)} \right] \times K_{1}\left(M_{m} \frac{r}{a}\right) \sin\left(\frac{m\pi z}{b}\right) \right\}$$
(8)

$$V_{z} = -\frac{Q}{\pi a p (l-d) r_{w}} \left\{ \sum_{n=1}^{\infty} \frac{1}{b} \left[\frac{\cos\left(\frac{m\pi d}{b}\right) - \cos\left(\frac{m\pi l}{b}\right)}{M_{m} K_{1}\left(M_{m} \frac{r_{w}}{a}\right)} \right] \times K_{0}\left(M_{m} \frac{r}{a}\right) \cos\left(\frac{m\pi z}{b}\right) \right\}$$
(9)

The analytical solutions of air pressure distribution and specific discharge were programmed for a personal computer with FORTRAN, and the program was used for simulations.

$$p^{2} = p_{atm}^{2} + \frac{2aQ\mu}{\pi^{2}k_{r}(l-d)r_{w}} \left\{ \sum_{n=1}^{\infty} \frac{1}{m} \left[\frac{\cos\left(\frac{m\pi d}{b}\right) - \cos\left(\frac{m\pi l}{b}\right)}{M_{m}K_{1}\left(M_{m}\frac{r_{w}}{a}\right)} \right] K_{0}\left(M_{m}\frac{r}{a}\right) \sin\left(\frac{m\pi z}{b}\right) \right\}$$
(2)

Table 1 Air Permeabilities and Soil Anisotropies (Vacuum Extraction Tests)									
Pwell (inch H2O)	Air Flow (scfm)	f^* (inch H ₂ O) ²	$k_r (ft^2) \times 10^{10}$	k_z (ft ²)×10 ¹⁰	Anisotropy (k_r/k_z)				
- 50	- 132	3.18	9.75	2.17	4.49				
- 50	- 117	2.99	8.62	1.94	4.44				
- 42	- 100	2.32	8.48	2.10	4.03				
- 36	- 83	1.05	7.81	2.37	3.29				
- 24	- 68	0.44	9.44	2.74	3.44				
- 20	- 57	0.23	9.09	3.21	2.83				
- 16	- 57	0.11	11.38	3.81	2.99				
Averaged Value			9.22	2.53	3.64				
Std. Deviation			1.15	0.68	0.68				
*: value of equation (10)									

Parameter Estimation by Non-linear Least Square Data Fitting

Two unknown parameters, pneumatic permeabilities, k, and k_z , can be obtained by a non-linear data fitting method [17]. The method adopts a pseudo-Newtonian algorithm to search the multivariable values which minimize the objective function. In this study, the objective function is defined as

$$f = \sum_{i}^{N} (p(r_{i}, z_{i}) - p_{i})^{2}$$
(10)

where $p(r_i, z_i)$ is the estimated pressure at the observation point (r_i, z_i) with equation (2), p_i , is the observed pressure at the point (r_i, z_i) , and N is the number of observation points. For each test, 16 observation point data, including the well pressure, were used.

RESULTS AND DISCUSSION

A series of vacuum extraction, air injection, and combined operation tests were conducted. Permeabilities obtained from vacuum extraction and air injection tests are listed in Tables 1 and 2. Directional permeabilities obtained in both tests show a reasonable consistency within one standard deviation. The horizontal permeabilities obtained from air-injection tests show slightly higher values than those from vacuum extraction tests, and the vertical permeabilities show lower values. From these differences, anisotropic ratios from air injection tests become ten times higher than those from vacuum extraction tests. In general, permeabilities obtained from pneumatic pumping tests agree reasonably well with the hydraulic permeability value of 4.6×10^{-10} ft² (4.2×10^{-7} cm²) reported by Ostendorf et al. [16] and the pneumatic permeability, 3.5×10^{-10} ft² (3.2×10^{-7} cm²), which was estimated from the particle size of $D_{15} = 0.16$ mm [18].

Figure 4 shows the pressure distributions at three different depths. The well pressure was -50 inch H₂O (-12.4 kPa) and the air flow rate was 132 scfm (3.68 std. m³/min). The solid lines were generated by substituting pneumatic permeabilities,

 $k_r = 9.8 \times 10^{-10}$ ft² (8.9×10⁻⁷ cm²) and $k_z = 2.2 \times 10^{-10}$ ft² (2.0×10⁻⁷ cm²), into equation (2).

A large pressure gradient and air flow are expected near the well, with a corresponding decrease as the distance from the well increases. In order to define the effective zone of a well, the radial discharges, V_r , at 14 ft (4.2 m) depth with two well pressures were calculated and are plotted in Figure 5. The radial discharge at the well screen is 8.8×10^3 ft/hr (2.7 × 10³ m/hr) for -50 inch H₂O (-12.4 kPa) well pressure. The specific discharge is reduced to 50 ft/hr (15 m/hr) at 15 ft (4.5 m) and 1.2 ft/hr (0.36 m/hr) at 30 ft (9 m) from the well. The specific discharges with smaller vacuum (-16 inch H₂O) also show the similar trend of decrease, but in a moderate rate. Volumetric flow rates and well vacuum pressures are plotted in Figure 6 which shows a linear correlation. A larger vacuum in the well can produce higher discharges at close vicinity around the well than the smaller vacuum application, but their differences are reduced rapidly as the distance increases from the well. It seems that increased vacuum application in the well does not provide a significant increase on the radius of influences in the unsaturated zone open to the atmosphere.

SUMMARY AND CONCLUSIONS

From the pilot-scale test, pneumatic permeabilities at the aviation gasoline spill site could be obtained. Permeability estimations could be achieved by using an analytical solution of the pressure distribution and a non-linear data fitting algorithm. Some degrees of anisotropy were observed in soils. Estimated pneumatic permeabilities were in a reasonable agreement with previously reported hydraulic permeabilities.

Concerning pumping efficiency, various factors should be considered in the design of a full-scale operation. Even with high vacuum in the well, most air flow is limited to the immediate vicinity of the well. The VOC removal rate within a short distance from the well is expected to be rapid and much slower at greater distances. Therefore, a single well with high pressure removes VOCs nonuniformly from the contaminated area and may not be efficient to clean up a wider area. The cleanup time depends on the amount of direct air flow through the contaminated soils. Prior to the initiation of a full-scale

	Table 2 Air Permeabilities and Soil Anisotropies (Air Injection Tests)							
P _{well} (inch H ₂ O)	Air Flow (scfm)	f^* (inch H ₂ O) ²	$k_r (\mathrm{ft}^2) \times 10^{10}$	k_z (ft ²)×10 ¹⁰	Anisotropy (k_r/k_z)			
44	174	13.7	16.1	0.569	28.3			
36	158	11.0	18.4	0.563	32.7			
24	113	7.33	21.7	0.386	56.2			
16	71	2.81	20.2	0.459	43.6			
Averaged Value			19.1	0.494	40.2			
Std. Deviation			2.41	0.088	12.5			



FIGURE 4. Soil-Air Pressure Distributions, $p_{well} = -50$ inch H₂O, (—) estimated from equation (2), (^{*}) measured values, (A) depth = 3 ft, (B) depth = 9 ft, (C) depth = 14 ft

design, a pilot-scale pneumatic pump test provides valuable information on optimum parameters. Additional tests, such as tracer gas injections, will be useful to verify the parameters.

ACKNOWLEDGMENTS

The authors are grateful to those who helped conduct the tests, especially engineers at the Traverse Group and Solar Universal Technology, Inc. Drs. John T. Wilson and Don Kampbell and Mr. Frank Beck at the Kerr Lab are acknowledged for their encouragement and help. Although the research reported in this article has been funded wholly or in part by the United States Environmental Protection Agency, it has not been subjected to Agency review and therefore does not necessarily reflect the views of the Agency. No official endorse-



FIGURE 5. Specific Discharges in Radial Direction, (--) $p_{well} = -50$ inch H₂O, (...) $p_{well} \simeq -16$ inch H₂O, (A) $r \ge 3$ ft, (B) $r_w \le r < 3$ ft

ment of the system design or trade names should be inferred. The reported testing was not conducted under an approved Quality Assurance Project Plan.

NOTATION



- b = depth to water table from soil surface
- d = depth to the top of well screen from soil surface
 k, = pneumatic permeabilities in radial (horizontal) direction



FIGURE 6. Volumetric Flow Rates vs. Extraction Well Vacuum

- k_z = pneumatic permeabilities in axial (vertical) direction
- K_0 = the zero-order modified Bessel function of the second kind
- K_1 = the first-order modified Bessel function of the second kind
 - l = depth to the bottom of well screen from soil surface
- N = Number of observation point data used in data fitting
- p = soil air pressure
- $p_{\rm atm}$ = atmospheric pressure
- $p(r_i, z_i) =$ estimated pressure at location $i(r_i, z_i)$
 - p_i = observed pressure at location *i*
 - Q = volumetric flow rate at standard state
 - scfm = standard cubic feet per minute
 - V_r = specific discharge in radial (horizontal) direction
 - V_z = specific discharge in axial (vertical) direction
 - μ = viscosity of air

LITERATURE CITED

- Hunt, J. R., N. Sitar, and K. S. Udell, "Nonaqueous Phase Liquid Transport and Cleanup 1. Analysis of Mechanisms," *Water Resources Research*, 24(8), p. 1247 (1988).
- Mohsen, M. F. N., G. J. Farquhar, and N. Kouwen, "Gas Migration and Vent Design at Landfill Sites," Water, Air, and Soil Pollution, 13, p. 79 (1980).
- 3. Agrelot, J. C., J. J. Malot, and M. J. Visser, "Vacuum: Defense System for Ground Water VOC Contamination," In Proceedings of The Fifth Symposium and Exposition on Aquifer Restoration and Ground Water Monitoring, Columbus, OH (1985).
- Bruckner, F., "Sanierung von mit leichtflüchtigen chlorierten Kohlenwasserstoffen (LCKW) kontaminierten Grundwasserleitern mittels Kombinatin von Druchlufteinblasung und Bodenluftabsaugung," Korrespondenz Abwasser, 34(4), p. 863 (1987).
 Malot, J. J., "Unsaturated Zone Monitoring and Recov-
- Malot, J. J., "Unsaturated Zone Monitoring and Recovery of Underground Contamination," In Proceedings of The Fifth Symposium and Exposition on Aquifer Restoration and Ground Water Monitoring, Columbus, OH (1985).
- Hutzler, N. J., B. E. Murphy, and J. S. Gierke, "State of Technology Review, Soil Vacuum Extraction Systems," Final Report to U.S. EPA, CR-814319, Hazardous Waste Engineering Research Laboratory, Cincinnati, OH (1989).

- Parker, J. C., R. J. Lenhard, and T. Kuppusamy, "Physics of Immiscible Flow in Porous Media," EPA/600/2-87/101, US EPA, Office of Research and Development, R. S. Kerr Environmental Research Laboratory, Ada, OK (1987).
- Katz, D. L., D. Cornell, R. Kobayashi, F. H. Poettman, J. A. Vary, J. R. Elenbaas, and C. F. Weinaug, "Handbook of Natural Gas Engineering," McGraw-Hill, New York, NY (1959).
- 9. Weeks, E. P., "Field Determination of Vertical Permeability to Air in the Unsaturated Zone," Geological Survey Professional Paper 1051, U.S. Department of Interior, Washington, D.C. (1978).
- Jakubick, E. F., "Permeability Measurements of the Nearfield of Excavations by Transient Pressure Testing," Energy Canada Technical Report, Whiteshell Nuclear Research Establishment, Pinawa, Manitoba, Canada (1983).
- Miller, C. H., D. R. Cunningham, and M. J. Cunningham, "An Air Injection Technic to Study Intensity of Fractures around a Tunnel in Volcanic Rocks," *Bulletin of the As*sociation of Engineering Geologists, 2(3), p. (1974).
- Kearl, P. M., R. J. Zinkl, J. J. Dexter, and T. Cronk, "Air Permeability Measurements of the Unsaturated Bandelier Tuff Near Los Alamos, New Mexico," *Journal of Hydrology*, 117, p. 225 (1990).
- Sellers, K. L., T. A. Pederson, and C. Y. Fan, "Soil Vacuum Extraction: Air Permeability Testing and Estimation Methods," In Proceedings of the 17th RREL Hazardous Waste Research Symposium, EPA/600/9-91/002, Cincinnati, OH (1991).
- Baehr, A. L., and M. F. Hult, "Evaluation of Unsaturated Zone Air-Permeability Through Pneumatic Tests," Water Resources Research, 27(10), p. 2605 (1991).
- Joss, C. J., A. L. Baehr, and J. M. Fischer, "A Field Technique for Determining Unsaturated Zone Air Permeability," Presentation at Robert S. Kerr Environmental Research Laboratory Soil Venting Workshop, Houston, TX (1991).
- Ostendorf, D. W., D. H. Kampbell, J. T. Wilson, and J. H. Sammons, "Mobilization of Aviation Gasoline from a Residual Source," *Research Journal of Water Pollution Control Federation*, 61, p. 1684 (1989).
- 17. Kahaner, D., C. Moler, and S. Nash, "Numerical Method and Software," Prentice-Hall, Englewood Cliffs, NJ (1989).
- Massman, J. W., "Applying Groundwater Flow Models in Vapor Extraction Systems Design," ASCE Journal of Environmental Engineering, 115(1), p. 129 (1989).

Adsorption/Desorption Characteristics of Lead on Various Types of Soil

Robert W. Peters and Linda Shem

Energy Systems Division, Argonne National Laboratory, 9700 South Cass Avenue, Argonne, Illinois 60439

Laboratory studies were conducted to address the phenomena of adsorption/ desorption of lead onto various types of soils, both in the absence and presence of the chelating agent, ethylenediaminetetraacetic acid (EDTA). The linear and Freundlich isotherm models provided adequate description of the adsorption/ desorption behavior. Over the range of EDTA concentrations employed in the study (0.1 to 0.10M), no significant difference in the isotherm parameters was observed as a result of the applied EDTA concentration. The presence of EDTA significantly altered the adsorption/desorption behavior of lead on the soil, resulting in less of the metal being adsorbed. The soil with the higher silt/clay content had a greater amount of lead adsorbed onto it (as compared with the sandy soil).

BACKGROUND

The contamination of soils by heavy metals has resulted from a number of activities, including vehicle emissions; mining, smelting, and metal plating/metal finishing operations; automobile battery production; application of industrial waste, fertilizers, and pesticides; and fly-ash from incineration/combustion processes. Reported ranges of heavy metal concentrations from various soils are summarized in Table 1.

Heavy Metal Removal Techniques

Techniques for removal of metals from soil generally involve bringing the soil into contact with an aqueous solution. Such methods as flotation and waste classification are solid/liquid separation processes. Metal contamination is generally found on the finer soil particles, so separating the finely divided material may substantially reduce the heavy metal content of the bulk soil. Separating the fines from the coarse fraction was found to remove 90% of the metals but only 30% of the total dredged material [9]. In flotation processes, metallic minerals become attached to air bubbles that rise to the surface, forming a froth and thereby being separated from particles that are wetted by water [10]. Although flotation has been successfully used in the mineral processing industry, it has not been widely used for the treatment of contaminated soils [11]. Solid/liquid separation processes represent a preliminary step in the remediation of contaminated soils. The metals are still bound to a solid phase, but in a much more concentrated form. The advantage of this preliminary processing step is that the mass of contaminated soil to be processed is reduced considerably; therefore, the treatment cost is reduced.

Pickering [12] identified four approaches whereby metals could be mobilized in soils: (1) changing the acidity, (2) changing the system ionic strength, (3) changing the oxidation/reduction (REDOX) potential, and (4) forming complexes. In the last technique, the addition of complexing ligands can convert solid-bound heavy metal ions into soluble metal com-

Table 1 Reported Heavy Metal Concentrations						
Heavy Metal	Site Type	Reported Concentration Range (ppm)	Reference			
Zn	Smelting	26,000-80,000	[1]			
Cd	Smelting	900-1,500	[7]			
Cr	Chromium Production	500-70,000	[2]			
Pb	Battery Reclamation					
	Soil	2-135,000	[3, 4]			
	On-Site Sediment	2.16-42,700	[3, 4]			
	Surface Water	0-140 mg/L	[5]			
Pb	Battery Recycling	-				
	Soil	210-75,950 mg/kg	[6]			
	Soil	0-211,300mg/kg	[7, 8]			

Correspondence should be directed to Robert W. Peters. The viewpoints expressed here are not necessarily those of Argonne National Laboratories or its sponsors.

	Table 2 Properties of Chelating Agents												
folecular		Acid	lity Con	stants			Log	of the S for the M	tability Metal Ch	Constant ielate			
Weight	pK ₁	pK ₂	pK ₃	pK₄	pK ₅	Cd	Zn	Cu	Pb	Mn ⁺⁺	Fe ⁺⁺⁺		
191	1.89	2.49	9.73			10.5	11.2	13.7	11.8	8.1	17.0		
292	2.08	3.01	6.40	10.44		17.5	17.2	19.7	17.7	14.5	26.5		
393	2.08	2.81	4.49	8.73	10.6	20.1	19.7	22.6	21.0	16.7	29.2		
	olecular Weight 191 292 393	olecular pK1 191 1.89 292 2.08 393 2.08	$\begin{array}{c} \text{olecular} \\ \hline \text{Weight} \\ \hline pK_1 \\ pK_2 \\ \hline pK_1 \\ pK_2 \\ \hline pK_2 \\ 292 \\ 2.08 \\ 3.01 \\ 393 \\ 2.08 \\ 2.81 \end{array}$	$\begin{tabular}{ c c c c c c c } \hline & & Acidity Constraints & & & & & & & & & & & & & & & & & & &$	Acidity Constants Weight pK1 pK2 pK3 pK4 191 1.89 2.49 9.73 292 2.08 3.01 6.40 10.44 393 2.08 2.81 4.49 8.73	$\begin{array}{c c} \mbox{olecular} \\ \hline \mbox{Weight} \\ \hline \mbox{W} \mbox{PK}_1 \\ \hline \mbox{PK}_1 \\ \hline \mbox{PK}_2 \\ \hline \mbox{PK}_3 \\ \hline \mbox{PK}_4 \\ \hline \mbox{PK}_5 \\ \hline \mbox{PK}_5 \\ \hline \mbox{PK}_4 \\ \hline \mbox{PK}_5 \\ \hline \mbox{PK}_5 \\ \hline \mbox{PK}_4 \\ \hline \mbox{PK}_5 \\ \hline $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		

^aNitrilotriacetic acid.

^bEthylenediaminetetraacetic acid.

^cDiethylenetriaminepentaacetic acid.

plexes. The effectiveness of complexing ligands in promoting the release of metals depends on the strength of bonding to the soil surface, the stability and adsorbability of the complexes formed, and the pH of the suspension [11]. From an application viewpoint, the type and concentration of the complexing ligand and the system pH are the operational parameters that can be controlled.

The ability of chelating agents to form stable metal complexes makes materials such as ethylenediaminetetraacetic acid (EDTA) and nitrilotriacetic acid (NTA) promising extractive agents for the treatment of soils polluted with heavy metals [13]. Elliott and Peters [11] have noted that although complexation is the major mechanism responsible for the metal solubilization, the overall release process depends on the hydrogen ion concentration and the system ionic strength. Because hydrous oxides of iron and manganese can coprecipitate and adsorb heavy metals, they are believed to play an important role in the fixation of heavy metals in polluted soils [14]. Their dissolution under reducing conditions may weaken the solid heavy-metal bond and thereby promote solubilization of the metal ions. Elliott and Peters [11] noted that there are five major considerations in the selection of complexing agents for soil remediation:

- Reagents should be able to form highly stable complexes over a wide pH range at a 1:1 ligand-to-metal molar ratio.
- (2) Biodegradability of the complexing agents and metal complexes should be low (especially if the complexing agent is to be recycled for reuse in the process).
- (3) The metal complexes that are formed should be nonadsorbable on soil surfaces.
- (4) The chelating agent should have a low toxicity and low potential for environmental harm.
- (5) The reagents should be cost effective.

Elliott and Peters [11] note that, although no compounds ideally satisfy all these criteria, there are several aminocarboxylic acids that form remarkably stable complexes with numerous metal ions. The properties of three chelating agents that have been used to extract contaminants from soil (either for analyses or remediation) are summarized in Table 2.

Extraction of heavy metals from contaminated soils can be performed by either using *in-situ* techniques or on-site extraction (following excavation). In the case of *in-situ* soil flooding, the aqueous extractive agent is allowed to percolate through the soil to promote metal mobilization. In the case of on-site extraction following excavation, the operation can be performed on a batch basis, semi-batch basis, or continuously. The contaminated soil is first pretreated for size reduction and classification and then brought into contact with the extractive agent. Finally, the soil is separated from the spent extractive agent, and the effluent is further recycled to decomplex and precipitate the heavy metals from solution. Alternatively, the solution is treated by using electrodeposition techniques to recover the heavy metals.

Adsorption Isotherm Characterization

The adsorption of heavy metals onto soil was examined by application of three commonly used adsorption isotherm models: the linear, Langmuir, and Freundlich models. Equations describing these models are listed below:

 $Linear: \qquad q_e = kC + q_{e,0} \tag{1}$

Langmuir: $q_e = \frac{Q^o bC}{1+bC}$ (2)

Freundlich: $q_e = K_N C^{1/n}$ (3)

where q_e is the amount of solute (heavy metal) adsorbed per unit weight of adsorbent, C is the measured solute concentration in solution at equilibrium, Q' is the amount of solute adsorbed per unit weight of adsorbent in forming a complete monolayer coverage on the adsorbent (soil) surface, b is a constant related to the net enthalpy of adsorption, and k, K_N , and q_{e0} are empirical constants.

These equations are linearized using the equations listed below.

inear:
$$q_e = kC + q_{e,0}$$
 (4)

Langmuir: $\frac{1}{q_e} = \frac{1}{O^0} + \frac{1}{bO^0} \frac{1}{C}$ (5)

Freundlich:
$$\ln q_e = \ln K_N + \frac{1}{n} \ln C$$
 (6a)

ог

$$\log q_e = \log K_N + \frac{1}{n} \log C.$$
 (6b)

GOALS AND OBJECTIVES

I.

The primary goal of this project was to determine and compare the performance of several chelating agents [ethylenediaminetetraacetic acid (EDTA) and nitrilotriacetic acid (NTA)] for their ability to extract lead (Pb) from contaminated soil. The focus of this paper is to determine the adsorption/desorption behavior of lead onto several soil types using EDTA as the extractant.

Table	3	Characteristics	of	the	Soils	Used	in	the	Ad-
		sorption/De	so	rptic	n Stu	dies			

Туре			Classi	fication (wt %)
	pH	Texture	Sand	Silt	Clay
Α	7.75	Silty clay	14.6	39.0	46.4
B		Loamy sand	78.5	12.2	9.3



FIGURE 1. Characteristics of lead adsorption onto Soil A in the presence of EDTA

EXPERIMENTAL PROCEDURE

The laboratory experiments were performed using a batch shaker technique to investigate the adsorption/desorption behavior of lead onto various soil types. The experimental procedure associated with the adsorption and desorption studies are described below.

Adsorption Studies

Uncontaminated soil was prepared by grinding the soil with a ceramic mortar and pestle until the ground soil could pass through a 850-µm (ASTM mesh No. 20) sieve. The initial characteristics of the two different soil types are summarized in Table 3. Each soil type was spiked (artificially contaminated) with a solution of lead nitrate and left in contact for a 3.0hour time period on the shaker table. Nominal initial lead concentrations applied to the soil were 500, 1000, 5000, and 10,000 mg/kg soil. The soil was weighed in 5-gm portions using a top-loading balance and placed in plastic, lidded shaker containers. To these containers, 45 mL of a EDTA solution was added. Replicate samples were prepared on every fourth sample. The EDTA concentrations of the extracting solution were nominally 0, 0.01, 0.05, and 0.10 M. This matrix of initial lead and EDTA concentrations enabled the adsorption/desorption characteristics of lead onto the various soil types to be described both in the absence and in the presence of EDTA.

The soil samples (to which the metal and extracting agent had been added) were shaken for 3 hours at a low setting on an Eberbach shaker table. This time requirement was determined on the basis of preliminary experiments in which the



FIGURE 2. Linear isotherm for lead adsorption onto Soil A in the presence of EDTA



FIGURE 3. Linearized Freundlich isotherm for lead adsorption onto Soil A in the presence of EDTA

residual concentration was monitored as a function of time on the batch shaker table [15]. This analysis showed that chemical equilibrium could be achieved within 3.0 hours. Following agitation, the samples were centrifuged in plastic Nalgene centrifuge tubes equipped with snap-on caps, filtered using No. 42 Whatman filter paper, and stored in glass vials maintained at pH < 2 (prepped using HNO₃) prior to atomic absorption spectroscopy (AAS) analysis. The Buck Scientific Atomic Absorption/Emission Spectrophotometer, Model 200-A, was calibrated using AAS lead standards. The lead analyses were performed in accordance with standard procedures [16]. The residual lead concentration remaining in solution was determined from the AAS analyses. The amount of lead adsorbed per unit weight onto the soil (for each concentration of EDTA) was determined using the difference between the initial concentration of the solution applied to the soil and the final residual concentration after treatment. Using the different initial lead concentrations, the adsorption isotherms were determined using the linearized isotherm equations presented in equations (4) through (6).

Desorption Studies

Uncontaminated soil was again prepared by grinding the soil with a ceramic mortar and pestle to enable the soil to pass through a 850-µm sieve. Each soil sample was spiked with a different lead nitrate concentration (nominally 500, 1000, 5000, and 10,000 mg/kg soil, which were assumed to be the initial



FIGURE 4. Linear isotherm desorption characteristics for lead from Soil A using EDTA as the extractant



FIGURE 5. Linearized Freundlich desorption isotherm for lead from Soil A using EDTA as the extractant

lead concentrations on the soil samples) and left to air dry. Concentrations were checked by AAS analysis of nitric acid extractions of the soil. The contaminated soils were then aged in glass jars (with screw-on covers) for approximately three months prior to performing the desorption experiments.



FIGURE 6. Comparison of the adsorption/desorption characteristics of lead onto Soil A for EDTA concentrations of (a) 0.01 M, (b) 0.05 M, (c) 0.10 M

	Adso	orption	Desc	orption
EDTA Concentration (moles/L)	<i>k</i> (L/kg)	$q_{e,0} \ ({ m mg/kg})$	<i>k</i> (L/kg)	$q_{\epsilon,0}$ (mg/kg)
0.00	5570	335		
0.01	5.40	107	5.90	10.5
0.05	6.28	17.4	5.96	-152
0.10	4.43	-46.7	5.38	-20.7

These contaminated soils were then weighed in 5-gm portions using a top-loading balance and placed in plastic containers that had lids. To these containers, 45 mL of an EDTA solution (or deionized water) was added. The procedure for determining the adsorption isotherms then follows that described for the adsorption studies.

RESULTS AND DISCUSSION

Figure 1 presents the adsorption characteristics of lead onto Soil A for various EDTA concentrations ranging from 0 to 0.10 M. Figures 2 and 3 present the linearized isotherms using the linear and Freundlich isotherm models. The Langmuir model did not adequately describe the adsorption of lead onto the various soils due to the fact that negative values of Q^o and b were obtained from the linearization of several Pb/EDTA combinations.

Figure 4 presents the desorption characteristics of all three EDTA concentrations on Soil A using the linear adsorption model. All three EDTA concentrations resulted in nearly the same desorption characteristics. Figure 5 presents the linearized Freundlich isotherm for the three EDTA concentrations. The slopes of the resulting lines are nearly equal, indicating the desorption behavior is independent of the EDTA concentrations used in this study (0.01 to 0.10 M).

Figures 6a through 6c compare the adsorption/desorption characteristics for the lead remaining on Soil A for EDTA concentrations of 0.01, 0.05, and 0.10 M, respectively. These plots did not indicate significant differences in the adsorption/ desorption characteristics on Soil A. Tables 4 through 6 show the adsorption/desorption isotherms using the linear, Freundlich, and Langmuir isotherm models. The values of kfor the linear isotherm and K_N and 1/n for the Freundlich isotherm models are similar for both the adsorption and desorption studies. Statistical analyses were conducted, and ttests performed on the differences in the mean values were not significant at a 0.9 level of significance, showing the common behavior of the adsorption and desorption systems. As described earlier, the Langmuir isotherm model does not adequately describe the adsorption of lead onto soil because negative values of Q° and b were obtained for the EDTA concentrations of 0.01 and 0.05 M.

Table	5	Freundlich	Adsorption/Desorption	Isotherm
		Model for	Lead onto Soil Type A	

EDTA Concentration (moles/L)	Adso	rption	Desor	rption
(moles/ E)	1/n	K _N	1/n	K _N
0.00	0.71	4760		
0.01	1.05	4.69	0.92	9.44
0.05	1.10	3.30	0.95	7.05
0.10	0.93	6.29	0.94	7.58

Table 6 Langmuir Adsorption/Desorption Isotherm Model for Lead onto Soil Type A

EDTA	Adso	orption	Desorption		
Concentration (moles/L) 0.00	Q°	b	Q°	b 	
	1790	15.6			
0.01	-26,800	-0.00019	6130	0.00127	
0.05	- 30,900	-0.00016	5920	0.00105	
0.10	5340	0.00096	6930	0.00092	

One striking feature noted from inspection of Tables 4 and 5 is the drastic change in adsorption behavior associated with the presence of EDTA. The value of k for the linear isotherm model is nearly 1000 times larger with no EDTA present as compared with the case when EDTA is present. Similarly, for the Freundlich model, the parameter K_N is likewise nearly 1000 times greater for the case of no EDTA present versus the case where EDTA is present. The exponent 1/n is least 24% lower for the case of no EDTA present versus EDTA being present. A value of 1/n < 1 indicates a strong affinity for adsorption. Thus, the presence of EDTA as a complexing agent decreases the affinity for lead to adsorb onto the soil, resulting in an inhibition to lead adsorption.

Furthermore, the extraction of lead from this soil type using EDTA was pH-insensitive over the pH range of 4 to 12 [15, 17]; the removal of lead ranged from 58% to 64% over the entire range of initial lead concentrations. Also, the applied EDTA concentrations over the range of 0.01 to 0.10 M had little effect on the removal efficiency of lead from the soil [15, 17].



FIGURE 7. Linear isotherm for lead adsorption onto Soil B in the presence of EDTA



FIGURE 8. Linearized Freundlich isotherm for lead adsorption onto Soil B in the presence of EDTA



FIGURE 9. Comparison of the adsorptive behavior of lead onto Soils A and B with no cheating agents present using the linear isotherm model

Figure 7 presents the adsorption characteristics onto Soil B for EDTA concentrations ranging from 0 to 0.10 M. This figure once again indicates a much higher amount of lead adsorbed onto the soil in the absence of EDTA as compared to the amount of lead adsorbed in the presence of EDTA. Figure 8 presents the analogous adsorption characteristics of lead onto Soil B using the Freundlich isotherm model. All three EDTA concentrations result in nearly identical values of the exponent, 1/n.

Figure 9 compares the adsorptive characteristics of lead onto the two soil types, A and B, for the case of no chelants being



FIGURE 10. Comparison of the adsorptive behavior of lead onto Soils A and B in the presence of EDTA using the linear isotherm model



FIGURE 11. Comparison of the adsorptive behavior of lead onto Soils A and B using the Freundlich isotherm model

Table 7	Comp	ar	ison	0	L	near	Lead	Ads	orp	tion	onto Soil
	Types	A	and	B	in	the	Prese	nce	of	EDT	A

EDTA Concentration (moles/L)	Soil	Гуре А	Soil Type B		
	k (L/kg)	<i>q_{e,0}</i> (mg/kg)	k (L/kg)	q _{e,0} (mg/kg)	
0.00	5570	335	5.68	2310	
0.01	5.40	107	4.20	- 53.5	
0.05	6.28	17.4	2.71	36.1	
0.10	4.43	- 46.7	1.78	175	

present. The adsorptive capacity for lead is much, much greater for the case of Soil A, due to the much larger silt/clay fraction for Soil A as compared with Soil B. Because Soil B is more sandy, the amount of lead adsorbed onto the soil remains relatively constant over large residual lead concentrations in solution. Figure 10 compares the adsorptive characteristics of lead onto the two soil types in the presence of EDTA. Once again, the adsorptive capacity of lead is much larger (generally 2-3 times larger) on Soil A as compared with Soil B due to the larger silt/clay fraction. Heavy metals are known to preferentially bind to clays and humic materials [9]. Figure 11 presents the analogous description of the adsorptive behavior of lead onto the two soil types in the presence of EDTA using the Freundlich isotherm model. The amount of lead adsorbed onto Soil A is typically about 0.5 order of magnitude larger than the amount of lead adsorbed by Soil B under comparable experimental conditions. Tables 7 and 8 compare the adsorptive behavior of Soils A and B using the linear and Freundlich isotherm models, respectively.

SUMMARY AND CONCLUSIONS

These laboratory studies investigated the adsorption/desorption phenomena for lead onto two soil types. Soil A had a high silt and clay content (\sim 85%), whereas Soil B had a very high sand content (>78%). The adsorption/desorption phenomena was studied for various initial lead concentrations, ranging from 500 to 10,000 mg/kg soil; these studies were performed both in the presence and absence of EDTA. The linear and Freundlich isotherm models provide an adequate description of the adsorption/desorption behavior of lead onto these two soil types. In contrast to this behavior, the Langmuir isotherm model did not provide an adequate description of this behavior, due to values of Q^0 and b obtained in the isotherm linearization having negative values.

There was no significant difference in the adsorption/desorption characteristics of lead on either soil for EDTA concentrations in the range of 0.01 to 0.10 M. The presence of EDTA, however, significantly altered the adsorption/desorption characteristics of lead onto both soil types. The presence of EDTA resulted in less of the metal being adsorbed onto the soil due to the strong complexing ability of the chelating agent.

There was a significant difference in the adsorptive behavior of lead onto these soils, which is postulated to be due to the

Table 8 Comparison of Freundlich Lead Adsorption onto Soil Types A and B in the Presence of EDTA

EDTA	Soil 7	Type A	Soil Type B		
(moles/L)	1/n	K _N	1/n	K _N	
0.00	0.71	4760	1.39	2.10	
0.01	1.05	4.69	1.21	0.99	
0.05	1.10	3.30	1.16	1.02	
0.10	0.93	6.29	1.11	1.05	

much higher silt/clay content for soil A as compared with a higher sand content for Soil B. Previous studies have indicated that heavy metals are preferentially bound to clays and humic materials [9].

ACKNOWLEDGMENTS

The work reported here was supported in part by the U.S. Department of Energy, under contract W-31-109-Eng-38. This paper was presented at the 23rd Annual Meeting of the Fine Particle Society, held in Las Vegas, NV, on July 13-17, 1992.

NOMENCLATURE

- b = Constant related to the net enthalpy of adsorption, used in the Langmuir isotherm model
- C = Measure solute (heavy metal) concentration in solution at equilibrium, (mg/L)
- EDTA = Ethylenediaminetetraacetic acid
 - k = Linear adsorption coefficient, (L/kg)
 - K_N = Emirical constant in the Freundlich isotherm model
 - 1/n = Exponent in the Freundlich isotherm model
- NTA = Nitrilotriacetic acid
- $pH = -\log [H^+]$
- $q_{e,0}$ = Empirical constant used in the linear adsorption isotherm model, corresponding to the amount adsorbed with no addition of soil (i.e., an extrapolation form the q_e versus C plot)
- Q^{o} = Amount of solute adsorbed per unit weight of soil in forming a complete monolayer on the surface of the soil, (mg/kg)

LITERATURE CITED

- Oyler, J. A., "Remediation of Metals-Contaminated Site Near a Smelter Using Sludge/Fly Ash Amendments," Proc. 44th Purdue Indus. Waste Conf., 44: 75-82 (1989).
- Hsieh, H. N., D. Raghu, J. W. Liskowitz, and J. Grow, "Soil Washing Techniques for Removal of Chromium Contaminants from Soil," *Proc. 21st Mid-Atlantic Indus. Waste Conf.*, 21: 651–660 (1989).
- Trnovsky, M., J. P. Oxer, R. J. Rudy, M. J. Hanchak, and B. Hartsfield, "Site Remediation of Heavy Metals Contaminated Soils and Groundwater at a Former Battery Reclamation Site in Florida," pp. 1581-1590 in *Hazardous Waste, Detection, Control, Treatment*, R. Abbou, Ed., Elsevier Science Publishers, B. V. Amsterdam, The Netherlands (1988).
- Trnovsky, M., J. P. Oxer, R. J. Rudy, G. L. Weinstein, and B. Hartsfield, "Site Remediation of Heavy Metals Contaminated Soils and Groundwater at a Former Battery Reclamation Site in Florida," pp. 88-90 in Vol. I, Sixth Internat. Conf. on Heavy Metals in the Environ., S. E. Lindberg and T. C. Hutchinson, Eds., New Orleans, LA (1987).
- Trnovsky, M., P. M. Yaniga, and R. S. McIntosh, "Lead Removal—The Main Remedial Design Parameter for Two Florida Superfund Sites," pp. 233-236 in Vol. II, Seventh Internat. Conf. on Heavy Metals in the Environ., J.-P. Vernet, Ed., Geneva, Switzerland (1989).
- Hassling, J. L., M. P. Esposito, R. P. Traver, and R. H. Snow, "Results of Bench-Scale Research Efforts to Wash Contaminated Soils at Battery-Recycling Facilities," pp. 497-514 in Vol. II, Metals Speciation, Separation, and Recovery, J. W. Patterson and R. Passino, Eds., Lewis Publishers, Inc., Chelsea, MI (1989).
- Elliott, H. A., J. H. Linn, and G. A. Shields, "Role of Fe in Extractive Decontamination of Pb-Polluted Soils," *Haz. Waste & Haz. Mater.*, 6: 223-229 (1989).

- Elliott, H. A., G. A. Brown, G. A. Shields, and J. H. Linn, "Restoration of Pb-Polluted Soils by EDTA Extraction," pp. 64-67 in Vol. II, Seventh Internat. Conf. on Heavy Metals in the Environ., J. P. Vernet, Ed., Geneva, Switzerland (1989).
- Werther, J., R. Hilligardt, and H. Kroning, "Sand from Dredge Sludge-Development of Processes for the Mechanical Treatment of Dredged Material," p. 887, in *Contaminated Soil*, J. W. Assink and W. J. van der Brink, Eds., Martinus Nijhoff, Dordrecht, The Netherlands (1986).
- Aplan, F., "Flotation," in *Encyclopedia of Chemical Technology*, Vol. 10, M. Grayson and D. Eckroth, Eds., 3rd ed., John Wiley & Sons, New York, NY (1980).
- Elliott, H. A., and R. W. Peters, "Decontamination of Metal-Polluted Soils Using Chelating Agents," in *Removal of Heavy Metals from Groundwaters*, CRC Press, Inc., Boca Raton, FL (in press).
- 12. Pickering, W. F., "Metal Ion Speciation-Soils and Sed-

iments (A Review)," Ore Geology Reviews, 1: 83 (1986).

- Assink, J. W., "Extractive Methods for Soil Decontamination; a General Survey and Review of Operational Treatment Installations," p. 655, in *Contaminated Soil*, J. W. Assink and W. J. van der Brink, Eds., Martinus Nijhoff, Dordrecht, The Netherlands (1986).
- 14. Slavek, J., and Pickering, W. F., "Extraction of Metal Ions Sorbed on Hydrous Oxides of Iron (III)," Water, Air, and Soil Pollut., 28: 151-162 (1986).
- 15. Peters, R. W., and L. Shem, "Use of Chelating Agents for Remediation of Heavy Metal Contaminated Soil," ACS Sympos. Series on Separations Science in Environmental Chemistry: Separation of Metal Ions and Organics from the Environment (in press).
- Standard Methods for the Examination of Water and Wastewater, 15th Ed., American Public Health Assoc., Washington, D.C. (1981).
- Peters, R. W., and L. Shem, Argonne National Laboratory, Argonne, IL, unpublished information (1992).

Electrokinetic Removal of Selected Heavy Metals from Soil

Sibel Pamukcu

Department of Civil Engineering, Fritz Engineering Laboratory 13, Lehigh University, Bethlehem, PA 18015-4791

and

J. Kenneth Wittle

Electro-Petroleum, Inc., Suite 1118, Old Eagle School Road, Wayne, PA

The electrokinetic process is an emerging technology for in situ soil decontamination, in which chemical species, both ionic and nonionic are transported to an electrode site in soil. These products are subsequently removed from the ground via collection systems engineered for each specific application. Electrokinetics refer to movement of water, ions and charged particles relative to one another under the action of an applied direct current electric field. In a porous compact matrix of surface charged particles such as soil, the ion containing pore fluid may be made to flow to collection sites under the applied field.

The work presented here describes part of the effort undertaken to investigate electrokinetically enhanced transport of soil contaminants in synthetic systems. The results of the laboratory study presented here indicate that electrokinetic enhancement of contaminant transport in soils is a viable technology; development of this technology for wide range of applications hinges upon better understanding of the transient chemical and physical processes during application of current through soil/contaminant systems.

INTRODUCTION

Methods for cleaning up hazardous waste sites have changed since 1980 when the Comprehensive Environmental Response, Compensation and Liability Act, (CERCLA) or Superfund, was enacted. Early remedial actions for contaminated soils consisted primarily of excavation and removal of the contaminated soil from the site and disposal at a landfill. More emphasis is now placed on *in situ* treatment of soil. Electrokinetic decontamination is perhaps one of the most promising *in situ* soil decontamination process capable of removing heavy metals and organic contaminants from soil feasibly.

Electrokinetic decontamination invokes three electrokinetic processes: electroosmosis, electrophoresis, and electrolytic mi-

gration of ionic and polar species. Electroosmosis produces a rapid flow of water in low permeability soils and probably contributes significantly to the decontamination process in clay soils. Electrophoresis is the migration of charged colloids in soil-liquid mixture. In a compact system electrophoresis should be of less importance since the solid phase is restrained from movement. In some cases, however, electrophoresis may play a role in decontamination if the migrating colloids have the chemical species of interest adsorbed on them. During application of electricity to soil, water at the electrode sites as well as in the soil pores undergoes electrolysis, dissociating into its components, H^+ and OH^- . Electrolytic migration is the migration of H^+ (produced at anode) and OH^- (produced at cathode) toward the opposite electrode. This migration is re-

sponsible for conducting the current in a soil-water system. As ions migrate toward an electrode, they can drag layers of water molecules, which may be of significant quantity if the ionic concentration of water content is high. Other reactions inherent to electrokinetics are ionic diffusion, and electrodic reactions at the electrode sites. These reactions may or may not play significant roles in decontamination depending on the specific field process implemented and process control.

One of the important aspects of electrokinetics in soil-water systems is the transient migration of an acid front from the anode site to the cathode site during treatment [1, 2, 3]. When water decomposes, it is reduced to hydrogen gas at the cathode and oxidized to oxygen gas at the anode. This results in progressive reduction of pH at the anode site, and increase of pH at the cathode site. Subsequently the hydrogen ions produced at the anode migrate toward the cathode whereas the hydroxide ions produced at the cathode migrate toward the anode. Since the ionic velocity of the hydrogen ion is higher than that of the hydroxide ion, the H⁺ would move faster into the soil. Sundberg [4] reports the relative mobility of H⁺ as 318.0 versus 17.4 for OH⁻. This movement would further be enhanced by the electroosmotic flow of water toward the cathode. Hydrogen ion will exchange with metal ions held on clay surfaces. In addition low pH condition is favorable for the dissolution of metal complexes and precipitates. Therefore, acid front migration is beneficial for metal extraction from soils, provided that soil's natural buffering capacity is sufficiently low. The transient and spatial variation of pH has further implications that aid in metal extraction. Soil surface properties such as cation exchange capacity, ion (cation and anion) adsorption capacity, and magnitude of surface potential and its sign are pH dependent. Furthermore, speciation and solubility of the contaminants are often pH dependent. The distribution of these contaminants would be transient and spatially varied with pH. In addition to the transient nature of pH, the soil redox potential would vary spatially in time during electrokinetic treatment. The pH-redox conditions at a point at any time during the treatment would determine the solubility and speciation of most heavy metal constituents. These conditions may limit or enhance the movement of the metals to an electrode site

Extraction of contaminants by electrokinetic method is based on the assumption that the contaminant is in liquid phase in the soil pores. Sufficient flux of water produced by electroosmosis should be able to move non-ionic as well as ionic species through soil toward the cathode. This is perhaps best achieved when the state of the material (dissolved, suspended, emulsified, etc.) is suitable for the flowing water to carry it through the tight pores of soil without causing an immovable plug of concentrated material to accumulate at some distance from the anode, which would also restrict the flow of water. Polar organic molecules should orient themselves in the direction of the electric field, similar to water molecules, and move toward one of the electrode sites depending on their polarity, mobility and the surface charge characteristics of the soil medium. Removal of cationic species would be enhanced with the combined effect of electroosmotic flow of water and the electrolytic migration of the cation toward the cathode electrode. It is yet unclear the relative magnitude of contribution of either process (electroosmosis and electrolytic migration) to decontamination under a given set of initial and boundary conditions. However, recent findings indicate that at low concentrations of cations, electroosmotic water flow may contribute a significant percentage, if not all, to the overall decontamination process [5]. At high concentrations of the ionic species, electrolytic migration, intensity of electrochemical reactions, auxiliary water movement by hydraulic drag exerted on the hydrated ions (cations and anions) may play more important roles than electroosmosis in the decontamination process.

Earlier experimental observations by others [6, 7] also support the conclusion stated above. These investigations had shown significantly higher current efficiencies (volume of flow

per quantity of electricity) for low ionic concentrations or undissociated organic compounds in the pore fluid than those with high ionic concentrations in the pore fluid. Others [8] have argued that current efficiencies increase with increased ionic concentration in clay suspensions. This is a valid argument for colloidal suspensions in which true electroosmotic flow is probably not the predominant mechanism of water transport, but is the hydraulic drag exerted on the migrating ions. On the basis of the Donnan theory, a high water to cation concentration ratio promotes high current efficiency. Therefore in slurried systems, increased transport of water with increased cation concentration should hold true provided that the water concentration remains constant. In a compact system of soil, the rate of frictional dragging of water by ionic species through the pores is limited by the size of the pores and the pore throats. Therefore the ions may move as fast as they would in a loose suspension, however water will not. Under these conditions, increased ionic concentration at constant water content should decrease current efficiency according to the Donnan concept and observations made by others [7] and investigators of the work presented here.

BACKGROUND

Historical Development of Electrokinetic Processes in Soils

Electrokinetic phenomenon was first discovered by Reuss in 1808. The phenomenon was first treated analytically by Helmholtz in 1879, which later was modified by Pellat in 1904 and Smoluchowski in 1921. This theory is widely known as Helmholtz-Smoluchowski theory which relates electroosmotic velocity of a fluid of certain viscosity and di-electric constant, through a surface charged porous medium of zeta potential, ζ , under an electric gradient. The theoretical computation of electroosmotic water transport, by this theory does not always agree well with experimental observations. The zeta potential term in the H-S theory varies with pH and ionic concentration of the pore fluid, which is not a constant during electrokinetic treatment. Similarly, electric gradient is not constant in time and space due to the changing resistivity and oxidation/reduction state of the soil.

A notable approach to electrokinetic process in porous medium was made by Spiegler [9]. He considered the interactions of the mobile components of soil (water and ions) on each other and the frictional interactions of these components with the pore walls. The true electroosmotic flow was expressed as the difference between the measured water transport and the ion hydration in moles per Faraday. This quantity was suggested to be directly proportional to concentration of free water in the soil pores and indirectly proportional to the concentration of mobile counterions in moles per volume. This suggests the phenomenon of water transport in the opposite direction of electroosmotic flux by the electrolytic migration of the anions. Therefore, at any time during electrokinetic treatment if the concentration of anionic species in the free water exceeds the cationic species, the opposite flow may retard the net flow toward the cathode.

Gray and Mitchell [7] showed experimentally that the electroosmotic flow increases with increasing water content of most soils, however decreases with increasing electrolyte concentration of the pore fluid. This electrolyte concentration effect is more evident in clays with high anion retention capacity, such as kaolinite, as opposed to montmorillonite [10]. Therefore high electroosmotic flows are observed in kaolinite than montmorillonite type of clays at the same concentration of dilute solutions of electrolytes. This is consistent with the prediction of electroosmotic water transport according to Donnan concept.

Khan [5] proposed a modified theory of electroosmotic flow through soil. In this theory the true electroosmotic flow is directly proportional to the current carried by the surface of the charged soil constituents of soil. The slipping plane potential (zeta potential) used in Helmholtz-Smoluchowski theory of electroosmotic flow is replaced by a constant surface potential which is invariable with electrolyte concentration. Hence, true electroosmotic flow becomes independent of electrolyte concentration in the pore fluid. The measured values agree well with the predicted flow using this modified theory. It also agrees with Speigler's [10] formulation and much earlier observations of Napier in 1846 [12] who stated that "... the measurable endosmose (electroosmosis) seems to be greatest when the current has greatest difficulty to pass through, and when the decomposition (of water) was least." This follows that the true electroosmotic flow is large when ratio of the surface current to the electrolytic current (carried by the ions in the pore fluid) is large due to reduced concentration of ions in the pore fluid.

Application of Electrokinetics to Soil Decontamination

Electrokinetics have been used for dewatering of soils and sludges since the first recorded use in the field by Casagrande in 1939 [12]. Work and subsequent research in electrokinetic decontamination of soils have been accelerated in recent years, probably after the report of the detection of high concentrations of metals and organics in electroosmotically drained water of a dredged sludge in the field by Segall and co-workers in 1980 [13]. Other field work [14, 15] has been conducted since then with reasonable success of heavy metal transport.

Hamnett [16] performed laboratory studies on electrically induced movement of ions in sand. In evaluating the electrolytic migration of ions of different salts, he found that smaller ions (i.e., Na) were more mobile than larger ions (i.e., K, Ca, Ni). Acar et al. [17] and Hamed et al. [18] showed that the migration of an acid front from the anode toward the cathode is significant in the removal of heavy metals from clay soils. Shapiro and co-workers [3] developed a model for electroosmosis to predict transient behavior of concentration profiles for chemical species in solution. The model agreed well with experimental results of acetic acid removal from soil via electrokinetics.

A simple prediction method was suggested by Khan [5] recently. The theoretical approach developed for electroosmotic flow was used to predict the contaminant removal from soil when the contaminants are either present at low concentrations in the soil pores or they are not adsorbed onto the charged surfaces, so that the true electroosmotic flow is the dominant mechanism for their migration. Tracer experiments in kaolinite clay utilizing O-Nitrophenol and zinc as representative contaminants were conducted. O-Nitrophenol represents a poorly adsorbed organic compound, the adsorbtion of which was measured on the order of 0.01 mg/g of soil. The zinc concentration used was 325 mg/kg of dry soil which was slightly over than the typical maximum concentration of naturally occurring zinc (10-300 mg/kg) in soils. There was good agreement between the measured removal of the contaminant and the predicted removal using electroosmotic flux of water based on clay surface current.

INVESTIGATION

Experimental Procedures

An extensive laboratory program was undertaken to investigate the feasibility of electrokinetic treatment of clay mixtures containing heavy metals. The test matrices were prepared in the laboratory by compressing slurries of clay to compact saturated soil samples. Each clay slurry was mixed with solution of one heavy metal salt of a predetermined concentration. The mixing water of the slurries were also changed as distilled water, a solution of a humus product representing humic acids in natural soils, and water with dissolved salts representing the ionic constituents of a selected groundwater type. Each specimen type (clay, pore solution and metal type) was prepared in triplicate and tested under the same conditions, but with variable duration of treatment for some of triplicate sets. In the overall testing program, 11 different metals (As, Cd, Co, Cr, Cs, Hg, Ni, Pb, Sr, U, Zn) were used. The initial mixing concentrations of some of these metals were varied from a typical low to a high value occurring in field situations. In this paper, results of electrokinetic treatment and implications of a few of the clay + pore water + metal combinations are presented. The metals used in these combinations were Cd, Co, Ni and Sr.

Materials

The clays used in preparing the soil matrices were kaolinite (Georgia kaolinite) and Na-montmorillonite (bentonite), both obtained commercially. The sand specimens were prepared with washed New Jersey beach sand passing No. 40 sieve and retained on No. 100 sieve. This variation in soil mineral matrix was intended to investigate the process in so assumed difficult situations of remediation when the ionic species are strongly held on the clay surface (Na-montmorillonite case), or when the electroosmotic flow efficiency might be reduced due to reduced surface charge density (sand case). The sand matrix was prepared with the clean beach sand mixed homogeneously with 10% Na-montmorillonite by dry weight.

The pore solutions were distilled water, humic solution and simulated groundwater. The humic solution was prepared by dissolving a commercially obtained humus product in distilled water at a concentration of 900 ppm. The organic carbon content of this stock solution was 257 as measured in mg C/1. This corresponded to DOC values typically measured in groundwaters associated with very high organic carbon fields, such as oil fields. The pH of this solution was 9.4. The simulated groundwater was prepared following an approximate formula of the groundwater salt constituents typically found in the Jefferson County of Idaho State. The pH of groundwater sampled in that region was measured at 8.2, and contained a substantial amount of (212 ppm) bicarbonate. The simulated groundwater did not contain this component. It's pH was measured as 7.9.

The salts used in preparing solutions of the 4 metals and the metal concentration of each resulting solution as well as the resulting concentration of the metal in soil and the soil pH prior to electrokinetic treatment are given in Table 1. Some of these metal solutions were prepared at so called "high" and "low" concentrations to represent variable degree of contamination of the soil matrix. In general the pHs of the mixtures were moderately acidic or near neutral. In most cases the pH was reduced with the addition of the metal salt, probably because they all were salts of strong acids.

Procedures

The slurries would be mixed at 100% water content of the kaolinite clay, 50% water content of the sand mixture, and 200-300% water content of Na-montmorillonite clay, all by dry weight of the solid components. The mixing would be done in large batches (2-3 gallons) with a tall hand held mixer. Then the slurry would be poured into individual compression units referred to as "consolidometers." The operation of these units are presented elsewhere [5]. The main features of the consolidometers are that they are capable of applying some increments of sustained stress (pneumatically) while allowing for

	Ta	able 1 Some Properties	of the Materi	als Used in Pr	eparing Specim	iens	
			Solu	tion	Se	oil	
Sample			Concentra	tion (mg/l)	Concentrati	on (mg/kg)	Initial
Code*	Metal	Salt Formula	High	Low	High	Low	Soil pH
KS	Blank		-	_	_	_	4.36
	Cd	$CdCl_2 \cdot (2.5)H_2O$	1000	40	666	40	4.15
	Co	CoCl ₂ ·6H ₂ O	1000		677	_	3.33
	Ni	Ni(NO ₃) ₂ ·H ₂ O	1000	-	395		3.40
	Sr	SrCl ₂ ·6H ₂ O	1000	-	686	-	4.31
KH	Blank			—	_	_	6.26
	Cd		1000		748		5.98
	Co	(same as above)	1000	_	599	-	5.73
	Ni		1000		701	_	6.26
	Sr		971	_	N/A	—	5.96
KG	Blank		_	-	-	-	5.37
	Cd		1000		630	-	5.38
	Co	(same as above)	1000		550		5.49
	Ni	*	1000	_	1084	-	3.90
	Sr		971	<u> </u>	N/A	-	5.97
MS	Blank		-	_	_	_	7.94
	Cd		1000	N/A	4972	882	6.37
	Co	(same as above)	1000	_	8603	-	6.13
	Ni		1000	-	5617	_	6.76
	Sr		971	_	4432	_	7.83
SS	Blank		_	_	_	_	7.34
	Cd		1000	N/A	N/A	N/A	5.46
	Co	(same as above)	1000	-	1020	_	6.18
	Ni	,	1000	_	598		7.41
	Sr		971	-	463	_	8.20

*KS: Kaolinite/distilled water

MS: Na-Montmorillonite/distilled water

KH: Kaolinite/Humic solution

SS: Sand + 10% Na-Montm./distilled water

KG: Kaolinite/Groundwater

the excess fluid to drain out slowly. This procedure creates, homogeneous, near saturated soil matrices compacted to a constant density and pressure. The most beneficial aspect of these custom manufactured units is that at the end of the consolidation process, the densified soil sample is packed into a cylindrical soil chamber (3.556 cm ID and 7.62 cm long) which can be detached from the unit and installed into the electrokinetic cell used in this investigation. Therefore the resulting soil specimen is not disturbed by handling such as trimming and placement.

All the specimens were consolidated to 200 kPa pressure. Although there is evidence, from this work and others [22] that, electroosmotic flow increases with consolidation pressure in clay soils, it is not well known to what extent the increased flow contributes to decontamination when the contaminant migration is not governed by electroosmosis. The average initial densities and water contents of the five artificial soil types are given in Table 2.

At the completion of each consolidation period, the tube containing the packed soil would be removed, the top extruding portion trimmed and stored for quantitative chemical and water content analysis. Samples of the waters extracted during consolidation were also taken for chemical analysis to assess the degree of metal retention by clays. The approximate period of consolidation to 200 kPa pressure was around 24 hours for kaolinite and it was around 7 days for Na-montmorillonite clays slurries. This period was much less for the sand + bentonite specimens, but pressure increment periods were maintained similar to that of kaolinite for consistency. When the soil cylinder was removed from the consolidometer, weighted and sampled, it would be ready for mounting on the electrokinetic apparatus.

The details of the electrokinetic apparatus used in this study are also provided elsewhere [5]. It should be noted that there

is no standard apparatus for measuring electrokinetic water flow in soils. The ones that are used in similar work elsewhere have been shown to produce consistent results. The equipment used in this particular program was developed with attention given to a few concerns. These were addressed by providing larger electrode surfaces than soil surface to achieve low current density, by isolating the electrodes from soil surfaces to minimize electrode reactions at these boundaries, and by providing convenient ports for gas expulsion from the electrode sites and also extracting inflow (anode site) and outflow (cathode site) water samples for analysis. The main features of these units are: the electrodes reside in water tight chambers which are connected to measuring burettes at one end, and to the soil specimen cylinder at the other end. The soil specimen is separated from these water filled chambers by carborandum po-

Table 2 Average water content and density of specimens before and after electrokinetic treatment

Sample Code	In	itial	Final		
	W %	density (g/cm ³)	W %	density (g/cm ³)	
KS	47.25	1.64	56.19	1.62	
KH	51.35	1.66	54	1.54	
KG	32.8	1.83	39.5	1.73	
MS	271.8	1.14	321.62	1.03	
SS	51.25	1.67	52.95	1.66	

KS: Kaolinite-distilled water

KH: Kaolinite-humic solution

KG: Kaolinite-ground water

MS: Montmorillonite-distilled water

SS: Sand & 10% Montmorillonite-distilled water



FIGURE 1. Variation of quantity of water flow versus number of moles of electrons transmitted in kaolinite with distilled water and cademium.

rous stones. The units can be operated under constant potential or constant current. The time dependent water movement through the soil is detected by monitoring the water levels in the inflow (anode side) and outflow (cathode side) tubes on individual control panels. These panels are also each equipped with a pressure regulator and gauge for hydraulic potential application capability. The control panels also carry small pumps to drain the electrode chamber periodically as necessary in order to remove the extracted material and alleviate concentration polarization.

The tests, results of which are reported here were conducted under a constant potential of 30 VDC across electrodes. The voltage in soil and the current generated were measured simultaneously with the flow measurements. The tests were continued for at least 24 hours. For a number of randomly selected combinations of soil, pore fluid and metal, the termination times were varied between the triplicate samples from 24 hours to up to 50 hours. For a number of extended hours of treatment, the electrode chambers (both anode and cathode) were



FIGURE 2. Normalized concentration variation of Cd (high concentration) from the anode to the cathode end of 3 replicate specimens of kaolinite with distilled water.



FIGURE 3. Normalized concentration variation of Cd (low concentration) from the anode to the cathode end of 3 replicate specimens of kaolinite with distilled water.

drained at around 24 hours and refilled with the appropriate pore water used in the soil specimen. This was done to alleviate concentration polarization and the high pH gradients created between the chambers which was thought to reduce water flow rate originally. However, this practice did not prove to be effective for increased contaminant removal or water flow rate, therefore it was abandoned in the later tests.

At the termination of each experiment, the chamber water would be collected. The cathode chamber water was acidified to solubilize the metal precipitates created at the high pH environment in that chamber. These water samples were analyzed for the particular metal extracted and their pHs were measured. The cathode water pHs ranged from around 10 to 12, whereas the anode water pHs ranged from 2 to 3 in general. The soil specimen would then be extracted, pH measurements taken at the anode end, center and the cathode end. Specimens from each one of these locations were collected for chemical analysis of the particular metal aimed for extraction.

RESULTS

The analysis of available data showed metal migration in all soil specimens. Considering the metal concentrations reached in the first half of the soil specimen (anode half) or the lowest concentration of a particular metal achieved at one of the three locations (anode end, center, cathode end) of the soil cylinder, the metal removal success ranged between 85-95% (based on original concentration). These values were mostly achieved with specimens containing the higher concentration of a particular metal. The slight variations in these removal percentages appeared to depend on soil matrix, the metal and pore fluid type. It should be noted here that, the concentration profiles of metals across the soil specimens were generated by sampling only three points along each specimen. Therefore the actual concentration profile at the time of sampling may not be well represented by the linear or best fit distributions assumed between the consecutive points. Therefore, it is not possible to predict degree of removal of the contaminant from the areas under the concentration curves, neither arrive at an accurate mass-balance prediction. Nevertheless, the data presented are consistent and show a clear trend of metal migration toward an electrode site in each case.



FIGURE 4. Normalized concentration variation of Cd (high concentration) from the anode to the cathode end of 3 replicate specimens of kaolinite with humic solution.

The lower concentration metal mixed soils did not show as high percentages of removal for the same durations of treatment. Figure 1 shows the correlation between moles of inflow water versus the quantity of electricity (moles of electrons) passed through two types of kaolinite soil specimens-each in triplicate-with distilled water and varying initial concentrations of cadmium measured in milligrams per kilogram of dry soil (Table 1). As observed, the low concentration Cd specimens exhibit larger influx of water than the high Cd concentration specimen for the same level of electricity. The average removal of Cd in the first half of soil cylinder was computed as 16.5% for low concentration case, whereas this percentage was 84 for the high Cd concentration case. Figures 2 and 3 illustrate the concentration profiles of Cd across the soil specimens based on the quantitative measurements made at three locations in the specimen. At low concentration of the metal, the current carried by the clay surface may constitute a larger portion of the total current and result in larger net flow of water in the cathode direction, or electroosmosis, as observed by Napier in 1846. At high concentrations of the metal the current is easier to pass through, because the bulk of it would be carried through the pore space. The quantity of true electroosmotic flow may remain the same, however the flow would probably be counteracted by the movement of a larger concentration of anions that drag water molecules along. The anion retention of the clay then plays an important role in the development of a net water flow to the cathode.

The other results pertaining to Cd removal in kaolinite in the presence of pore fluids other than distilled water are presented in Figures 4 and 5, for the humic solution and simulated groundwater, respectively. As observed, the initial retention of Cd in kaolinite does not change significantly when the pore water changes from distilled water. The removal trends, however, changes slightly. The low concentrations of the metal in the first half of the cell are repeated, but at the cathode end of the soil the concentrations are increased compared to that of the distilled water specimens. The average pHs measured at the cathode ends of these specimens were determined as 5.3, 9.0 and 8.1 for the distilled water, humic solution and groundwater specimens, respectively. The accumulation of a metal product may be due to two conditions: one is that the time rate of migration of a metal may be higher than its rate of release into the electrode chamber, and the other is that in-



FIGURE 5. Normalized concentration variation of Cd (high concentration) from the anode to the cathode end of 3 replicate specimens of kaolinite with simulated groundwater.

creased pH environment may favor non-ionic or anionic species of the metal which would then precipitate or tend to move in the opposite direction of the flow. The former condition may be due to a number of reasons such as concentration polarization in the electrode chamber, ionic diffusion due to the large pH gradient at the interface between the soil and the cathode chamber, increased level of electrochemical reactions close to the boundary which may result in retardation of the flux, or physical entrapment due to reduced permeability at the boundary. The increased pH in the latter two cases of Cd with humic solution and groundwater might have contributed to accumulation of the metal at the cathode end. This is in agreement with the distribution of hydrolysis products of Cd⁺ with pH. Until around pH 8, Cd remains in ionic form with + 2 charge. Above this value it starts forming complex species which are either charged positively or negatively or neutral. The tendency of each one of these products and their abun-



FIGURE 6. Average normalized concentration variation of Ni from anode to cathode end of specimens of kaolinite with distilled water, humic solution and simulated groundwater.



FIGURE 7. Average pH distribution from anode to cathode end in 3 duplicate specimens of four clay types with nickel at the end of electrokinetic treatment.

(KS: kaolinite/distilled water, KH: kaolinite/humic sol. KG: kaolinite/groundwater, MS: Na-Montmorillonite/dist. water)

dance will control the removal rate for some time until the acid front reaches the cathode end. Even so, the removal rate into the electrode chamber may still be affected significantly by the boundary conditions discussed above. Another important aspect that needs to be addressed is the increased adsorption and cation exchange capacity of the clay at high pH environments. Therefore, the accumulation of the metal at the cathode end would be favored by the changing surface characteristic of the soil at the interface with the cathode chamber where the pH is typically measured above 10.

A similar trend was observed with nickel mixed at high concentration into kaolinite. Figure 6 shows the average of the concentration profiles for high concentration nickel in kaolinite clay with the three different pore fluids. Once again the metal removal in the anode half of the soil is moderately high, on the order of 50% for distilled water and humis solution specimens and it is about 80% for the groundwater specimens. The average pH distribution at the completion of treatment from anode (x/L = 0) to cathode (x/L = 1) end of four Ni



FIGURE 8. Normalized concentration variation of Sr (high concentration) from the anode to the cathode end of 3 replicate specimens of kaolinite with distilled water.

2.5 INITIAL CONCENT #3 (24.5 HR.) Sr (mg/kg) 2153 2.0 2- 5588 NORMALIZED CONCENTRATION OF #1 (40.5 HR. 1.5 #2 (44 75HR.) 1.0 TOTAL Sr, C/Co 0.5 0.0 MIX WALER CONCENTRATION OF Sr=970.92 mg/l -0.5 -0.2 00 0.2 0.4 0.6 0.8 1.0 1.2 NORMALIZED DISTANCE FROM ANODE END OF SOIL, X/L (L=3 in)

FIGURE 9. Normalized concentration variation of Sr (high concentration) from the anode to the cathode end of 3 replicate specimens of Na-montmorillonite with distilled water.

carrying clay soils are presented in Figure 7. As expected, the smallest quantity of electricity corresponds to the distilled water kaolinite specimens, while the highest value corresponds to Na-montmorillonite specimens. The hydrolysis products of Ni are also complex ions of positive or negative charge or solid, which start to form above pH 7. Observing the chart in Figure 7, one can predict the types of Ni complex species that may occur along the specimen.

Strontium remains a divalent ion for a large range of pH values. Figure 8 illustrates successful removal of Sr from distilled water kaolinite specimens. The progressive reduction of the element at the cathode end also shows the transient nature of the process. Similar trends were observed with Sr in Namontmorillonite and sand + montmorillonite mixture with distilled water. These are presented in Figures 9 and 10, respectively. The average pH distributions are given in Figure 11, where the cathode end pHs are high for two of the soil types. Considering the Na-montmorillonite and sand + ben-



FIGURE 10. Normalized concentration variation of Sr (high concentration) from the anode to the cathode end of 3 replicate specimens of sand + 10% Na-montmorillonite with distilled water.



FIGURE 11. Average pH distribution from anode to cathode end in 3 duplicate specimens of five soil types with strontium at the end of electrokinetic treatment.

(KS: kaolinite/distilled water, KH: kaolinite/humic sol. KG: kaolinite/groundwater, MS: Na-montmorillonite/dist. water, SS: clayey sand/distilled water)

tonite specimens, the cathode pH should not effect strontium since it will remain a divalent ion even at high pHs. However, there is accumulation of the metal at the cathode end which appears to deplete in time, as shown in Figures 9 and 10. Once again, this delayed flux of the metal into the cathode chamber may be due to a number of factors some of which are ionic diffusion, migration of counterions, or increased levels of electrochemical reactions at the increased concentrations of the metal and also the exchange potential of the soil.

Finally, the concentration profile of cobalt in distilled water kaolinite is presented in Figure 12. The pH distributions for the clay soils with Co are given in Figure 13. The average pH at the cathode end of the distilled water kaolinite specimen was over 9, which is within the range of pHs for complex ion formations of Co. Depending on the local concentration of Co^{+2} at this pH, the predominant species may either be a positively charged ion or an hydroxide salt, Co(OH)₂.



FIGURE 12. Normalized concentration variation of Co (high concentration) from the anode to the cathode end of 3 replicate specimens of kaolinite with distilled water.



FIGURE 13. Average pH distribution from anode to cathode end in 3 duplicate specimens of four clay types with cobalt at the end of electrokinetic treatment.

(KS: kaolinite/distilled water, KH: kaolinite/humic sol. KG: kaolinite/groundwater, MS: Na-montmorillonite/dist. water)

Figure 14 presents the correlations between inflow quantity normalized by updated pore water volume, versus moles of electrons transferred for the three different soil types with distilled water. Each one of the soil samples contain one of the 4 metals, Sr, Ni, Co or Cd. The linear relationship is evident in each soil type, with kaolinite representing the most consistent behavior with a narrow span of variation. Sand + bentonite specimens have a larger span than that of kaolinite, whereas the montmorillonite relationship varies considerably. Flow through montmorillonite specimens appear to be more affected by the type of metal than it is for kaolinite specimens. At the top of the montmorillonite data span resides the flow data of Sr clay mixtures, then for Co mixtures and towards the bottom is the data for Ni mixtures. A trend is observed from the overall data, that is, for the same number of moles of electrons, the inflow to pore water volume ratio is highest for kaolinite and it is more or less independent of the type of the metal ion present in the pore fluid, when compared to other two soil types. It should be noted here that, this is valid for the four metals tested. The sand + montmorillonite trend is similar but slightly lower than that of kaolinite. Although montmorillonite appears to produce the lowest ratio, because of its large span,



FIGURE 14. The variation of normalized cumulative inflow versus the number of moles of electrons transmitted in distilled water specimens of the three types of soils with Cd, Co, Ni or Sr.



FIGURE 15. The variation of normalized cumulative inflow versus the number of moles of electrons transmitted in distilled water specimens of kaolinite with Cd, Co, Ni or Sr.

such a conclusion is not made at this time until further data is processed.

In Figure 15 normalized inflow versus moles of electrons is reproduced for kaolinite only. The slope of approximate straight line correlations change as the metal in the pore fluid varies. In general the trends are similar to those of metal + Na-montmorillonite mixtures shown in Figure 14. In this case, Sr and Cd specimen data coincide, exhibiting the largest flow per mole of electron value, whereas Ni shows the smallest value, among the four metal mixtures. The reduced flow may be due to the formation of complex species of ions produced as pH changes across a specimen. At times when there are predominantly negative complexes of a metal, these species may carry significant number of molecules of water as they migrate in the opposite direction of electroosmotic water flow, reducing the net flow towards cathode.

CONCLUSIONS

The main conclusion of this research has been that electrokinetics promises to be a viable technology to remove contaminants from soils. The interaction of various processes that take place simultaneously is yet to be understood to better predict the efficiency of the process in a variety of soil-contaminant situations. The results presented in this paper shed some light on the influence of electrochemistry and physicochemistry of the soil-contaminant system on the process. The transient acid front movement in soil from anode site to cathode, is beneficial for metal desorption and dissolution, which in turn contributes to the removal process. However, the transient pH condition cause formation of complex species of some it. Therefore pH control at both ends (anode and cathode) should create suitable environments for the metal ions to remain in solution as single species, so that realistic predictions of rate of electrokinetic extraction can be made.

The following specific conclusions were obtained from this study:

1. Electroosmosis is probably the dominant mechanism of electrokinetic extraction of metal contaminants when they are present at low concentrations. At high concentrations of the metal, electrolytic migration of their ionic species plays a more important role than the electroosmotic transport.

2. The removal of the 4 metals tested in this study from a variety of soil types, including montmorillonite, kaolinite, clayey sand, clay with simulated groundwater and clay with humic acids was achieved at 85-95% of the original concentration, characterized as typical high concentration of these metals occurring in some common sites.

3. At low concentration of Cadmium, the flow efficiency was higher than the high concentration case of the metal. However, percent removal was 16.5 as opposed to 84 percent for the latter case.

4. Among the 3 soil types tested, kaolinite exhibited highest efficiency of electrokinetic water flow, followed by clayey sand. The lowest efficiency appeared to occur with Na-montmorillonite. Kaolinite exhibited consistent efficiency with little variation with respect to the metal ion type. Whereas, montmorillonite and clayey sand appeared to be more affected by the type of metal ion.

ACKNOWLEDGMENTS

This research was funded by the Office of Technology Development, Office of Environmental Restoration and Waste Management, U.S. Department of Energy, managed by Argonne National Laboratory, Contract No. 2112406.

LITERATURE CITED

- 1. Acar, Y. B., Gale, R. J., Putnam, G., and Hamed, J., "Electrochemical Processing of Soils: Its Potential Use in Environmental Geotechnology and Significance of pH Gradients," Proc. of the 2nd Int. Sym. on Envir. Geotech., Vol. 1, pp. 25-38, Envo Publishing, Bethlehem, PA, 1989.
- 2. Acar, Y. B., Gale, R. J., Putnam, G., Hamed, J., and Wong, R., "Electro-chemical Processing of Soils: Theory of pH Gradient Development by Diffusion and Linear Convection," J. Envir. Sci. and Health, Part (a); Envir. Sci. and Engrg., 25(6), 1990.
- 3. Shapiro, A. P., Renaud, P. C., and Probstein, R. F., "Preliminary Studies on the Removal of Chemical Species from Saturated Porous Media by Electro-Osmosis," Physico Chemical Hydrodynamics, 11(5/6), 1989.
- Sundberg, K., *The Log Analyst*, May-June, 19-31, 1980.
 Khan, L. I., "Study of Electroosmosis in Soil: A Modified Theory and Its Application in Soil Decontamination," Dissertation, in partial fulfillment of requirements for the degree of Doctor of Philosophy, Lehigh University, Bethlehem, PA, 1991.
- 6. Renaud, P. O., and Probstein, R. F., "Electro-osmotic Control of Hazardous Waste," Physico Chemical Hydrodynamics, 9(1/2), pp. 345-360, 1987.
- 7. Gray, D. H., and Mitchell, J. K., "Fundamental Aspects of Electro-Osmosis in Soil," J. of the Soil Mech. and Found. Div., ASCE, SM 6, pp. 209-236, 1967.
- 8. Lockhart, N. C., "Electroosmotic Dewatering of Fine Tailings from Mineral Processing," Int. J. of Mineral Processing, Vol. 10, pp. 131-140, 1983.

- 9. Spiegler, K. S., "Transport Process in Ionic Membranes," Transactions of the Faraday Society, Vol. 54, Part 9, pp. 1408-1428, 1958.
- 10. Bohn, H. L., McNeal, B. L., and O'Connor, G. A., Soil Chemistry, John-Wiley & Sons, New York, 1985.
- 11. Napier, J., Phil. Mag., 29(10), 1846.
- 12. Casagrande, L., "Electro-Osmosis in Soils," Geotechnique, London, England, Vol. 1, No. 3, p. 159, 1949.
- 13. Segal, B. A., O'Bannon, C. E., and Matthias, J. A., "Electro-Osmosis Chemistry and Water Quality," J. of the Geotech. Eng. Div., ASCE, Vol. 106, No. GT10, pp. 1143-1147, 1980.
- 14. Lageman, R., "Theory and Practice of Electro-Reclamation," Nato Pilot Study: Demonstration and Remedial Action Technologies for Contaminated Land and Ground Water, NATO/CCMS, Copenhagen, Denmark, 1989.
- 15. Banerjee, S., Horng, J., Ferguson, J. F., and Nelson, P. O., "Field Scale Feasibility Study of Electrokinetic Remediation," Report, USEPA, Risk Reduction Laboratory, Cincinnati, OH, 1988.
- 16. Hamnett, R., "A Study of the Processes Involved in Electro-Reclamation of Contaminated Soils," Dissertation, University of Manchaster, England, 1980.
- 17. Mitchell, J. K., "Potential Uses of Electro-kinetics for Hazardous Waste Site Remediation," Presented at USEPA-University of Washington Workshop on Electrokinetic Treatment of Hazardous Waste Treatment, Seattle, Washington, 1986.
- 18. Acar, Y. B., Hamed, J., Gale, R. J., and Putnam, G., "Acid/Base Distributions in Electroosmosis," Transportation Research Record No. 1288, pp. 23-34, TRB, Washington, D.C., 1991.
- 19. Hamed, J., Acar, Y. B., and Gale, R. J., "Pb(II) Removal from Kaolinite by Electrokinetics," J. of Geotech. Eng. Div., ASCE, Vol. 117, No. 2, pp. 241-271, 1991.

Who needs ESL Information Services?

Find out how we can serve your needs:

(212) 705-7611

(212) 753-9568

- Literature Research
- Document Delivery
- Engineering & Technical Reference

DOCTORS

Biomedical Engineering Medical Instrumentation

□ LAWYERS

Patent searches; Patent infringement cases Author searches; experts, engineering literature

CHEMICAL ENGINEERS

Chemical process plants and chemical hazards Environmental issues and regulations

CIVIL ENGINEERS

Design of structures, structural analysis Historical civil engineering literature, bridges

□ ELECTRICAL ENGINEERS

Computer hardware and software engineering Electronics market, predictions and trends

MECHANICAL ENGINEERS Finite element analysis literature New products and instrumentation

MINING AND PETROLEUM ENGINEERS

Paraffin build-up, buckle propagation in pipelines Mining ventilation and safety

ALL OF THE ABOVE

ESL Information Services has the Engineering Societies Library (ESL) at its core, one of the world's largest technical libraries. We can find the literature on your engineering or technical subject, deliver you the documents and respond to a variety of engineering reference questions.

If the key to your success is information, stay ahead of the competition with ESL Information Services.



United Engineering Center 345 East 47th Street, NY, NY 10017 TEL: (212) 705-7611 FAX: (212) 753-9568

With Nearly \$20 Billion on the Line ... CPI Companies Can't Afford Improper Equipment These essential to standardized way • Perform deta

Nearly \$20 billion dollars! That's what the Chemical Processing Industries will spend this year on new plant and equipment. And that's why more and more CPI companies have made AIChE's Equipment Testing Procedures an *esential* part of company policy. Each concise manual shows you how to run your equipment through complete "top-to-bottom" tests designed to detect even the smallest problem or loss of efficiency.

All procedures were developed by the AIChE Equipment Testing Procedures Committee and reviewed by leading practitioners across the nation. AIChE's procedures are the sure way to maximize every dollar of your equipment investments.

New!

PACKED DISTILLATION COLUMNS

Complete procedures for testing the packing efficiency, capacity limitations, energy consumption, and pressure drop factors of continuously operated distillation and batch columns. Covers all types of randomly dumped, structured, and grid type packings with appropriate supports and distributors. The Most Thorough Testing Procedures Available for CPI Equipment!

These essential test procedures give CPI companies a reliable, standardized way to ...

- Perform detailed "acceptance" tests on new equipment.
- Calculate the capacities and optimum use of equipment.
- Collect accurate data for maintenance and troubleshooting.
- Provide "real world" numbers for computer simulations and design studies.

Each step-by-step manual helps the CPI engineer . . .

- Plan and implement "top-to-bottom" equipment performance tests.
- Set up rigorous procedures for every stage of equipment testing.
- Compute accurate test results.
- Evaluate and interpret final test data.

Also Available ...

FIRED HEATERS SPRAY DRYERS MIXING EQUIPMENT *(Impeller Type)* TRAY DISTILLATION COLUMNS CONTINUOUS DIRECT-HEAT ROTARY DRYERS CENTRIFUGAL PUMPS *(Newtonian Liquids)* CENTRIFUGES PARTICLE SIZE CLASSIFIERS EVAPORATORS

appropriate supports and distributors. DRY SOLIDS, PASTE, AND DOUGH MIXING EQUIPMENT



Ordering ...

Call (212) 705-7657. Or Fax us at: (212) 752-3294. Send your prepaid order to: AIChE Publication Sales Department, 345 E. 47th St., New York, NY 10017.

Test procedures are \$40 per copy. Or save 10% for the entire set of twelve copies. AIChE membership price is \$20 per copy.

American Institute of Chemical Engineers

345 East 47 Street, New York, NY 10017

17 We. 35

17 2011 7535